SOME FAMILIES OF COMBINATORIAL MATRICES AND THEIR ALGEBRAIC PROPERTIES

KOMBİNATORYAL MATRİSLERİN BAZI AİLELERİ VE ONLARIN CEBİRSEL ÖZELLİKLERİ

TALHA ARIKAN

PROF. DR. ADNAN TERCAN
Supervisor

PROF. DR. EMRAH KILIÇ Co-Supervisor

Submitted to Graduate School of Science and Engineering of Hacettepe University
as a Partial Fulfillment to the Requirements
for the Award of the Degree of Doctor of Philosophy
in Mathematics

This work named "Some Families of Combinatorial Matrices and Their Algebraic Properties" by TALHA ARIKAN has been approved as a thesis for the Degree of DOCTOR OF PHILOSOPHY IN MATHEMATICS by the below mentioned Examining Committee Members.

Prof. Dr. A. Çiğdem ÖZCAN

Head

Logia

Prof. Dr. Adnan TERCAN

Supervisor



Assoc. Prof. Dr. Neşe ÖMÜR

Member



Assoc. Prof. Dr. İlker AKKUŞ

Member



Assist. Prof. Dr. Nurettin IRMAK

Member

This thesis has been approved as a thesis for the Degree of **DOCTOR OF PHILOS-OPHY IN MATHEMATICS** by Board of Directors of the Institute for Graduate School of Science and Engineering.

Prof. Dr. Menemşe GÜMÜŞDERELİOĞLU

Director of the Institute of

Graduate School of Science and Engineering

YAYINLAMA VE FİKRİ MÜLKİYET HAKLARI BEYANI

Enstitü tarafından onaylanan lisansüstü tezimin/raporumun tamamını veya herhangi bir kısmını, basılı (kağıt) ve elektronik formatta arşivleme ve aşağıda verilen koşullarla kullanıma açma iznini Hacettepe Üniversitesine verdiğimi bildiririm. Bu izinle Üniversiteye verilen kullanım hakları dışındaki tüm fikri mülkiyet haklarım bende kalacak, tezimin tamamının ya da bir bölümünün gelecekteki çalışmalarda (makale, kitap, lisans ve patent vb.) kullanım hakları bana ait olacaktır.

Tezin kendi orijinal çalışmam olduğunu, başkalarının haklarını ihlal etmediğimi ve tezimin tek yetkili sahibi olduğumu beyan ve taahhüt ederim. Tezimde yer alan telif hakkı bulunan ve sahiplerinden yazılı izin alınarak kullanılması zorunlu metinlerin yazılı izin alınarak kullandığımı ve istenildiğinde suretlerini Üniversiteye teslim etmeyi taahhüt ederim.

Yükseköğretim Kurulu tarafından yayınlanan " **Lisansüstü Tezlerin Elektronik Ortamda Toplanması, Düzenlenmesi ve Erişime Açılmasına İlişkin Yönerge**" kapsamında tezim aşağıda belirtilen koşullar haricinde YÖK Ulusal Tez Merkezi / H. Ü. Kütüphaneleri Açık Erişim Sisteminde erişime açılır.

- o Enstitü / Fakülte yönetim kurulu kararı ile tezimin erişime açılması mezuniyet tarihimden itibaren 2 yıl ertelenmiştir. (1)
- o Enstitü / Fakülte yönetim kurulunun gerekçeli kararı ile tezimin erişime açılması mezuniyet tarihimden itibaren Ay ertelenmiştir. (2)
- o Tezimle ilgili gizlilik kararı verilmiştir. (3)

25 / 09 / 2018

Talha ARIKAN

"Lisansüstü Tezlerin Elektronik Ortamda Toplanması, Düzenlenmesi ve Erişime Açılmasına İlişkin Yönerge"

- (1) Madde 6. 1. Lisansüstü tezle ilgili patent başvurusu yapılması veya patent alma sürecinin devam etmesi durumunda, tez danışmanının önerisi ve enstitü anabilim dalının uygun görüşü üzerine enstitü veya fakülte yönetim kurulu iki yıl süre ile tezin erişime açılmasının ertelenmesine karar verebilir
- (2) Madde 6. 2. Yeni teknik, materyal ve metotların kullanıldığı, henüz makaleye dönüşmemiş veya patent gibi yöntemlerlekorunmamış ve internetten paylaşılması durumunda 3. Şahıslara veya kurumlara haksız kazanç imkanı oluşturabilecek bilgi ve bulguları içeren tezler hakkında tez danışmanının önerisi ve enstitü anabilim dalının uygun görüşü üzerine enstitü ve fakülte yönetim kurulunun gerekçeli kararı ile altı ayı aşmamak üzere tezin erişime açılması engellenebilir.
- (3) Madde 7. 1. Ulusal çıkarları veya güvenliği ilgilendiren, emniyet, istihbarat, savunma ve güvenlik, sağlık vb. konulara ilişkin lisansüstü tezlerle ilgili gizlilik kararı, tezin yapıldığı kurum tarafından verilir*. Kurum ve kuruluşlarla yapılan işbirliği protokolü çerçevesinde hazırlanan lisansüstü tezlere ilişkin gizlilik kararı ise, ilgili kurum ve kuruluşun önerisi ile enstitü veya fakültenin uygun görüşü üzerine üniversite yönetim kurulu tarafından verilir. Gizlilik kararı verilen tezler Yükseköğretim Kuruluna bildirilir.

Madde 7. 2. Gizlilik kararı verilen tezler gizlilik süresince enstitü veya fakülte tarafından gizlilik kuralları çerçevesinde muhafaza edilir, gizlilik kararının kaldırılması halinde Tez Otomasyon Sistemine yüklenir.

* Tez danışmanının önerisi ve enstitü anabilim dalının uygun görüşü üzerine enstitü veya fakülte yönetim kurulu tarafından karar verilir.

ETHICS

In this thesis study, prepared in accordance with the spelling rules of Institute of Graduate Studies in Science of Hacettepe University,

I declare that

- all the information and documents have been obtained in the base of the academic rules
- all audio-visual and written information and results have been presented according to the rules of scientific ethics
- in case of using others Works, related studies have been cited in accordance with the scientific standards
- all cited studies have been fully referenced
- I did not do any distortion in the data set
- and any part of this thesis has not been presented as another thesis study at this or any other university.

21/09/2018

TALHA ARIKAN

ABSTRACT

SOME FAMILIES OF COMBINATORIAL MATRICES AND THEIR ALGEBRAIC PROPERTIES

Talha ARIKAN

Doctor of Philosophy, Department of Mathematics

Supervisor: Prof. Dr. Adnan TERCAN

Co-Supervisor: Prof. Dr. Emrah KILIÇ

2018 September, 130 Pages

In this thesis, we will study some properties of certain families of combinatorial matrices. While some of the families will be examined throughout this thesis are new and firstly investigated, the others are the generalizations of some of the previously known matrices. We gather our studies into six different groups. They are non-symmetric band matrices with Gaussian q-binomial entries, generalization of the super Catalan matrix, families of Max and Min matrices, a non-symmetric variant of the Filbert matrix, a nonlinear generalization of the Filbert matrix and some certain Hessenberg matrices. For all matrices will be studied except the Hessenberg matrices, we present explicit formulæ for the LU-decompositions, determinants, inverse and LU-decompositions of the inverses of the matrices as well as the Cholesky decompositions when the matrix is symmetric. Additionally, we evaluate some certain Hessenberg determinants via generating function method. We use some new and existing methods to prove our claims. Particularly, we present a new method to evaluate determinants of some Hessenberg matrices whose entries consist of terms of higher order linear recursive sequences.

Keywords: LU and Cholesky decomposition, inverse matrix, determinant, Fibonacci numbers, Gaussian q-binomial and Fibonomial coefficients, Zeilberger's algorithm, Filbert matrix, Hessenberg matrices.

ÖZET

KOMBİNATORYAL MATRİSLERİN BAZI AİLELERİ VE ONLARIN CEBİRSEL ÖZELLİKLERİ

Talha ARIKAN

Doktora, Matematik Bölümü

Tez Danışmanı: Prof. Dr. Adnan TERCAN

İkinci Tez Danışmanı: Prof. Dr. Emrah KILIÇ

Eylül 2018, 130 Sayfa

Bu tez çalışmasında belirli kombinatoryal matris ailelerinin bazı özelliklerini çalışacağız. Tez boyunca incelenecek matris ailelerinin bazıları yeni ve ilk olarak araştırılacak olup diğerleri ise daha önceden bilinen bazı matrislerin genellemeleri olacaktır. Çalışmalarımızı, girdileri Gauss q-katsayıları olan simetrik olmayan bant matrisleri, süper Catalan matrisinin genellemesi, Max ve Min matris aileleri, Filbert matrisinin simetrik olmayan bir varyantı, Filbert matrisinin lineer olmayan bir genellemesi ve bazı Hessenberg matrisleri olmak üzere altı farklı grupta bir araya getirdik. Hessenberg matrisleri dışındaki çalışacağımız tüm matrislerin, LU-ayrışımları, determinantları, tersleri ve terslerinin LU-ayrışımları için net formüller vereceğiz. Bunun yanında matrislerin simetrik olmaları durumunda Cholesky ayrışımları için de ilgili formülleri sunacağız. Ayrıca bazı belirli Hessenberg determinantlarını üreteç fonksiyonları yardımıyla hesaplayacağız. Bu iddialarımızı ispatlamak için bazı yeni ve daha önceden kullanılan metotları kullanacağız. Özel olarak, girdileri yüksek mertebeden lineer indirgeme dizilerinin terimlerinden oluşan bazı Hessenberg matrislerinin determinantlarını hesaplamak için yeni bir yöntem vereceğiz.

Anahtar Kelimeler: LU ve Cholesky ayrışımı, ters matris, determinant, Fibonacci sayıları, Gauss q-binom ve Fibonomial katsayıları, Zeilberger'in algoritması, Filbert matrisi, Hessenberg matrisleri.

ACKNOWLEDGMENTS

First and foremost, I would like to extend my sincere thanks to my supervisors Prof. Dr. Adnan Tercan and Prof. Dr. Emrah Kılıç for their supports and helps. Particularly, I am grateful to Prof. Dr. Emrah Kılıç for his patient, guidance, enthusiastic encouragement and valuable advices during my Ph.D. education.

I would like to thank Assoc. Prof. Dr. İlker Akkuş and Assist. Prof. Dr. Nurettin Irmak for sparing their time and effort as members of thesis monitoring committee. Many thanks are also to Prof. Dr. Ayşe Çiğdem Özcan, Assoc. Prof. Dr. Neşe Ömür, Prof. Dr. Sait Halıcıoğlu and Prof. Dr. Yücel Tıraş for being committee members of my Ph.D. defense.

I wish to record my deep sense of appreciation and thankfulness to all members of Department of Mathematics of Hacettepe University. In particular, I am grateful to all my colleagues for their close friendship and pleasant times.

Also I would like to thank my mathematician friends Ekrem Çelik, Kamil Otal and Ernist Tilenbaev for their supports and friendship.

I would like to express my deep gratitude to Prof. Dr. Helmut Prodinger for his guidance and collaboration during my visit in Stellenbosch. It is a great experience for me to work with and learn from him.

I gratefully acknowledge generous financial support from the Scientific and Technological Research Council of Turkey (TÜBİTAK) via programs 2211/E and 2214/A. I also owe to my university for the support to attend international conferences.

Lastly and most importantly, I wish to express my deepest gratitude to my wife for her endless support. And many thanks to my daughter who makes me more motivated for the last two years. Besides, I really appreciate my dear mother, father and brothers for their valuable supports.

CONTENTS

	AB	STRACT	i
	ÖZ	${f ET}$	iii
	\mathbf{AC}	KNOWLEDGMENTS	${f v}$
	CO	NTENTS	vi
	LIS	T OF SYMBOLS AND ABBREVIATIONS	ix
	LIS	T OF TABLES	xi
1	INT	TRODUCTION	1
2	PR	ELIMINARIES	4
	2.1	Linear Recursive Sequences	4
	2.2	Binomial Coefficients	7
	2.3	<i>q</i> -World	9
	2.4	Generating Functions	11
	2.5	Hypergeometric Series	13
		2.5.1 Zeilberger's Algorithm	16
		2.5.2 <i>q</i> -Zeilberger Algorithm	17
3	LIT	ERATURE REVIEW	19
	3.1	About Matrices	19
		3.1.1 Special Matrices	22
	3.2	Previous Studies	25

1

		3.2.1	Band Matrices	25
		3.2.2	Pascal Matrices	27
		3.2.3	Max and Min Matrices	27
		3.2.4	Hilbert, Filbert Matrices and Their Variants	29
		3.2.5	Hessenberg Matrices	31
4	RES	SULTS	}	35
	4.1	Auxili	ary Results	35
	4.2	A Fan	nily of the Non-Symmetric Band Matrices	39
		4.2.1	Main Results	41
		4.2.2	Proofs	44
		4.2.3	The Case Bandwidth $r+s+1$ with the Binomial Coefficients $\ .$	46
	4.3	A Ger	neralization of the Super Catalan Matrix	52
		4.3.1	The Matrix \mathcal{M}	53
		4.3.2	The Matrix \mathcal{T}	57
		4.3.3	The Matrices M and T	60
	4.4	Gener	al Family of the Max and Min Matrices	63
		4.4.1	Max Matrices and Their Reciprocal Analogues	65
		4.4.2	Min Matrices and Their Reciprocal Analogues	69
		4.4.3	Further Applications	72
	4.5	A Nor	n-symmetric Variant of the Filbert Matrix	75
		4.5.1	Main Results	76
		4.5.2	Proofs	78
		4.5.3	Applications	82
	4.6	A Nor	alinear Generalization of the Filbert Matrix	88
		4.6.1	A Nonlinear Filbert Matrix	89
		462	A Nonlinear Lilbert Matrix	96

CU:	RRIC	ULUM VITAE	129
REI	FEREI	NCES	121
	4.7.2	A Matrix Method to Compute a Class of Hessenberg Determinants	3116
	4.7.1	Main Results	102
4.7	Hesser	nberg Determinants via Generating Functions	102
	4.6.3	q-Analogues	98

LIST OF SYMBOLS AND ABBREVIATIONS

Symbols	
F_n	nth Fibonacci number
L_n	nth Lucas number
P_n	nth Pell number
$U_n(p,q)$	nth generalized Fibonacci number
$V_n(p,q)$	nth generalized Lucas number
U_n	$U_n(p,1)$
V_n	$V_n(p,1)$
Δ	$p^2 + 4$
W_n	nth Horadam number
T_n	nth Tribonacci number
\mathscr{H}_n	nth harmonic number
N	natural numbers
\mathbb{Z}	integers
\mathbb{C}	complex numbers
$\binom{n}{k}$	binomial coefficient
$\begin{bmatrix} n \\ k \end{bmatrix}_q$	Gaussian q -binomial coefficient
${n \brace k}_U$	generalized Fibonomial coefficient
${n \brace k}_F$	Fibonomial coefficient
$[n]_q$	$q ext{-integer}$
$(x;q)_n$	q-Pochhammer symbol
i	$\sqrt{-1}$
$(a)_k$	usual Pochhammer symbol (rising factorial = $a^{\overline{k}}$)
$a^{\underline{k}}$	falling factorial
[P]	Iverson notation (1 if P is true, 0 otherwise.)
M_N	square matrix M of order N
$[M_{kj}]$	infinite matrix M
$[M_{kj}]_{0 \le k, j \le N-1}$	square matrix M of order N whose indexes start at 0
$[M_{kj}]_{1 \le k, j \le N}$	square matrix M of order N whose indexes start at 1

 $D(a_n)$ diagonal matrix with a_k in kth row

 $\|M_N\|_{\infty}$ infinity-norm of the matrix M

 $||M_N||_1$ 1-norm of the matrix M

[] floor function

 $[x^n]f(x)$ coefficient of x^n in f(x)

 $\det M,\, |M| \qquad \qquad \det \mathrm{erminant} \,\, \mathrm{of} \,\, \mathrm{the} \,\, \mathrm{matrix} \,\, M$

 ${
m per} M$ permanent of the matrix M

 M^{-1} inverse of the matrix M

 M^T transpose of the matrix M

Abbreviations

RHS right hand side

LHS left hand side

LIST OF TABLES

1 So:	me particular	linear recursive	sequence exampl	es	5
-------	---------------	------------------	-----------------	----	---

1 INTRODUCTION

A matrix is a rectangular array of any algebraic objects for which addition and multiplication are defined. Matrices are the main subject of the fundamental mathematical branch linear algebra. Historically, it was not the matrix but a certain number associated with a square array of numbers called the determinant that was first recognized. The term **matrix** was coined by James Joseph Sylvester in 1850. Then Arthur Cayley developed algebraic properties of matrices. He firstly applied them to the study of systems of linear equations. So matrices are mostly used as a way to describe systems of linear equations as well as to represent data in a tabular view.

Matrix arises in several branches of science, as well as different mathematical disciplines. For example, they are frequently used in physics, computer graphics, probability theory and statistics. Moreover, in some social sciences like economics, the scientists often use matrices. Thus, manipulating matrices has drawn interest. A major branch of numerical analysis is devoted to the development of efficient algorithms for the computations of some properties of matrices. So certain matrices with known properties are important to check accuracy of newly developed algorithms.

In literature, there are various special matrices. Band, Toeplitz, Pascal, Lehmer, Hilbert, Filbert and Hessenberg matrices are some examples of these special matrices and the main interest of this thesis. Combinatorial matrices are the matrices whose entries consist of some combinatorial numbers. Pascal and Filbert matrices are the common examples of the combinatorial matrices. Band, Toeplitz and Hessenberg matrices are mainly used in numerical analysis. Thus several authors have been studied certain properties of them. Pascal, Lehmer, Hilbert and Filbert matrices have nice algebraic properties so they are useful to test accuracy of algorithms. Some authors studied various generalizations and variants of these kinds of matrices.

The aim of this thesis is to present some new combinatorial matrix families, which have not been studied before, and some generalizations of already known combinatorial matrices and their properties. We study some certain properties such as LU-decompositions, determinants, inverses etc. of these families of combinatorial matrices. We hope that the matrices we have studied in this thesis could be also used as test matrices. We will give some new methods and use existing methods to prove our claims.

Especially, we present a new computational method to evaluate certain Hessenberg determinants. We derive explicit formulæ related with the properties of these matrices and our main tool for finding them is to guess relevant quantities. This was done by experiments with a computer algebra system and spotting patterns. This becomes increasingly complicated when more new parameters are introduced. We have frequently used computer algebra systems Mathematica and Maple for our calculations.

In Section 2, we provide some essential information, which will be used throughout the thesis. In Section 3.1, we present some basic notions about matrices and introduce some special kinds of matrix families. Besides, in Section 3.2, we present previous studies related to these kinds of matrix families. This main section is divided into parts and each part includes a special family. Moreover, at the end of each this part, we indicate our motivation and brief introduction to our problem related to that matrix family.

Section 4 is devoted to the results which obtained in this thesis. In Section 4.1, we present some auxiliary results, which we use to prove some of our main results. On the other hand, these auxiliary results could be also applied to other matrices which are not considered in this thesis. In Sections 4.2-4.7, we provide our main results. Each section includes the results for a special family of combinatorial matrices.

In Section 4.2, we present a class of non-symmetric Toeplitz band matrices with upper bandwidth s and lower bandwidth r whose entries are defined via the Gaussian qbinomial coefficients. We provide explicit formulæ for the LU-decomposition, determinant and LU-decomposition of the inverse matrix. The case r = s is the generalization of the results given in [1]. Furthermore, we derive some complementary results related to matrix which includes usual binomial coefficients. Our results are presented in [2].

In Section 4.3, we obtain the generalizations with two additional parameters of the results in [3]. We also present the idea how one can obtain similar generalizations by the help of already known results. In [4], we publish our studies.

In Section 4.4, we define two new families of the matrices, which are called Max and Min matrices, whose entries run in left-reversed and up-reversed L-shaped pattern, respectively. Our results also cover the results given in [5]. In [5] the authors used a method based on another auxiliary matrix family. But we use elementary linear algebra tools to derive our results which are simpler and more convenient. As an application,

we obtain a sequential generalization of the Lehmer matrix and its reciprocal analogue.

In Section 4.5, we define a new non-symmetric matrix via q-integers. Non-symmetric variants of the Filbert and Lilbert matrices come out as corollaries for the special choices of the parameters. We derive explicit formulæ for the LU-decompositions, inverse matrices L^{-1} and U^{-1} and inverses for whole matrices.

Some authors have studied many generalizations and variant of the Filbert matrix as we do in Section 4.5. But so far no one has studied a generalization or variant where the indexes of the recursive sequence are in nonlinear form. In Section 4.6, we introduce a new nonlinear generalization of the Filbert matrix with indexes in geometric progression for some parameters as well as a nonlinear generalization of the Lilbert matrix. As in Section 4.5, we present the LU-decompositions, inverse matrices L^{-1} and U^{-1} and inverses for the nonlinear generalizations of the Filbert and Lilbert matrices as well as we provide the Cholesky decompositions when the matrices are symmetric. We present our results in [6].

Finally, in Section 4.7, we present the generating function method to evaluate the determinant of new three classes of the Hessenberg matrices. This method was introduced in [7]. We extend it and obtain some further results. We also provide many special examples to see how the method works. By the help of our results, many determinantal formula which have been found in the previous studies, can be easily retrieved. As an application, we give a new and an elegant method to compute the determinants of the Hessenberg matrices whose entries consist of the terms of the higher order linear recursive sequences, which based on to find an adjacency-factor matrix. Our results are published in [8].

2 PRELIMINARIES

In this section, we will present some fundamental notions, which will be used throughout the thesis.

2.1 Linear Recursive Sequences

A recursive sequence is defined by a rule which gives the next term as a function of the previous terms. This rule is called the **recurrence relation** of the corresponding sequence. If we denote the nth term of a given sequence by u_n , such a recurrence relation is of the form

$$u_n = f(u_{n-1}, u_{n-2}, \dots, u_{n-k}),$$

where f is a function with k inputs.

Definition 2.1. For any reals p_i such that $i \in \{1, 2, ..., k\}$ and $p_k \neq 0$, the kth order linear recursive sequence $\{u_n\}$ with constant coefficients is defined by the rule for $n \geq k$,

$$u_n = p_1 u_{n-1} + p_2 u_{n-2} + \dots + p_k u_{n-k}$$
(2.1)

with arbitrary initial values u_t for $0 \le t < k$ and assumed that at least one of them is different from zero.

It is obviously seen that the terms of the kth order linear recursive sequence $\{u_n\}$ defined by the rule (2.1) are uniquely determined by the coefficients p_i 's and its initial values. We give the most common special cases of the sequence $\{u_n\}$ with Table 1. These number sequences have been studied by many authors. We refer to [9, 10, 11, 12] for more details about them.

Now we shall give two important definitions about linear recursive sequences.

Definition 2.2. The characteristic polynomial of the sequence $\{u_n\}$ defined by (2.1) is the polynomial

$$f(x) = x^k - p_1 x^{k-1} - p_2 x^{k-2} - \dots - p_k$$

and the equation

$$x^k - p_1 x^{k-1} - p_2 x^{k-2} - \dots - p_k = 0$$

is called **characteristic equation** of the sequence $\{u_n\}$.

Coefficients	Initials	Sequence Name
$p_1 = p_2 = 1$	$u_0 = 0, u_1 = 1$	Fibonacci Seq. $\{F_n\}$
$p_1 = p_2 = 1$	$u_0 = 2, u_1 = 1$	Lucas Seq. $\{L_n\}$
$p_1 = 2, p_2 = 1$	$u_0 = 0, u_1 = 1$	Pell Seq. $\{P_n\}$
$p_1 = 1, p_2 = 2$	$u_0 = 0, u_1 = 1$	Jacobsthal Seq. $\{J_n\}$
$p_1 = p, \ p_2 = q$	$u_0 = 0, \ u_1 = 1$	Gen. Fibonacci Seq. $\{U_n(p,q)\}$
$p_1 = p, \ p_2 = q$	$u_0 = 2, \ u_1 = p$	Gen. Lucas Seq. $\{V_n(p,q)\}$
$p_1 = p, \ p_2 = -q$	$u_0 = a, \ u_1 = b$	Horadam Seq. $\{W_n\}$
$p_1 = 2, p_2 = -1$	$u_0 = 0, u_1 = 1$	Natural Numbers ℕ
$p_1 = p_2 = p_3 = 1$	$u_0 = 0, \ u_1 = u_2 = 1$	Tribonacci Seq. $\{T_n\}$
$p_1 = 0, p_2 = p_3 = 1$	$u_0 = 3, u_1 = 0, u_2 = 2$	Perrin Seq. $\{\mathcal{P}_n\}$

Table 1: Some particular linear recursive sequence examples

It is possible to derive any term of the sequence $\{u_n\}$ by the help of its recurrence relation. Nevertheless, it is not useful to compute higher terms. So one needs a closed formula such that

$$u_n = g(n) (2.2)$$

to compute any desired term of the sequence $\{u_n\}$ with less computation.

Definition 2.3. Such an explicit formula given by (2.2) is called **Binet formula** of the linear recursive sequence $\{u_n\}$.

The following theorem provides us how to find the Binet formula of the linear recursive sequence $\{u_n\}$.

Theorem 2.1. Let the characteristic polynomial of the sequence $\{u_n\}$ factor over the complex number as

$$f(x) = (x - r_1)^{m_1} (x - r_2)^{m_2} \dots (x - r_d)^{m_d},$$

where r_1, r_2, \ldots, r_d are distinct nonzero complex numbers and m_1, m_2, \ldots, m_d are positive integers such that $m_1 + m_2 + \cdots + m_d = k$. Then there exist polynomials $g_1(n), g_2(n), \ldots, g_d(n)$ with $\deg g_i \leq m_i - 1$ for all $i \in \{1, 2, \ldots, d\}$ such that

$$u_n = g_1(n)r_1^n + g_2(n)r_2^n + \dots + g_d(n)r_d^n$$

for $n \geq 0$.

One can easily find the proof of the above theorem in any textbook about difference equations. As a special case, when $m_1 = m_2 = \cdots = m_k = 1$, i.e. the characteristic polynomial has no multiple root, the Binet formula of the sequence $\{u_n\}$ takes the form

$$u_n = c_1 r_1^n + c_2 r_2^n + \dots + c_k r_k^n,$$

where for all $i \in \{1, 2, ..., k\}$, c_i 's are constants determined according to the initial values of the sequence $\{u_n\}$.

Now we will apply the above theorem to the generalized Fibonacci sequence $\{U_n(p,q)\}$, which is defined by the recurrence relation for n > 2,

$$U_n(p,q) = pU_{n-1}(p,q) + qU_{n-2}(p,q),$$

in order to find its Binet formula for the case $p^2+4q \neq 0$. The characteristic polynomial of it is

$$f(x) = x^2 - px - q = (x - \alpha)(x - \beta),$$

where $\alpha, \beta = (p \mp \sqrt{p^2 + 4q})/2$. So the Binet formula is of the form

$$U_n(p,q) = c_1 \alpha^n + c_2 \beta^n.$$

Since $U_0(p,q) = 0$ and $U_1(p,q) = 1$, finally the Binet formula of the sequence $\{U_n(p,q)\}$ is found as

$$U_n(p,q) = \frac{\alpha^n - \beta^n}{\alpha - \beta}.$$

Similarly, the Binet formula of the generalized Lucas sequence $\{V_n(p,q)\}$ is

$$V_n(p,q) = \alpha^n + \beta^n$$
.

Throughout the thesis, we will frequently study the generalized Fibonacci sequence $\{U_n(p,1)\}$ and Lucas sequence $\{V_n(p,1)\}$ and briefly denote them by $\{U_n\}$ and $\{V_n\}$, respectively, unless otherwise specified. They satisfy the recurrence relations for n > 2,

$$U_n = pU_{n-1} + U_{n-2},$$

$$V_n = pV_{n-1} + V_{n-2},$$

with initial values $U_0 = 0$, $U_1 = 1$ and $V_0 = 2$, $V_1 = p$, respectively. Especially, their Binet formulæ are

$$U_n = \frac{\alpha^n - \beta^n}{\alpha - \beta}$$
 and $V_n = \alpha^n + \beta^n$, (2.3)

where $\alpha, \beta = (p \mp \sqrt{\Delta})/2$ and $\Delta = p^2 + 4$, respectively.

We need the following known identity including the generalized Fibonacci numbers for later use.

Proposition 2.1. For integers n, m and k, the following equation holds.

$$U_m U_n = (-1)^{n+k} U_{m-n+k} U_k + U_{m+k} U_{n-k}.$$
(2.4)

The proof can be immediately done by the Binet formula. For the numerous identities and properties, we refer to comprehensive books [11, 12].

2.2 Binomial Coefficients

The binomial coefficients occur in almost all areas of mathematics. The binomial coefficients get their name from the **binomial theorem**, which describes the expansion of the powers of a binomial (x + y). Also, the binomial coefficients have a combinatorial interpretation, which counts the number of the ways of choosing k objects among n objects without replacement, and are denoted by the symbol $\binom{n}{k}$. From the combinatorial meaning of them, it is easily seen that they are defined by the ratio for $0 < k \le n$,

$$\binom{n}{k} = \frac{n!}{k!(n-k)!}$$

with $\binom{n}{0} = 1$, which represents the empty choice. The integers n and k are called the **upper index** and the **lower index**, respectively. The indexes are restricted to be non-negative integers by the combinatorial interpretation. Since the binomial coefficients have many usages besides its combinatorial interpretation, it is useful to define them for any real upper index. So for any real r and integer k, they are formally defined by

$$\binom{r}{k} = \begin{cases} \frac{r(r-1)\dots(r-k+1)}{k!} & \text{if } k \ge 0, \\ 0 & \text{if } k < 0. \end{cases}$$

The two variable sequence $\binom{n}{k}_{n,k\geq 0}$ is also a recursive sequence which satisfies the recurrence relation for $1\leq k\leq n-1$,

$$\binom{n}{k} = \binom{n-1}{k} + \binom{n-1}{k-1}$$

with $\binom{n}{0} = \binom{n}{n} = 1$. This relation is easily verified by the definition of the binomial coefficients. By the help of this relation, one can prove that each binomial coefficient

with integer indexes is an integer. More explanation and information can be found in the book "Concrete Mathematics" written by Graham, Knuth and Patashnik [13].

Recall the well-known Vandermonde's identity for later use.

Proposition 2.2. For nonnegative integers k, m and n,

$$\binom{m+n}{k} = \sum_{d=0}^{k} \binom{m}{d} \binom{n}{k-d}.$$
 (2.5)

We will recall three different proofs for the Vandermonde's identity in the forthcoming subsections. For many variants of it, we refer to [13, p. 169].

By replacing each integer appearing in the numerator and denominator of $\binom{n}{k}$ with its respective generalized Fibonacci number, i.e. replace i by U_i , we can define a recursive analogue of the binomial coefficients, which is called generalized Fibonomial coefficients introduced by Jarden and Motzkin [14]. Formally, we have the following definition.

Definition 2.4. For integers such that 0 < k < n, the **generalized Fibonomial** coefficients are defined by

$${n \brace k}_{U} := \frac{U_{n}U_{n-1}\dots U_{1}}{(U_{1}U_{2}\dots U_{k})(U_{1}U_{2}\dots U_{n-k})}$$

with $\binom{n}{n}_U = \binom{n}{0}_U = 1$ and 0 otherwise.

When the case p = 1, i.e. $U_n = F_n$, the generalized Fibonomial coefficients $\binom{n}{k}_U$ are reduced to the usual Fibonomial coefficients $\binom{n}{k}_F$. We refer to [15, 16, 17, 18] for more details about the generalized and usual Fibonomial coefficients.

The generalized Fibonomial coefficients satisfy the following recurrence relation for $1 \le k \le n-1$,

$${n \brace k}_U = U_{k+1} {n-1 \brace k}_U + U_{n-k-1} {n-1 \brace k-1}_U$$

with $\binom{n}{n}_U = \binom{n}{0}_U = 1$. This relation follows by the equation $U_n = U_{k+1}U_{n-k} + U_kU_{n-k-1}$ (This can be shown by taking m = 1 in (2.4) and using the fact that $U_{-n} = (-1)^{n+1}U_n$). Surprisingly, as in the binomial coefficients, the generalized Fibonomial coefficients are always integers. From the recurrence relation, it is easy to see this fact by induction. On the other hand, this is not always true for the generalized Fibonomial coefficients defined by the sequence $\{U_n(p,q)\}$.

2.3 *q*-World

In classical q-analysis, the q-analogue of a nonnegative integer, q-integer, is defined by

$$[n]_q = \frac{1 - q^n}{1 - q} = \sum_{k=0}^{n-1} q^k.$$

From the above definition, it is easily seen that

$$\lim_{q \to 1} [n]_q = n.$$

For nonnegative integer n, the **q-Pochhammer symbol** is defined as

$$(x;q)_n = (1-x)(1-xq)\dots(1-xq^{n-1})$$

with $(x;q)_0 = 1$. So one can derive the q-analogue of n! as follows

$$[n]_q! = \frac{(q;q)_n}{(1-q)^n}.$$

Now we can move on the q-analogue of the binomial coefficient.

Definition 2.5. For nonnegative integers such that $n \ge k$, the **Gaussian q-binomial** coefficients are defined by

$$\begin{bmatrix} n \\ k \end{bmatrix}_q = \frac{[n]_q!}{[k]_q![n-k]_q!} = \frac{(q;q)_n}{(q;q)_k(q;q)_{n-k}}$$

and 0 otherwise.

Obviously, we have

$$\lim_{q \to 1} \begin{bmatrix} n \\ k \end{bmatrix}_q = \binom{n}{k}.$$

The Gaussian q-binomial coefficients satisfy the following two equivalent recurrence relations for $1 \le k \le n$,

$$\begin{bmatrix} n \\ k \end{bmatrix}_q = q^k \begin{bmatrix} n-1 \\ k \end{bmatrix}_q + \begin{bmatrix} n-1 \\ k-1 \end{bmatrix}_q$$

and

$$\begin{bmatrix} n \\ k \end{bmatrix}_q = \begin{bmatrix} n-1 \\ k \end{bmatrix}_q + q^{n-k} \begin{bmatrix} n-1 \\ k-1 \end{bmatrix}_q$$

with $\begin{bmatrix} n \\ 0 \end{bmatrix}_q = 1$. Inductively, by the above relations, it is seen that every Gaussian q-binomial coefficient is a polynomial in q.

9

It is possible to extend Definition 2.5 for any real upper index by considering

$$\begin{bmatrix} r \\ k \end{bmatrix}_q = \frac{(q^{1+r-k}; q)_k}{(q; q)_k},$$

where r is an arbitrary real and $k \geq 0$ is an integer.

The Binet formulæ given by (2.3) can be rewritten as

$$U_n = \alpha^{n-1} \frac{1 - q^n}{1 - q} \tag{2.6}$$

and

$$V_n = \alpha^n (1 + q^n) \tag{2.7}$$

with $q = \beta/\alpha = -\alpha^{-2}$, so that $\alpha = \mathbf{i}q^{-1/2}$, where $\mathbf{i} = \sqrt{-1}$. The RHS of the equations (2.6) and (2.7) are the q-forms of the generalized Fibonacci and Lucas numbers, respectively.

Thus, by the help of the q-forms, the link between the generalized Fibonomial and Gaussian q-binomial coefficients is found as

$${n \brace k}_U = \alpha^{k(n-k)} {n \brack k}_q = \mathbf{i}^{k(n-k)} q^{\frac{1}{2}k(k-n)} {n \brack k}_q.$$
 (2.8)

As it is seen from the above relationship, if we have an identity including generalized Fibonacci numbers then we can convert it into the q-form or vice versa. Since there are many useful identities and tools in q-analysis, studying the q-form of an identity is more advantageous than studying the original form. In this thesis, we will often use this idea. In other words, we will prove some q-identities for general parameter q then the identities including the generalized Fibonacci numbers or Fibonomial coefficients would come out as corollaries for special value of q.

Now we shall give some known identities. The following theorems are the one version of the Cauchy binomial theorem and Rothe's formula.

Theorem 2.2. For $n \geq 0$,

$$\sum_{k=0}^{n} {n \brack k}_q q^{\binom{k+1}{2}} x^k = (-xq;q)_n = \prod_{k=1}^{n} (1+xq^k).$$
 (2.9)

Theorem 2.3. For $n \geq 0$,

$$\sum_{k=0}^{n} {n \brack k}_q (-1)^k q^{\binom{k}{2}} x^k = (x; q)_n = \prod_{k=0}^{n-1} (1 - xq^k).$$
 (2.10)

For the proofs of the above theorems and more identities, we refer to [19]. Furthermore, one can look at [20, 21] for more information about the q-analysis.

Now we present the q-analogue of the Vandermonde's identity (2.5).

Theorem 2.4 (q-Vandermonde identity). For nonnegative integers k, m and n,

$$\begin{bmatrix} m+n \\ k \end{bmatrix}_q = \sum_{d=0}^k \begin{bmatrix} m \\ k-d \end{bmatrix}_q \begin{bmatrix} n \\ d \end{bmatrix}_q q^{d(m-k+d)}.$$
(2.11)

A proof of this identity can be found in [22]. We will provide a computer-based proof of it later. Now suppose that we verify this identity. Then if we let $q \to 1$ in the equation (2.11), we will get the Vandermonde's identity (2.5). This approach gives us another advantage of studying q-identities. In other words, if we have a q-identity and the limit $q \to 1$ is applicable to it, then we achieve another useful identity, as well. Thus studying in q-world enables us to obtain more general results.

2.4 Generating Functions

Generating functions are one of the most useful inventions in mathematics. The generating function is a way of encoding a sequence $\{a_n\}_{n\geq 0}$ by treating them as the coefficients of a power series. Briefly, generating functions transform the problems about the sequences into the problems about power series or functions. In this manner, manipulating infinite sequences gets easier. Wilf's book "Generatingfunctionology" [23] is totally devoted to the generating functions. We refer to it in order to get more information about the generating functions.

For a given sequence $\{a_n\}_{n\geq 0}$, the **generating function** of $\{a_n\}_{n\geq 0}$ is the power series

$$A(x) := \sum_{n \ge 0} a_n x^n. {2.12}$$

It is sometimes called **ordinary generating function** to distinguish from other types of generating functions (see [13, 22, 23]).

In general, the power series (2.12) is considered as a formal power series, i.e. an algebraic object. Thus we are not worried about the convergence. The power series (2.12) may also be considered as an analytic function on the interval of convergence of

it. This treatment allows us to get some asymptotic information about the terms of the sequence (see [24]).

Now, we shall compute the generating function of the Fibonacci sequence as an illustration. Let $F(x) = \sum_{n\geq 0} F_n x^n$. So we have

$$F(x) = F_0 + F_1 x + F_2 x^2 + \cdots + F_n x^n + \cdots$$

$$-xF(x) = -F_0 x - F_1 x^2 - \cdots - F_{n-1} x^n - \cdots$$

$$-x^2 F(x) = -F_0 x^2 - \cdots - F_{n-2} x^n - \cdots$$

After adding these three equations, we obtain

$$F(x)(1-x-x^2) = F_0 + (F_1 - F_0)x + (F_2 - F_1 - F_0)x^2 + \cdots + (F_n - F_{n-1} - F_{n-2})x^n + \cdots$$

Since for $n \ge 2$, $F_n - F_{n-1} - F_{n-2} = 0$ by its recurrence relation, the generating function of the Fibonacci numbers is

$$F(x) = \frac{x}{1 - x - x^2}.$$

Similarly, the generating function of the Lucas numbers is

$$L(x) = \frac{2 - x}{1 - x - x^2}.$$

By the above approach, one can easily compute the generating function of the kth order linear recursive sequence $\{u_n\}$, defined by (2.1), as

$$\sum_{n>0} u_n x^n = \frac{p(x)}{1 - p_1 x - p_2 x^2 - \dots - p_k x^k},$$

where p(x) is a polynomial, which will be determined according to the initial values of the sequence $\{u_n\}$ such that $\deg p(x) < k$.

Let $A(x) = \sum_{n\geq 0} a_n x^n$ and $B(x) = \sum_{n\geq 0} b_n x^n$. Then for any complex constants c_1 and c_2 , the following properties hold.

$$c_1 A(x) + c_2 B(x) = \sum_{n \ge 0} (c_1 a_n + c_2 b_n) x^n,$$

$$A(x) B(x) = \sum_{n \ge 0} \left(\sum_{k=0}^n a_k b_{n-k} \right) x^n.$$
(2.13)

The proofs are straightforward. Both of the above equations can be inductively generalized by considering arbitrary finite number of generating functions. We can select just even powers or odd powers out of the power series A(x) as follows

$$\sum_{n>0} a_{2n} x^{2n} = \frac{A(x) + A(-x)}{2} \text{ or } \sum_{n>0} a_{2n+1} x^{2n+1} = \frac{A(x) - A(-x)}{2}, \tag{2.14}$$

respectively. For more properties and some special generating functions, we refer to [23].

We would like to recall the following useful theorem.

Theorem 2.5. Let $A(x) = \sum_{n\geq 0} a_n x^n$ and $B(x) = \sum_{n\geq 0} b_n x^n$ be analytic complex functions in a non-empty open neighborhood D of zero. If A(x) = B(x) for all $x \in D$, then $a_n = b_n$ for all $n \in \mathbb{Z}$.

The proof of this theorem can be found in [25]. In [23], the author gave a method called "Snake Oil", which is based on this theorem.

Now we will present another proof of the Vandermonde's identity (2.5) by the help of Theorem 2.5. By the binomial theorem, we know that

$$\sum_{k\geq 0} {m+n \choose k} x^k = (1+x)^{m+n}$$

and by the identity (2.13), we write

$$\sum_{k\geq 0} \left(\sum_{d=0}^k \binom{m}{d} \binom{n}{k-d} \right) x^k = (1+x)^m (1+x)^n.$$

Since the both sides of (2.5) have the same generating function $(1+x)^{m+n}$ for all $x \in \mathbb{C}$, they must be equal for all nonnegative integers k, m and n by Theorem 2.5. We will also use this argument in Section 4.7.

2.5 Hypergeometric Series

Hypergeometric series appear in many areas of mathematics such as combinatorics, analysis, applied mathematics etc. The history of hypergeometric series was launched many years ago by Euler, Gauss, and Riemann. Although the topic is very old, it is still the subject of a lot of ongoing research. There are many books devoted only to

the hypergeometric series. Since we encountered some hypergeometric sums, which are truncated hypergeometric series, in this thesis, we would like to give very brief introductory information about them.

Definition 2.6. The series

$$\sum_{k>0} t_k$$

is called **hypergeometric series** if $t_0 = 1$ and the ratio t_{k+1}/t_k is a rational function of k, i.e.

$$\frac{t_{k+1}}{t_k} = \frac{p(k)}{h(k)},\tag{2.15}$$

where p(k) and h(k) are polynomials of k. In this case, t_k is called **hypergeometric** term. The functions generated by hypergeometric terms are called **hypergeometric** functions and truncated hypergeometric series are called **hypergeometric** sums.

For example, since

$$\frac{\binom{m}{d+1}\binom{n}{k-d-1}}{\binom{m}{d}\binom{n}{k-d}} = \frac{(m-d)(k-d)}{(d+1)(n-k+d+1)},$$

the sum on the RHS of the Vandermonde's identity (2.5) is a hypergeometric sum.

Consider the hypergeometric function

$$\sum_{k\geq 0} t_k z^k. \tag{2.16}$$

If the polynomials in (2.15) are completely factored, then we write

$$\frac{t_{k+1}}{t_k} = \frac{p(k)}{h(k)} = \frac{(k+a_1)(k+a_2)\dots(k+a_n)}{(k+b_1)(k+b_2)\dots(k+b_m)(k+1)},$$
(2.17)

where the factor (k + 1) in the denominator is presented for some historical reasons of notation. If -1 is not a root of the polynomial h(k), then we can multiply both the numerator and denominator with the factor (k + 1) for the convenience. Then the hypergeometric function (2.16) is notationally shown as

$$\sum_{k\geq 0} t_k z^k = {}_n F_m \left[\begin{array}{ccc} a_1 & a_2 & \cdots & a_n \\ b_1 & b_2 & \cdots & b_m \end{array} ; z \right] = \sum_{k\geq 0} \frac{(a_1)_k (a_2)_k \dots (a_n)_k}{(b_1)_k (b_2)_k \dots (b_m)_k} \frac{z^k}{k!},$$

where $(a)_k$ is the **usual Pochhammer symbol** defined by $(a)_k = (a)(a+1)\dots(a+k-1)$, also known as rising factorial.

In general, the first term t_0 doesn't have to be 1. If it is not equal to 1 but different from zero and t_k satisfies the equation (2.17), then the hypergeometric function (2.16) is represented as follows

$$\sum_{k\geq 0} t_k z^k = t_0 \times {}_n F_m \left[\begin{array}{cccc} a_1 & a_2 & \cdots & a_n \\ b_1 & b_2 & \cdots & b_m \end{array} ; z \right].$$

For more extensive knowledge about hypergeometric series, we refer to [26, 27]. Nowadays, hypergeometric series are also well understood from an algorithmic point of view. There are some algorithms [28] to deal with the hypergeometric series, sums or functions and also their implementations to computer algebra systems, such as Maple and Mathematica. The most efficient and modern of them is celebrated Zeilberger's algorithm. We will mention about it after a while.

Inherently, the most natural question is "What is the q-analogue of the hypergeometric series?".

Definition 2.7. The series

$$\sum_{k>0} t_k(q)$$

is called **q-hypergeometric series** if the ratio t_{k+1}/t_k is a rational function of q^k , i.e.

$$\frac{t_{k+1}(q)}{t_k(q)} = \frac{p(x)}{h(x)},$$

where p(x) and h(x) are polynomials of x and $x = q^k$. In this case, $t_k(q)$ is called q-hypergeometric term. The functions generated by q-hypergeometric terms are called q-hypergeometric functions and truncated q-hypergeometric series are called q-hypergeometric sums.

As an example, since

$$\frac{\begin{bmatrix} m \\ d+1 \end{bmatrix}_q \begin{bmatrix} n \\ k-d-1 \end{bmatrix}_q}{\begin{bmatrix} m \\ d \end{bmatrix}_q \begin{bmatrix} n \\ k-d \end{bmatrix}_q} = q^{m-k+1} \frac{(q^d-q^k)(q^d-q^n)}{(1-q^dq^{m-k+1})(1-qq^d)},$$

the sum on the RHS of the q-Vandermonde's identity (2.11) is a q-hypergeometric sum.

In order not to get out of the subject of our thesis, we don't want to give more details. For more details, see [19, 27, 29].

Naturally, there is q-analogue of the Zeilberger's algorithm, which deals with the q-hypergeometric series, sums and functions.

2.5.1 Zeilberger's Algorithm

Since the people have encountered hypergeometric sums and series in many areas of mathematics, some mechanical methods to deal with them have been derived. The first mechanical method for hypergeometric sums was discovered by Sister Mary Celine Fasenmyer [30] in 1945. Such methods serve to compute them directly or prove equalities [28].

One of the most popular mechanical methods is Gosper's algorithm [31], which computes some indefinite hypergeometric sums in terms of another hypergeometric term. This method based on rewriting the hypergeometric term into telescoping form. Unfortunately, it can find closed forms for only a few classes of the hypergeometric sums we meet in practice. Namely, it is applicable to the limited numbers of the hypergeometric sums.

In 1991, Doron Zeilberger [32, 33, 34] showed how to extend Gosper's algorithm so that it becomes even more effective, making it succeed in vastly more cases. With Zeilberger's extension, we can also handle hypergeometric series, not just sums. For very brief and understandable introduction, we refer to [13]. In a few words, it produces a polynomial recurrence for the hypergeometric series or sums. In the following years, some authors also did some improvements on Zeilberger's algorithm.

Moreover, as with Gosper's original method or other algorithms, for Zeilberger's algorithm, the calculations can be done by the help of computers. Maple packages of the Zeilberger's algorithm have been written by Zeilberger [35] and Koornwinder [36]. Paule and Schorn [37] implemented Zeilberger's algorithm for the Mathematica system. Thus one can easily manage these types of sums or series by using the computer algebra systems without much time-consuming.

Now by using the Mathematica implementation of Zeilberger's algorithm, we will prove the Vandermonde's identity (2.5) (We refer to [37] for the guide of Mathematica package). Let's denote the RHS of (2.5) by S_k , then the algorithm produces the following recurrence relation

$$S_{k+1} = \frac{m+n-k}{k+1} S_k.$$

So after solving this relation by going backward, we obtain

$$S_{k+1} = \frac{(m+n-k)(m+n-k+1)\dots(m+n)}{(k+1)!}S_0.$$

Since $S_0 = 1$, the proof of (2.5) follows. For many other examples and the Mathematica package, see [38].

2.5.2 q-Zeilberger Algorithm

Since the q-hypergeometric series have lots of applications on many areas of mathematics, such as combinatorics, partition theory etc., people need similar mechanical methods for the q-hypergeometric series or sums as an extension of Zeilberger's algorithm.

Zeilberger also observed that his algorithm can be carried over to the q-hypergeometric cases. He and Wilf [39, 40] extended his algorithm for the q-hypergeometric series and sums and wrote Maple package for this algorithm. Furhermore, Koornwinder [36] wrote another Maple package for this algorithm. Afterwards, Paule and Riese [41] developed the Mathematica implementation of the q-Zeilberger algorithm. We refer to [41] for the user guide of this package and some applied examples.

The q-Zeilberger algorithm is the q-analogue of Zeilberger's algorithm. In other words, the q-Zeilberger algorithm performs some computations for q-hypergeometric series and sums as same as Zeilberger's algorithm does for the hypergeometric series and sums.

As we seen before, the sum on the RHS of the q-Vandermonde identity (2.11) is a q-hypergeometric sum. Now we shall apply this algorithm to this sum as an illustration by using Mathematica implementation. Denote the RHS of (2.11) by S_k . Then q-Zeilberger algorithm gives the recurrence relation

$$S_k = \frac{1 - q^{m+n-k+1}}{1 - q^k} S_{k-1}.$$

By solving this relation and the fact that $S_0 = 1$, we obtain

$$S_k = \frac{(q^{m+n-k+1}; q)_k}{(q; q)_k} = \begin{bmatrix} m+n \\ k \end{bmatrix}_q,$$

which is the LHS of (2.11). Thus we provide a computer based proof for the qVandermonde identity. We will use this algorithm to prove some of our results throughout the thesis.

It is worthwhile to mention that we encountered some cases in which although the summand term is q-hypergeometric, the q-Zeilberger algorithm does not work in general. This is interesting weakness of the q-Zeilberger algorithm. For this reason, we used different approaches to prove those identities.

Lastly, for the interested readers, we refer to [38] for the Mathematica packages of various symbolic computation methods and their user friendly guides.

3 LITERATURE REVIEW

In this section, firstly, we will present some basic definitions and notions about matrix theory and special kinds of matrix families. Afterwards, we will provide some historical background about these kinds of special matrix families.

3.1 About Matrices

Not only in every area of mathematics but also in fundamental sciences and engineering, somehow matrices occur and are used. Thus they have an important duty and have been studied for years.

As mentioned in Introduction, we will study some special combinatorial matrices in this thesis. Before giving previous works in the literature related to our thesis, we will recall some fundamental notions and the definitions of some special matrices, which we will study.

In general, we will obtain some algebraic properties of some combinatorial matrices such as LU-decomposition, Cholesky decomposition, inverse, determinant etc. Firstly, we would like to mention about these basic concepts and explain why these are important.

If the entries of a matrix are some combinatorial numbers such as Fibonacci numbers, binomial coefficients etc., then we call **combinatorial matrix**. Also in the literature, there is combinatorial matrix theory which investigates the combinatorial meanings of the matrices [42]. So the both differ from each other.

The **inverse** of a square matrix A is a matrix B if the equation

$$A \cdot B = B \cdot A = I$$
.

holds, where I is the identity matrix, whose (i, j)th entry is [i = j]. Here $[\cdot]$ denotes the Iverson notation which means:

$$[P] = \begin{cases} 1 & \text{if } P \text{ is true,} \\ 0 & \text{otherwise,} \end{cases}$$

where P is a statement that can be true or false. The inverse of the matrix A is denoted by A^{-1} . The inverse matrices are frequently used especially in solving linear system of equation and obtaining inverse transformations. There are some methods to

compute the inverse of a given matrix. It is worth mentioning that it is getting harder to identify the inverse matrix for the higher order matrix. So the matrices, whose inverses are explicitly known, are important.

It is often useful to summarize a multivariate phenomenen with a singular number. For the matrices, **determinant** is an example of this and it is denoted by $\det A$ or |A| for a given square matrix A. Another example of this is **permanent** and it is denoted by $\det A$. For more details and explanations about determinant and permanent, we refer to [43].

There are various different methods to evaluate the determinant of a matrix. Also in the literature, there are many determinant formulæ of some special matrices. Krattenthaler's surveys [44, 45] are elegant sources to find some of determinant evaluation methods and get the idea where we need determinants. Also they are comprehensive databases for some known determinant formulæ.

For a square matrix A, LU-decomposition refers to the factorization of A into two factors, a lower unit triangular matrix L and an upper triangular matrix U such that

$$A = L \cdot U$$
.

The LU-decomposition can be considered as the matrix form of Gaussian elimination. Computer algebra systems usually use it to solve square systems of linear equations. Also it helps to find the inverse matrix and compute the determinant of the matrix. For example, since $\det L = 1$ and U is a triangular matrix, we can easily compute the determinant of A by the formula $\det A = \det U$. On the other hand, if we know the inverse matrices L^{-1} and U^{-1} , then we may find a formula for the inverse matrix A^{-1} by the fact $A^{-1} = U^{-1}L^{-1}$. We will also use these advantages of the LU-decomposition to evaluate the determinant and inverse of a matrix.

There may not exist the LU-decomposition of any square matrix A. The matrix should satisfy some conditions to have LU-decomposition. However, there is an alternative decomposition by the help of the permutation matrix, which is a square matrix that has exactly one entry of 1 in each row and each column and 0's elsewhere. We refer to [46] for the necessary and sufficient conditions for the existence of the LU-decomposition and this alternative decomposition.

The **Cholesky decomposition** of a symmetric positive-definite matrix A, that is a matrix such that for every non-zero column vector x, $x^T A x > 0$, is a factorization of the form

$$A = C \cdot C^T, \tag{3.1}$$

where C is a lower triangular matrix and C^T is the transpose of the matrix C. The Cholesky decomposition is unique for a symmetric positive-definite matrix. It is possible to extend this definition for complex valued matrix by considering Hermitian matrix and conjugate transpose instead of symmetric matrix and transpose, respectively. In this thesis, we are always interested in real valued matrix. Note that when the matrix is not positive-definite but symmetric matrix, we still use the phrase "Cholesky decomposition" only to point out the relation (3.1). Cholesky decomposition has similar advantages with the LU-decomposition, but it is more efficient than the LU-decomposition. For example, the Cholesky decomposition is nearly twice as efficient as the LU-decomposition for solving systems of linear equations. Thus LU-decomposition and Cholesky decomposition help to simplify computations, both theoretically and practically.

For more details, explanations, examples and advantages about the notions given just above, we refer to [43, 47].

Nowadays, computer is one of the best friends of the scientists. Since the notions, mentioned just above, are many advantages in matrix theory, one desire to compute them easily and correctly. In the literature, there are many different methods and algorithms to evaluate them. Thus people need some special matrices, whose certain algebraic properties are explicitly known, to apply these methods and algorithms to see the accuracy and efficiency. These types of matrices are known as **test matrices**. Briefly, test matrices are key to test the accuracy of an algorithm or a method. In this thesis, we will provide many explicit formulæ for some algebraic properties of various special combinatorial matrices. We hope that some matrices, we studied, will be used as test matrices.

Before mentioning about some special matrix families, we would like to give some notations and remarks, we will regularly use from now on.

Remarks:

- 1) In general, the size of the matrices does not really matter, so that we may think about an infinite matrix M and restrict it whenever necessary to the first N rows, respectively, columns and use the notation M_N .
- 2) We denote the (k, j)th entry of a given matrix M and its inverse M^{-1} by M_{kj} and M_{kj}^{-1} , respectively. If the size of the matrix M is N, then we denote its (k, j)th entry by $(M_N)_{kj}$. Furthermore, $[M_{kj}]$, $[M_{kj}]_{0 \le k, j \le N-1}$ and $[M_{kj}]_{1 \le k, j \le N}$ mean an infinite matrix M, a matrix M of size N whose indexes start at 0 and a matrix M of size N whose indexes start at 1, respectively. Unless otherwise specified, we assumed that the indexes start at 1.
- 3) We use the letters L, U and A, B for the LU-decompositions of a given matrix and its inverse, respectively. Also the letter C is used for the Cholesky decomposition. For the related matrix to given matrix M, we may frequently use calligraphic letter M. In that cases, we apply the same representation to the factor matrices coming from LU-decomposition and Cholesky decomposition.
- 4) Since we will study many matrices, distinguishing them from each other is difficult. For this reason, the letter, which identifies the matrix, is only valid in the related subsection. In other words, we may use same letter for different matrices in different subsections.
- 5) We denote a sequence whose first term starts at the index 1 by $\{a_n\}$. Moreover, $\{a_n\}_{n\geq 0}$ stands for the sequence whose first term is a_0 .

Now, we will introduce some special matrix families, which we will encounter throughout the thesis.

3.1.1 Special Matrices

Diagonal Matrix

A diagonal matrix is a square matrix in which the entries outside of the main diagonal are all zero. The matrix $D(a_n) = [D_{kj}]$ stands for a diagonal matrix constructed via a given sequence $\{a_n\}$, and is defined by

$$D_{kj} = \begin{cases} a_k & \text{if } k = j, \\ 0 & \text{otherwise.} \end{cases}$$

Toeplitz Matrix

A **Toeplitz matrix** is a square matrix in which the entries on each descending diagonal from left to right are constant. It satisfies that $M_{kj} = M_{k+1,j+1}$ for all $k, j \geq 1$. In general, by considering this relation, the definition could be extended for the nonsquare matrices. As an example, any square Toeplitz matrix of size N is of the form:

$$M_N = \begin{bmatrix} a_0 & a_{-1} & a_{-2} & \cdots & \cdots & a_{-(N-1)} \\ a_1 & a_0 & a_{-1} & \ddots & & \vdots \\ a_2 & a_1 & \ddots & \ddots & \ddots & \vdots \\ \vdots & \ddots & \ddots & \ddots & a_{-1} & a_{-2} \\ \vdots & & \ddots & a_1 & a_0 & a_{-1} \\ a_{N-1} & \cdots & \cdots & a_2 & a_1 & a_0 \end{bmatrix}.$$

Band Matrix

A band matrix is a matrix whose nonzero entries are confined between an upper and a lower diagonal bands, comprising the main diagonal and zero outside. Formally, if M is a band matrix than there are nonnegative integers r and s such that

$$M_{kj} = 0 \text{ if } j < k - r \text{ or } j > k + s.$$

The quantities r and s are called the **lower bandwidth** and **upper bandwidth**, respectively. Moreover, the **bandwidth** of this band matrix is equal to r + s + 1. As an example, when r = 2, s = 3 and N = 6:

$$\begin{bmatrix} a_{11} & a_{12} & a_{13} & a_{14} & 0 & 0 \\ a_{21} & a_{22} & a_{23} & a_{24} & a_{25} & 0 \\ a_{31} & a_{32} & a_{33} & a_{34} & a_{35} & a_{36} \\ 0 & a_{42} & a_{43} & a_{44} & a_{45} & a_{46} \\ 0 & 0 & a_{53} & a_{54} & a_{55} & a_{56} \\ 0 & 0 & 0 & a_{64} & a_{65} & a_{66} \end{bmatrix},$$

where a_{ij} 's are arbitrary nonzero reals.

Diagonal matrices, upper and lower triangular matrices are most known examples of band matrices. When r = s = 1, then the corresponding band matrices are called the **tridiagonal matrices**.

Furthermore, kth diagonal band which is above (rsp. below) the main diagonal is called kth superdiagonal (rsp. kth subdiagonal).

Another important class of band matrices is the family of Toeplitz band matrices, which are both Toeplitz and band matrices.

Hankel Matrix

A **Hankel matrix** is a square matrix in which each ascending skew-diagonal from left to right is constant. This could be considered as an upside down Toeplitz matrix. For a given sequence $\{a_n\}_{n\geq 0}$, Hankel matrix is of the form

$$\begin{bmatrix} a_0 & a_1 & a_2 & \cdots \\ a_1 & a_2 & a_3 & \cdots \\ a_2 & a_3 & a_4 & \cdots \\ \vdots & \vdots & \vdots & \ddots \end{bmatrix}.$$

Formally, any Hankel matrix M is defined for a given sequence $\{a_n\}_{n\geq 0}$, as follows for $k, j \geq 0$,

$$M_{kj} = M_{jk} = a_{k+j-2}.$$

Hankel matrices have very important applications, especially in operator theory. For more details about them, see [48].

Considering some particular number sequences instead of $\{a_n\}_{n\geq 0}$, there are many special matrices with nice algebraic properties. Some authors also studied the Hankel matrix by considering the reciprocal sequence of $\{a_n\}_{n\geq 0}$ of the form

$$\begin{bmatrix} \frac{1}{a_0} & \frac{1}{a_1} & \frac{1}{a_2} & \dots \\ \frac{1}{a_1} & \frac{1}{a_2} & \frac{1}{a_3} & \dots \\ \frac{1}{a_2} & \frac{1}{a_2} & \frac{1}{a_4} & \dots \\ \vdots & \vdots & \vdots & \ddots \end{bmatrix}.$$

Some of the known examples of the Hankel matrices are Hilbert and Filbert matrices, which we will discuss in Section 3.2.4.

Hessenberg Matrix

An upper Hessenberg matrix has zero entries below the first subdiagonal, and a

lower Hessenberg matrix has zero entries above the first superdiagonal. The lower Hessenberg matrix H_N is of the form

$$H_{N} = \begin{bmatrix} h_{11} & h_{12} & & & & 0 \\ h_{21} & h_{22} & h_{23} & & & & \\ h_{31} & h_{32} & h_{33} & \ddots & & & \\ \vdots & \vdots & \vdots & \ddots & \ddots & & & \\ h_{N-1,1} & h_{N-1,2} & h_{N-1,3} & \cdots & \ddots & h_{N-1,N} \\ h_{N1} & h_{N2} & h_{N3} & \cdots & \cdots & h_{NN} \end{bmatrix}.$$
(3.2)

Similarly, the upper Hessenberg matrix of size N could be considered as transpose of the matrix H_N . In this thesis, we will study the lower Hessenberg matrices. Nevertheless, one can easily adapt our results to the upper Hessenberg matrices, as well. A triangular matrix is both lower and upper Hessenberg matrix. Moreover, a Hessenberg matrix is a band matrix whose one of the upper or lower bandwidth is 1.

Lehmer Matrix

The **Lehmer matrix** $[M_{kj}]$ (see [49]) is the symmetric matrix defined by

$$M_{kj} = \begin{cases} \frac{k}{j} & \text{if } j \ge k, \\ \frac{j}{k} & \text{if } j < k. \end{cases}$$

Equivalently, this may be written as

$$M_{kj} = \frac{\min(k, j)}{\max(k, j)}.$$

3.2 Previous Studies

In this subsection, we present some previous studies related to our results obtained in this thesis. We divide this subsection into different parts and each part includes some previous results about different matrix families. At the end of each part, we provide our motivations and what we will do in the following Results section. In other words, we briefly indicate our problems.

3.2.1 Band Matrices

Band matrices and their special cases such as Toeplitz band matrices, symmetric Toeplitz band matrices, especially tridiagonal matrices have been extensively studied by many authors [50, 1, 51, 52, 53, 54, 55]. These matrices arise in many areas of mathematics and its applications. Especially, the band matrices have many applications in numerical analysis. The matrices from finite element or finite difference problems are often banded. Tridiagonal matrices are used in telecommunication system analysis, for solving linear recurrence systems with non-constant coefficients, etc. For these reasons, the band matrices with known algebraic properties are important.

In 1972, for $r \geq 0$ and $1 \leq k, j \leq N$, Hoskins and Ponzo [1] defined the $N \times N$ symmetric Toeplitz band matrix $M_N = [M_{kj}]$ of bandwidth 2r + 1 via the binomial coefficients as

$$M_{kj} = (-1)^{r+k-j} \binom{2r}{r+k-j}.$$

For example, when r = 3 and N = 7, M_7 is of the form

$$M_7 = \begin{bmatrix} -20 & 15 & -6 & 1 & 0 & 0 & 0 \\ 15 & -20 & 15 & -6 & 1 & 0 & 0 \\ -6 & 15 & -20 & 15 & -6 & 1 & 0 \\ 1 & -6 & 15 & -20 & 15 & -6 & 1 \\ 0 & 1 & -6 & 15 & -20 & 15 & -6 \\ 0 & 0 & 1 & -6 & 15 & -20 & 15 \\ 0 & 0 & 0 & 1 & -6 & 15 & -20 \end{bmatrix}.$$

The authors gave formulæ for the determinant, the inverse matrix and the LU-decomposition of the matrix M_N . For example, they gave

$$\det M_N = (-1)^{N+r-1} \prod_{d=1}^{N} {2r+d-1 \choose r} {d+r-1 \choose r}^{-1}$$

and

$$(M_N^{-1})_{kj} = (-1)^r \binom{k+r-1}{r} \binom{j+r-1}{r}$$

$$\times \sum_{d=1}^N \binom{d+r-1-k}{r-1} \binom{d+r-1-j}{r} \binom{d+r-1}{r}^{-1} \binom{d+2r-1}{r}^{-1}.$$

The authors only considered the symmetric band matrix M_N with upper bandwidth r and lower bandwidth r.

It is worthwhile to note that a non-symmetric band matrix with upper bandwidth s and lower bandwidth r via the binomial coefficients has not been considered and studied

up to now. In Section 4.2, we will consider non-symmetric Toeplitz band matrix via the Gaussian q-binomial coefficients defined by for $k, j \geq 0$,

$$H_{kj} = (-1)^{r(k+j)+j} \mathbf{i}^{k(1+r-s)+j(1-r+s)-r(1-s-r)} q^{\frac{1}{2}(k-j)(k-j-r+s)-\frac{1}{2}rs} \begin{bmatrix} r+s\\r+j-k \end{bmatrix}_{q}.$$
(3.3)

We will obtain some algebraic properties of the matrix H. Also we will provide some further results derived from this matrix. Our results do not only generalize the results of Hoskins and Ponzo but also include new families of the band matrices.

3.2.2 Pascal Matrices

The Pascal matrices are defined via the binomial coefficients [56, 57]. They are mainly three kinds: the first is the left adjusted Pascal matrix P_N , the second is the right adjusted Pascal matrix Q_N and the third is the symmetric Pascal matrix S_N . They are defined by for $0 \le k, j < N$,

$$P_{kj} = {k \choose j}$$
, $Q_{kj} = {k \choose N-1-j}$ and $S_{kj} = {k+j \choose k}$,

respectively. In [58], the author studied the reciprocal of the symmetric Pascal matrix $\left[\binom{k+j}{k}^{-1}\right]_{k,j\geq 0}$ and its some parametric generalizations.

Recently, Prodinger [3] defined a matrix whose entries consist of the super Catalan numbers $\left\{\frac{(2i)!(2j)!}{i!j!(i+j)!}\right\}_{i,j\geq 0}$. He also defined its reciprocal analogue as well as their q-analogues whose (k,j)th entries are defined by

$$\binom{2k}{k}\binom{2j}{j}\binom{k+j}{k}^{-1}$$
 and $\binom{2k}{k}^{-1}\binom{2j}{j}^{-1}\binom{k+j}{k}$,

and

$$\begin{bmatrix} 2k \\ k \end{bmatrix}_a \begin{bmatrix} 2j \\ j \end{bmatrix}_a \begin{bmatrix} k+j \\ k \end{bmatrix}_a^{-1}$$
 and $\begin{bmatrix} 2k \\ k \end{bmatrix}_a^{-1} \begin{bmatrix} 2j \\ j \end{bmatrix}_a^{-1} \begin{bmatrix} k+j \\ k \end{bmatrix}_a$,

respectively. Then he gave some algebraic properties of these matrices.

In Section 4.3, we will study parametric generalizations of the just above matrices, introducing two additional parameters. We also mention how one can obtain further generalizations of these types of matrices.

3.2.3 Max and Min Matrices

In the literature, for some special sequences $\{a_n\}$, some authors studied the matrices $[\max(a_k, a_j)]_{1 \le k, j \le N}$ and $[\min(a_k, a_j)]_{1 \le k, j \le N}$. We listed them below:

• The author of [59] studied the matrix

$$[\max(N+1-k, N+1-j)]_{1 \le k, j \le N}$$

which is called the Franc matrix.

• The author of [60] gave the Cholesky decomposition of the matrix

$$\left[\frac{1}{\max(k+1,j+1)}\right]_{k,j>0},$$

which is called the loyal companion of the Hilbert matrix.

• In [61], the author found eigenvalues and eigenvectors of the matrices

$$[\min(k,j)]_{1 \leq k,j \leq N} \quad \text{and} \quad [\min(2k-1,2j-1)]_{1 \leq k,j \leq N} \,.$$

• Fonseca [62] studied the general cases of the matrices considered in [61] by defining the matrix $[\min(ak-b,aj-b)]_{1\leq k,j\leq N}$ for a>0 and $a\neq b$. Then he computed eigenvalues and eigenvectors of this general matrix by computing its inverse.

Recently, Mattila and Haukkanen [5] studied more general matrix families. Let $T = \{a_1, a_2, \ldots, a_N\}$ be a finite multiset of real numbers, such that $a_1 \leq a_2 \leq \cdots \leq a_N$. They considered the matrices $[\max(a_k, a_j)]_{1 \leq k, j \leq N}$ and $[\min(a_k, a_j)]_{1 \leq k, j \leq N}$ defined on the set T. Clearly, they may be written explicitly as

$$\begin{bmatrix} a_1 & a_2 & a_3 & \cdots & a_N \\ a_2 & a_2 & a_3 & \cdots & a_N \\ a_3 & a_3 & a_3 & \cdots & a_N \\ \vdots & \vdots & \vdots & \ddots & \vdots \\ a_N & a_N & a_N & \cdots & a_N \end{bmatrix} \text{ and } \begin{bmatrix} a_1 & a_1 & a_1 & \cdots & a_1 \\ a_1 & a_2 & a_2 & \cdots & a_2 \\ a_1 & a_2 & a_3 & \cdots & a_3 \\ \vdots & \vdots & \vdots & \ddots & \vdots \\ a_1 & a_2 & a_3 & \cdots & a_N \end{bmatrix},$$

respectively. They computed the determinants, inverses, Cholesky decompositions of these matrices and examined the positive definiteness of them. They used the meet and join matrices, see [63], as a tool to obtain their results. Moreover, they indicated that it is difficult to verify their results by using only basic linear algebra methods.

In Section 4.4, we will introduce more general families by considering the matrices $[a_{\max(k,j)}]$ and $[a_{\min(k,j)}]$, whose entries run in left-reversed and up-reversed L-shaped pattern, respectively, as well as their reciprocal analogues for an arbitrary sequence

 $\{a_n\}$. These matrices cover the all previous studied matrices and also generalize the results in [5]. We use basic linear algebra methods to prove our results, which also yield new and alternative proofs for the results in [5]. We will also present some interesting and useful applications of our results.

3.2.4 Hilbert, Filbert Matrices and Their Variants

The Hilbert matrix $H = [H_{kj}]$ is defined with entries

$$H_{kj} = \frac{1}{k+j-1}.$$

As a recursive analogue of the Hilbert matrix, Richardson [64] defined and studied the Filbert matrix $R = [R_{kj}]$ with entries

$$R_{kj} = \frac{1}{F_{k+j-1}},$$

where F_n is the *n*th Fibonacci number. Clearly, both the Hilbert and Filbert matrices are the examples of the Hankel matrices.

After the Filbert matrix, several generalizations and variants of it have been investigated and studied by some authors. We briefly summarize some of these:

- In [65], Kiliç and Prodinger studied a generalization of the Filbert matrix by defining the matrix $\left[\frac{1}{F_{k+j+r}}\right]$, where $r \geq -1$ is an arbitrary integer parameter.
- After this, Prodinger [66] defined a new generalization of the generalized Filbert matrix by introducing 3 additional parameters by taking its (k, j)th entry as $\frac{x^k y^j}{F_{\lambda(k+j)+r}}$, where $r \geq -1$ and $\lambda \geq 1$ are arbitrary integers and x, y are any reals.
- In another study [67], Kılıç and Prodinger obtained two variants of the generalized Filbert matrix by considering the matrices $\left[\frac{F_{\lambda(k+j)+r}}{F_{\lambda(k+j)+s}}\right]$ and $\left[\frac{L_{\lambda(k+j)+r}}{L_{\lambda(k+j)+s}}\right]$, where s, r and λ are integer parameters such that $s \neq r$, and $s \geq -1$ and $\lambda \geq 1$. The second matrix is the first instance, where the entries of the matrix include the Lucas numbers.
- Kılıç and Prodinger [68] gave a further generalization of the Filbert matrix by defining the matrix Q with entries

$$Q_{kj} = \frac{1}{F_{k+j+r}F_{k+j+r+1}\dots F_{k+j+r+d-1}},$$

where $r \geq -1$ and $d \geq 1$ are arbitrary integers. The generalized Filbert matrix is the particular case, when d = 1, of the matrix Q.

• In another paper [69], Kılıç and Prodinger introduced the matrix G as a parametric generalization of the matrix Q by

$$G_{kj} = \frac{1}{F_{\lambda(k+j)+r}F_{\lambda(k+j+1)+r}\dots F_{\lambda(k+j+d-1)+r}},$$

where $r \geq -1$, $d \geq 1$ and $\lambda \geq 1$ are integer parameters.

 Kılıç and Prodinger [70] gave new four variants of the Filbert matrix, by defining the matrices P, T, Y and Z with entries

$$P_{kj} = \frac{1}{F_{\lambda k + \mu j + r}}, \ T_{kj} = \frac{F_{\lambda k + \mu j + r}}{F_{\lambda k + \mu j + s}}, \ Y_{kj} = \frac{1}{L_{\lambda k + \mu j + r}} \text{ and } Z_{kj} = \frac{L_{\lambda k + \mu j + r}}{L_{\lambda k + \mu j + s}},$$

respectively, where s, r, λ and μ are integer parameters such that $s \neq r, r, s \geq -1$ and $\lambda, \mu \geq 1$. When $\lambda = \mu = 1$, the matrix Y is also known as generalized Lilbert matrix, which is the Lucas analogue of the generalized Filbert matrix.

• More recently, as the Lucas analogue of the matrix G, Kılıç and Prodinger [71] defined the matrix W by

$$W_{kj} = \frac{1}{L_{\lambda(k+j)+r} L_{\lambda(k+j+1)+r} \dots L_{\lambda(k+j+d-1)+r}},$$

where λ and r are arbitrary integers and d is a positive integer.

The authors of the all-above mentioned works have studied various properties of the given matrices such as LU and Cholesky decompositions, determinants, inverses, etc. All these results yield some further combinatorial identities, as well. In many of them, firstly the authors converted the entries of the matrices into q-forms and obtained related results for these q-forms. Afterwards, they proved all their claims in the q-forms by the means of the celebrated q-Zeilberger algorithm for the general parameter q. But only in [67, 70], q-Zeilberger algorithm did not work and because of that they used some traditional methods. We will encounter the same situation in Section 4.5.

In Section 4.5, we will introduce a new non-symmetric variant of the Filbert matrix defined by the entries for $k, j \geq 0$,

$$\frac{U_{\lambda k - \mu j + d}}{U_{\lambda k + \mu j + d}}$$

with positive integers λ , μ and d, where U_n stands for the nth generalized Fibonacci number. Note that the interesting feature of this matrix is that it includes some zero terms as entries. Specially, when $\lambda = \mu = 1$, then the entries on the dth superdiagonal are all zero. Furthermore, it would be never a symmetric matrix for any choice of the parameters. For this reason, we will also obtain related results for the transposed matrix.

If we look closely, the indexes of the Fibonacci or Lucas numbers in the Filbert or Lilbert matrix and all its generalizations or variants, studied before, are in the linear forms. Any nonlinear forms of the indexes have not been studied anywhere, yet. In Section 4.6, we will present a new generalization of the Filbert matrix whose indexes will be in the nonlinear form. This will be the first example in the literature. In brief, we will study the matrix as a nonlinear generalization of the Filbert matrix defined with the entries

$$\frac{1}{U_{\lambda(k+r)^n + \mu(j+s)^m + c}},$$

where λ , μ , n and m are positive integers, r, s and c are any integers such that $\lambda(k+r)^n + \mu(j+s)^m + c > 0$ for all positive integers k and j. Moreover, we will present its Lucas analogue.

3.2.5 Hessenberg Matrices

Hessenberg matrices were firstly investigated by Karl Hessenberg (1904-1959), a German engineer.

They are one of the most important matrices in numerical analysis [47, 72]. For example, the Hessenberg decomposition played an important role in computation of the matrix eigenvalues [47]. So in applied mathematics, they have important role.

In [73, 74], authors introduced a constructive way to compute the inverse of the finite and infinite Hessenberg matrices, respectively.

Note that we indicated that we would use N for the order of the matrices. However, in the sections about Hessenberg, matrices we prefer to use n rather than N. Because we will consider the value of the determinant of the matrices as the sequences indexed with their order.

Cahill et al. [75] gave a recurrence relation for the determinant of the matrix defined by (3.2) as follows for n > 0,

$$\det H_n = h_{nn} \det H_{n-1} + \sum_{r=1}^{n-1} \left((-1)^{n-r} h_{nr} \prod_{j=r}^{n-1} h_{j,j+1} \det H_{r-1} \right),$$

where $H_0 = 1$. Unfortunately, this result is not useful for the higher order Hessenberg matrices.

In [76, 77, 78, 79, 80], the authors gave the relationships between some certain recursive sequences and the determinants or permanents of some certain Hessenberg matrices. Meanwhile, some authors computed the determinants and permanents of various type of tridiagonal matrices which are indeed Hessenberg matrices [81, 82, 83, 84]. For example, in [83], Kılıç provided the formula

$$\begin{vmatrix} 2 & 1 & & & 0 \\ -1 & 2 & 1 & & & \\ & -1 & 2 & \ddots & & \\ & & \ddots & \ddots & 1 \\ 0 & & & -1 & 2 \end{vmatrix} = P_{n+1},$$

where P_n is the *n*th Pell number given in Table 1. The authors of the all works mentioned above used the cofactor expansion of the determinant as their main tool and then evaluated the determinants recursively.

The authors of [85] gave an algorithm to compute determinant of the Hessenberg matrices.

Moreover, the authors of [86, 87] evaluated the determinants of some special families of the Hessenberg matrices by using the combinatorial approaches.

Recently, Macfarlane [88] considered the Hessenberg matrix whose entries consist of

the terms of the sequence $\{W_n\}$:

$$M_{n} = \begin{bmatrix} W_{1} & W_{2} & W_{3} & \cdots & W_{n-2} & W_{n-1} & W_{n} \\ -x & W_{1} & W_{2} & \cdots & W_{n-3} & W_{n-2} & W_{n-1} \\ & -x & W_{1} & \cdots & W_{n-4} & W_{n-3} & W_{n-2} \\ & & \ddots & \ddots & \vdots & \vdots & \vdots \\ & & & \ddots & W_{1} & W_{2} & W_{3} \\ & & & -x & W_{1} & W_{2} \\ 0 & & & -x & W_{1} \end{bmatrix},$$

where $\{W_n\}$ is the Horadam sequence given in Table 1. Again by using the cofactor expansion of the determinant, he showed that the sequence $\{\det M_n\}$ satisfies the recurrence relation for n > 2,

$$\det M_n = (b + px) \det M_{n-1} - qx (a + x) \det M_{n-2}.$$

More recently, by using generating functions, Merca [89] showed that the determinant of an $n \times n$ Toeplitz-Hessenberg matrix is expressed as a sum over the integer partitions of n.

In the literature, Getu [7] firstly computed the determinants of a class of the Hessenberg matrices by using the generating functions. He considered the infinite matrix

$$R = \begin{bmatrix} b_0 & 1 & 0 & 0 & \dots \\ b_1 & c_1 & 1 & 0 & \dots \\ b_2 & c_2 & c_1 & 1 & \dots \\ b_3 & c_3 & c_2 & c_1 & \dots \\ b_4 & c_4 & c_3 & c_2 & \dots \\ \vdots & \vdots & \vdots & \vdots & \ddots \end{bmatrix}.$$

Then he showed that if the equation

$$A(x) = \frac{B(x)}{C(x) + 1}$$

holds then

$$a_n = (-1)^n \det R_n,$$

where A(x), B(x) and C(x) are the generating functions of the sequences $\{a_{n+1}\}_{n\geq 0}$, $\{b_n\}_{n\geq 0}$ and $\{c_{n+1}\}_{n\geq 0}$, respectively.

As it is seen, the determinants of the Hessenberg matrices have drawn attentions of many researchers. In Section 4.7, we will use the generating functions to determine the relationships between the determinants of the new three classes of the Hessenberg matrices whose entries are terms of the certain number sequences and the generating functions of these sequences. This method is more efficient and applicable than cofactor expansion and the determinants of many previously studied Hessenberg matrices are easily computed by this method. Furthermore, we will give an elegant method to evaluate the determinants of the Hessenberg matrices whose entries consist of the terms of the higher order linear recursive sequences.

4 RESULTS

In this section, we present the results which are obtained in our thesis. Firstly, in the following subsection we give some auxiliary results for further use. The next subsections are devoted to our main results. In each these subsections, we present some results for the different combinatorial matrix families.

4.1 Auxiliary Results

The following propositions are the general results about some matrix families. We use them as tools to prove some of our results. They may apply to other matrices which are not considered throughout the thesis, as well. All of them are new and useful results about some special matrix families.

Firstly, we shall start with a proposition about Toeplitz matrices.

Proposition 4.1. If M is a Toeplitz matrix of order N, then there exist the following relationships between the factor matrices coming from the LU-decompositions of the matrices M and M^{-1} , for $0 \le k, j \le N - 1$,

(i)
$$A_{kj} = L_{N-1-j,N-1-k}^{-1}$$
,

(ii)
$$A_{kj}^{-1} = L_{N-1-j,N-1-k}$$
,

(iii)
$$B_{kj} = U_{N-1-j,N-1-k}^{-1}$$
,

(iv)
$$B_{kj}^{-1} = U_{N-1-j,N-1-k}$$
,

(v)
$$M_{kj}^{-1} = M_{N-1-j,N-1-k}^{-1}$$
.

Proof. For the claims (i) and (ii), consider

$$\sum_{d=j}^{k} A_{kd} A_{dj}^{-1} = \sum_{d=j}^{k} L_{N-1-d,N-1-k}^{-1} L_{N-1-j,N-1-d}$$

$$= \sum_{d=N-1-k}^{N-1-j} L_{d,N-1-k}^{-1} L_{N-1-j,d} = [N-1-j,N-1-k] = [k=j],$$

which gives us $AA^{-1} = I$, as claimed. For the claims (iii) and (iv), we have

$$\sum_{d=k}^{j} B_{kd} B_{dj}^{-1} = \sum_{d=k}^{j} U_{N-1-d,N-1-k}^{-1} U_{N-1-j,N-1-d}$$

$$= \sum_{d=N-1-i}^{N-1-k} U_{d,N-1-k}^{-1} U_{N-1-j,d}^{-1} = [N-1-j,N-1-k] = [k=j],$$

as desired.

For the LU-decomposition of M^{-1} , we should show that $M^{-1} = A \cdot B$ or equivalently $M = B^{-1} \cdot A^{-1}$. So it is sufficient to show that

$$\sum_{\max(k,j) < d < N-1} B_{kd}^{-1} A_{dj}^{-1} = M_{kj}.$$

Thus consider

$$\sum_{\max(k,j) \le d \le N-1} B_{kd}^{-1} A_{dj}^{-1} = \sum_{\max(k,j) \le d \le N-1} U_{N-1-d,N-1-k} L_{N-1-j,N-1-d}$$

$$= \sum_{0 \le d \le N-1 - \max(k,j)} L_{N-1-j,d} U_{d,N-1-k}$$

$$= \sum_{0 \le d \le \min(N-1-j,N-1-k)} L_{N-1-j,d} U_{d,N-1-k}.$$

Since $M = L \cdot U$ and M is a Toeplitz matrix, we have $\sum_{0 \le d \le \min(k,j)} L_{kd} U_{dj} = M_{kj}$ and $M_{kj} = M_{N-1-j,N-1-k}$. Finally, we obtain

$$\sum_{\max(k,j) \le d \le N-1} B_{kd}^{-1} A_{dj}^{-1} = M_{N-1-j,N-1-k} = M_{kj},$$

which completes the proof. By the fact $M^{-1} = A \cdot B = U^{-1} \cdot L^{-1}$ and the relationships (i) and (iii), we have

$$\begin{split} M_{kj}^{-1} &= \sum_{0 \leq d \leq \min(k,j)} L_{N-1-d,N-1-k}^{-1} U_{N-1-j,N-1-d}^{-1} \\ &= \sum_{N-1-\min(k,j) \leq d \leq N-1} L_{d,N-1-k}^{-1} U_{N-1-j,d}^{-1} \\ &= \sum_{\max(N-1-j,N-1-k) \leq d \leq N-1} L_{d,N-1-k}^{-1} U_{N-1-j,d}^{-1} = M_{N-1-j,N-1-k}^{-1}. \end{split}$$

So the claim (v) follows.

By the above proposition, one can easily derive the LU-decomposition of the inverse of a Toeplitz matrix from its LU-decomposition.

The next proposition is about the matrices whose entries include separable factors with respect to the indexes k and j.

Let $H = [H_{kj}]$ be a square matrix, whose LU-decomposition, inverse, LU-decomposition of its inverse and Cholesky decomposition are known with the matrices $L = [L_{kj}]$, $U = [U_{kj}], H^{-1} = [H_{kj}^{-1}], A = [A_{kj}], B = [B_{kj}] \text{ and } C = [C_{kj}], \text{ respectively.}$

Proposition 4.2. Assume that $\mathcal{H} = [\mathcal{H}_{kj}]$ is a square matrix and there exist the sequences $\{s_n\}$ and $\{m_n\}$ with nonzero terms such that $\mathcal{H}_{kj} = \mathcal{H}_{kj}s_km_j$. Then for the matrix \mathcal{H} , one can determine the LU-decomposition, inverse, LU-decomposition of its inverse and Cholesky decomposition as shown

$$\mathcal{L}_{kj} = L_{kj} \frac{s_k}{s_j} \qquad and \qquad \mathcal{U}_{kj} = U_{kj} s_k m_j,$$

$$\mathcal{L}_{kj}^{-1} = L_{kj}^{-1} \frac{s_k}{s_j} \qquad and \qquad \mathcal{U}_{kj}^{-1} = U_{kj}^{-1} \frac{1}{s_j} \frac{1}{m_k},$$

$$\mathcal{H}_{kj}^{-1} = H_{kj}^{-1} \frac{1}{s_j} \frac{1}{m_k},$$

$$\mathcal{A}_{kj} = A_{kj} \frac{m_j}{m_k} \qquad and \qquad \mathcal{B}_{kj} = B_{kj} \frac{1}{s_j} \frac{1}{m_k},$$

$$\mathcal{A}_{kj}^{-1} = A_{kj}^{-1} \frac{m_j}{m_k} \qquad and \qquad \mathcal{B}_{kj}^{-1} = B_{kj}^{-1} s_k m_j$$

and when for all $k \geq 1$, $s_k = m_k$,

$$\mathcal{C}_{kj} = C_{kj} s_k.$$

Proof. By the assumption for the matrix \mathcal{H} , firstly we can write

$$\mathcal{H} = D(s_n) \cdot H \cdot D(m_n),$$

where $D(a_n)$ is the diagonal matrix defined as before. Since the LU-decomposition of the matrix H is known, namely $H = L \cdot U$, we can write

$$\mathcal{H} = D(s_n) \cdot L \cdot U \cdot D(m_n) = D(s_n) \cdot L \cdot D\left(\frac{1}{s_n}\right) \cdot D(s_n) \cdot U \cdot D(m_n).$$

Here we see that $D(s_n) \cdot L \cdot D\left(\frac{1}{s_n}\right)$ is a unite lower triangular matrix and $D(s_n) \cdot U \cdot D(m_n)$ is an upper triangular matrix. So we have

$$\mathcal{L} = D(s_n) \cdot L \cdot D\left(\frac{1}{s_n}\right) \text{ and } \mathcal{U} = D(s_n) \cdot U \cdot D(m_n),$$

which gives the LU-decomposition of the matrix \mathcal{H} . Moreover, by the rule of the inverse of the multiplication of the matrices, we may immediately derive

$$\mathcal{H}^{-1} = D\left(\frac{1}{m_n}\right) \cdot H^{-1} \cdot D\left(\frac{1}{s_n}\right). \tag{4.1}$$

The matrices \mathcal{A} and \mathcal{B} follow after applying the LU-decomposition to the matrix \mathcal{H}^{-1} . The relations of the inverse matrices $\mathcal{L}^{-1}, \mathcal{U}^{-1}, \mathcal{A}^{-1}$ and \mathcal{B}^{-1} can be easily derived as in (4.1). For the Cholesky decomposition of \mathcal{H} , consider

$$\mathcal{H} = D(s_n) \cdot H \cdot D(s_n) = D(s_n) \cdot C \cdot C^T \cdot D(s_n)^T = (D(s_n) \cdot C) (D(s_n) \cdot C)^T,$$

then the claim follows. \Box

This proposition is very useful to obtain new matrix identities. We will frequently use it in the forthcoming subsections.

Finally, we have the following proposition to derive the LU-decomposition of the transposed matrix.

Proposition 4.3. Let K be a nonsingular square matrix whose LU-decomposition is known with the matrices $L = [L_{kj}]$, $U = [U_{kj}]$, respectively. Then for the LU-decomposition of the transposed matrix K^T , we have

$$K^T = L' \cdot U',$$

where

$$L'_{kj} = \frac{U_{jk}}{U_{jj}}$$
 and $U'_{kj} = L_{jk}U_{kk}$.

Proof. Since K is nonsingular, for all $k \geq 1$, we have $U_{kk} \neq 0$. Then consider

$$K^T = U^T \cdot L^T = U^T \cdot D\left(\frac{1}{U_{nn}}\right) \cdot D(U_{nn}) \cdot L^T.$$

Then $L' = U^T \cdot D(\frac{1}{U_{nn}})$ and $U' = D(U_{nn}) \cdot L^T$, which completes the proof.

We have the following useful corollary.

Corollary 4.1. Let S be a nonsingular symmetric matrix. Then S can be written as

$$S = U^T \cdot D\left(\frac{1}{U_{nn}}\right) \cdot U,$$

where U is the factor matrix coming from the LU-decomposition of the matrix S. Furthermore, the Cholesky decomposition of S is derived as

$$C_{kj} = U_{jk} \frac{1}{\sqrt{U_{jj}}}.$$

Proof. Since S is a nonsingular symmetric matrix, by Proposition 4.3, the factor matrix L of S is equal to $U^T \cdot D\left(\frac{1}{U_{nn}}\right)$. So the first claim follows. Consider

$$S = U^T \cdot D\left(\frac{1}{U_{nn}}\right) \cdot U = U^T \cdot D\left(\frac{1}{\sqrt{U_{nn}}}\right) \cdot D\left(\frac{1}{\sqrt{U_{nn}}}\right) \cdot U$$
$$= U^T \cdot D\left(\frac{1}{\sqrt{U_{nn}}}\right) \cdot \left(U^T \cdot D\left(\frac{1}{\sqrt{U_{nn}}}\right)\right)^T.$$

Thus $C = U^T \cdot D\left(\frac{1}{\sqrt{U_{nn}}}\right)$, which completes the proof.

The above corollary allows us to derive the Cholesky decomposition of a matrix from its LU-decomposition if it is symmetric. Remind that the Cholesky decomposition means that the matrix satisfies the relation (3.1). If the matrix S is a positive definite, i.e. for all n > 0, $U_{nn} > 0$, then our result is valid for the general theory and also provides an alternative proof for the fact that the matrices comes from the Cholesky decomposition of a positive definite matrix consist of real entries.

Now we are ready to move to our main results.

4.2 A Family of the Non-Symmetric Band Matrices

As mentioned in Section 3.2.1, in this section we introduce a class of non-symmetric Toeplitz band matrices with upper bandwidth s and lower bandwidth r whose entries are defined via the Gaussian q-binomial coefficients to obtain the generalizations of the results of [1]. The case s = r gives us the q-analogue of the result of [1]. When s = r + 1, we have a Toeplitz band matrix with even number of bands, which has not been studied before.

Briefly, we define a matrix H with bandwidth r + s + 1 via the Gaussian q-binomial coefficients. We provide explicit formulæ for the LU-decomposition, determinant and LU-decomposition of the inverse matrix H^{-1} . Furthermore, we derive some complementary results for the work [1] related to the case of bandwidth r+s+1. We presented obtained results in [2].

Our main tool is usually to guess relevant quantities. Then we use the q-Zeilberger algorithm to prove our claims. All identities we obtain hold for the general quantity q, so that results about the Fibonomial coefficients come out as corollaries for the special

choice of q. Finally, by the help of the limit $q \to 1$, we derive further results including the usual binomial coefficients.

For nonnegative arbitrary integers r and s, we define the matrix $H = [H_{kj}]_{k,j\geq 0}$ with upper bandwidth s and lower bandwidth r by

$$H_{kj} = (-1)^{r(k+j)+j} \mathbf{i}^{k(1+r-s)+j(1-r+s)-r(1-s-r)} q^{\frac{1}{2}(k-j)(k-j-r+s)-\frac{1}{2}rs} \begin{bmatrix} r+s \\ r+j-k \end{bmatrix}_{q}.$$
(4.2)

For example, when r=2 and s=4, the matrix H is of the form

$$\begin{bmatrix} -q^{-4} \begin{bmatrix} 6 \\ 2 \end{bmatrix}_{q} & -\mathbf{i}q^{-\frac{9}{2}} \begin{bmatrix} 6 \\ 3 \end{bmatrix}_{q} & q^{-4} \begin{bmatrix} 6 \\ 4 \end{bmatrix}_{q} & \mathbf{i}q^{-\frac{5}{2}} \begin{bmatrix} 6 \\ 5 \end{bmatrix}_{q} & -1 & 0 \\ \mathbf{i}q^{-\frac{5}{2}} \begin{bmatrix} 6 \\ 1 \end{bmatrix}_{q} & -q^{-4} \begin{bmatrix} 6 \\ 2 \end{bmatrix}_{q} & -\mathbf{i}q^{-\frac{9}{2}} \begin{bmatrix} 6 \\ 3 \end{bmatrix}_{q} & q^{-4} \begin{bmatrix} 6 \\ 4 \end{bmatrix}_{q} & \mathbf{i}q^{-\frac{5}{2}} \begin{bmatrix} 6 \\ 5 \end{bmatrix}_{q} & \ddots \\ 1 & \mathbf{i}q^{-\frac{5}{2}} \begin{bmatrix} 6 \\ 1 \end{bmatrix}_{q} & -q^{-4} \begin{bmatrix} 6 \\ 2 \end{bmatrix}_{q} & \ddots & \ddots & \ddots & \mathbf{i}q^{-\frac{5}{2}} \begin{bmatrix} 6 \\ 5 \end{bmatrix}_{q} \\ & \ddots & \ddots & \ddots & \mathbf{i}q^{-\frac{5}{2}} \begin{bmatrix} 6 \\ 4 \end{bmatrix}_{q} \\ & \ddots & \ddots & \ddots & \mathbf{i}q^{-\frac{9}{2}} \begin{bmatrix} 6 \\ 3 \end{bmatrix}_{q} \\ 0 & 1 & \mathbf{i}q^{-\frac{5}{2}} \begin{bmatrix} 6 \\ 1 \end{bmatrix}_{q} & -q^{-4} \begin{bmatrix} 6 \\ 2 \end{bmatrix}_{q} \end{bmatrix}$$

When $q = \beta/\alpha$, where $\alpha, \beta = (p \mp \sqrt{p^2 + 4})/2$, we get the Fibonomial analogue of the matrix H and denote it by $\mathcal{H} = [\mathcal{H}_{kj}]_{k,j \geq 0}$, where

$$\mathcal{H}_{kj} = (-1)^{r(k+j)+j(k+1)} \mathbf{i}^{j(j+1)+k(k+1)+r(r-1)} \begin{Bmatrix} r+s \\ r+j-k \end{Bmatrix}_{U}.$$

For r=2 and s=4, we have

$$\mathcal{H} = \begin{bmatrix} -\{^{6}_{2}\}_{U} & -\{^{6}_{3}\}_{U} & \{^{6}_{4}\}_{U} & \{^{6}_{5}\}_{U} & -1 & 0 \\ \{^{6}_{1}\}_{U} & -\{^{6}_{2}\}_{U} & -\{^{6}_{3}\}_{U} & \{^{6}_{4}\}_{U} & \{^{6}_{5}\}_{U} & \ddots \\ 1 & \{^{6}_{1}\}_{U} & -\{^{6}_{2}\}_{U} & \ddots & \ddots & \ddots & -1 \\ & & \ddots & \ddots & \ddots & \ddots & \{^{6}_{5}\}_{U} \\ & & & \ddots & \ddots & \ddots & \ddots & \{^{6}_{4}\}_{U} \\ & & & & \ddots & \ddots & \ddots & -\{^{6}_{3}\}_{U} \\ 0 & & & & 1 & \{^{6}_{1}\}_{U} & -\{^{6}_{2}\}_{U} \end{bmatrix}. \tag{4.3}$$

Before giving main results, it is worthwhile to note that one may ignore some power terms which are separable with respect to the indexes by the help of Proposition 4.2. We prefer this way because as it is seen in (4.3), the sign pattern from the lowest subdiagonal " $++--++--\cdots$ " looks nicer.

4.2.1 Main Results

Now we shall start with the matrices L and U as well as their inverses.

Theorem 4.1. For $k, j \geq 0$,

$$L_{kj} = (-1)^{r(k+j)+j} \mathbf{i}^{k+j+(r-s)(k-j)} q^{\frac{1}{2}(k-j)(k-j+s-r)} \begin{bmatrix} r \\ k-j \end{bmatrix}_q \begin{bmatrix} k \\ k-j \end{bmatrix}_q \begin{bmatrix} s+k \\ k-j \end{bmatrix}_q^{-1},$$

$$L_{kj}^{-1} = (-1)^{r(k+j)} \mathbf{i}^{(k-j)(r-1-s)} q^{\frac{1}{2}(k-j)(s-r+1)} \begin{bmatrix} k-j+r-1 \\ r-1 \end{bmatrix}_q \begin{bmatrix} s+j \\ j \end{bmatrix}_q \begin{bmatrix} s+k \\ k \end{bmatrix}_q^{-1},$$

$$U_{kj} = (-1)^{r(k+j)+j} \mathbf{i}^{k(1+r-s)+j(1-r+s)-r(1-s-r)} q^{\frac{1}{2}(k-j)(k-j-r+s)-\frac{1}{2}rs}$$

$$\times \begin{bmatrix} s \\ j-k \end{bmatrix}_q \begin{bmatrix} r+s+k \\ r+j \end{bmatrix}_q \begin{bmatrix} s+k \\ j \end{bmatrix}_q^{-1}$$

and

$$\begin{split} U_{kj}^{-1} &= (-1)^{(k+j)(r+1)} \mathbf{i}^{k-j-r+(k+r-j)(r-s)} q^{\frac{1}{2}(k-j)(s-r-1)+\frac{1}{2}rs} \\ &\times \begin{bmatrix} j-k+s-1 \\ s-1 \end{bmatrix}_q \begin{bmatrix} r+k \\ k \end{bmatrix}_q \begin{bmatrix} r+s+j \\ r \end{bmatrix}_q^{-1}. \end{split}$$

As their Fibonomial analogues, for the matrix \mathcal{H} , we have the following corollary.

Corollary 4.2. For k, j > 0,

$$\mathcal{L}_{kj} = (-1)^{r(k+j)+j(k+1)} \mathbf{i}^{k(k+1)+j(j+1)} \begin{Bmatrix} r \\ k-j \end{Bmatrix}_{U} \begin{Bmatrix} k \\ k-j \end{Bmatrix}_{U} \begin{Bmatrix} s+k \\ k-j \end{Bmatrix}_{U},$$

$$\mathcal{L}_{kj}^{-1} = (-1)^{r(k+j)} \begin{Bmatrix} k-j+r-1 \\ r-1 \end{Bmatrix}_{U} \begin{Bmatrix} s+j \\ j \end{Bmatrix}_{U} \begin{Bmatrix} s+k \\ k \end{Bmatrix}_{U}^{-1},$$

$$\mathcal{U}_{kj} = (-1)^{r(k+j)+j(k+1)} \mathbf{i}^{j(j+1)+k(k+1)+r(r-1)} \begin{Bmatrix} s \\ j-k \end{Bmatrix}_{U} \begin{Bmatrix} r+s+k \\ r+j \end{Bmatrix}_{U} \begin{Bmatrix} s+k \\ j \end{Bmatrix}_{U}^{-1}$$

and

$$\mathcal{U}_{kj}^{-1} = (-1)^{(j+k)(r+1)} \mathbf{i}^{r(r-1)} \begin{Bmatrix} j - k + s - 1 \\ s - 1 \end{Bmatrix}_{U} \begin{Bmatrix} r + k \\ k \end{Bmatrix}_{U} \begin{Bmatrix} r + s + j \\ r \end{Bmatrix}_{U}^{-1}.$$

Consequently, we could give the values of the determinants of the matrices H_N and \mathcal{H}_N . They are simply evaluated as the products of the main diagonal entries of the upper triangular matrices U and \mathcal{U} , respectively.

Theorem 4.2. For $N \geq 1$, we have

$$\det H_N = \mathbf{i}^{r(r+s-1)N} q^{-\frac{1}{2}Nrs} \prod_{d=0}^{N-1} \begin{bmatrix} r+s+d \\ d \end{bmatrix}_q \begin{bmatrix} s+d \\ d \end{bmatrix}_q^{-1}.$$

As the Fibonomial analogue, we have

Corollary 4.3. For $N \geq 1$,

$$\det \mathcal{H}_N = \mathbf{i}^{r(r-1)N} \prod_{d=0}^{N-1} \begin{Bmatrix} r+s+d \\ r+d \end{Bmatrix}_U \begin{Bmatrix} s+d \\ d \end{Bmatrix}_U^{-1}.$$

Specially, when r = s = 3, we have the following nice formula

$$\det \mathcal{H}_N = (-1)^N \frac{U_{N+1}}{U_1} \frac{U_{N+2}^2}{U_2^2} \frac{U_{N+3}^3}{U_3^3} \frac{U_{N+4}^2}{U_4^2} \frac{U_{N+5}}{U_5}.$$

Now recall a result from [1]. For r = 3, the determinant of the matrix they studied is equal to

$$\frac{N+1}{1} \frac{(N+2)^2}{2^2} \frac{(N+3)^3}{3^3} \frac{(N+4)^2}{4^2} \frac{N+5}{5}.$$

As it is seen, there is a remarkable similarity.

Moreover, when r = 5 and s = 4, we have

$$\det \mathcal{H}_N = \frac{U_{N+1}}{U_1} \frac{U_{N+2}^2}{U_2^2} \frac{U_{N+3}^3}{U_3^3} \frac{U_{N+4}^4}{U_4^4} \frac{U_{N+5}^4}{U_5^4} \frac{U_{N+6}^3}{U_6^3} \frac{U_{N+7}^2}{U_7^2} \frac{U_{N+8}}{U_8}.$$

For the inverse matrix H^{-1} , unfortunately there isn't any explicit formula. Nevertheless by the LU-decomposition, we get the following theorem.

Theorem 4.3. For $0 \le k, j \le N - 1$,

$$(H_N^{-1})_{kj} = (-1)^{r(j+k)+k} \mathbf{i}^{k(1+r-s)+j(1-r+s)+r(r-s-1)} q^{\frac{1}{2}((k-j)(s-r)+rs-j-k)} \begin{bmatrix} r+k \\ k \end{bmatrix}_q \begin{bmatrix} s+j \\ j \end{bmatrix}_q \\ \times \sum_{d=0}^{N-1} q^d \begin{bmatrix} d-k+s-1 \\ s-1 \end{bmatrix}_q \begin{bmatrix} d-j+r-1 \\ r-1 \end{bmatrix}_q \begin{bmatrix} r+s+d \\ r \end{bmatrix}_q^{-1} \begin{bmatrix} s+d \\ d \end{bmatrix}_q^{-1}.$$

Although there is no closed formula for the inverse matrix, we may express it in another way. The following theorem helps us to express the matrix H_N^{-1} by its LU-decomposition and moreover we can explicitly find the inverses of these factor matrices.

Theorem 4.4. For $0 \le k, j \le N - 1$,

$$A_{kj} = (-1)^{r(k+j)} \mathbf{i}^{(k-j)(r-1-s)} q^{\frac{1}{2}(k-j)(s-r+1)} \times \begin{bmatrix} k-j+r-1 \\ r-1 \end{bmatrix}_q \begin{bmatrix} s+N-1-k \\ s \end{bmatrix}_q \begin{bmatrix} s+N-1-j \\ s \end{bmatrix}_q^{-1},$$

$$A_{kj}^{-1} = (-1)^{r(k+j)+j} \mathbf{i}^{k+j+(r-s)(k-j)} q^{\frac{1}{2}(k-j)(k-j+s-r)}$$

$$\times \begin{bmatrix} r \\ k-j \end{bmatrix}_q \begin{bmatrix} N-1-j \\ k-j \end{bmatrix}_q \begin{bmatrix} s+N-1-j \\ k-j \end{bmatrix}_q^{-1},$$

$$B_{kj} = (-1)^{(k+j)(r+1)} \mathbf{i}^{k-j-r+(k+r-j)(r-s)} q^{\frac{1}{2}(k-j)(s-r-1)+\frac{1}{2}rs}$$

$$\times \begin{bmatrix} j-k+s-1 \\ s-1 \end{bmatrix}_q \begin{bmatrix} r+N-1-j \\ r \end{bmatrix}_q \begin{bmatrix} r+s+N-1-k \\ r \end{bmatrix}_q^{-1}$$

and

$$\begin{split} B_{kj}^{-1} &= (-1)^{r(k+j)+j} \mathbf{i}^{k(1+r-s)+j(1-r+s)-r(1-s-r)} q^{\frac{1}{2}(k-j)(k-j-r+s)-\frac{1}{2}rs} \\ &\times \begin{bmatrix} s \\ j-k \end{bmatrix}_q \begin{bmatrix} r+s+N-1-j \\ s+k-j \end{bmatrix}_q \begin{bmatrix} s+N-1-j \\ s+k-j \end{bmatrix}_q^{-1}. \end{split}$$

As the Fibonomial analogue, we have

Corollary 4.4. For $0 \le k, j \le N-1$,

$$\mathcal{A}_{kj} = (-1)^{(k+j)r} \begin{Bmatrix} k - j + r - 1 \\ r - 1 \end{Bmatrix}_{U} \begin{Bmatrix} s + N - 1 - k \\ s \end{Bmatrix}_{U} \begin{Bmatrix} s + N - 1 - j \\ s \end{Bmatrix}_{U}^{-1},$$

$$\begin{split} \mathcal{A}_{kj}^{-1} &= (-1)^{r(k+j)+j(k+1)} \mathbf{i}^{k(k+1)+j(j+1)} \\ &\times \left\{ \begin{matrix} r \\ k-j \end{matrix} \right\}_{U} \! \left\{ \begin{matrix} N-1-j \\ k-j \end{matrix} \right\}_{U} \! \left\{ \begin{matrix} s+N-1-j \\ k-j \end{matrix} \right\}_{U}^{-1}, \end{split}$$

$$\mathcal{B}_{kj} = (-1)^{(j+k)(r+1)} \mathbf{i}^{r(r-1)} \begin{Bmatrix} j - k + s - 1 \\ s - 1 \end{Bmatrix}_{U} \times \begin{Bmatrix} r + N - 1 - j \\ r \end{Bmatrix}_{U} \begin{Bmatrix} r + s + N - 1 - k \\ r \end{Bmatrix}_{U}^{-1}$$

and

$$\mathcal{B}_{kj}^{-1} = (-1)^{r(k+j)+j(k+1)} \mathbf{i}^{j(j+1)+k(k+1)+r(r-1)} \times \begin{Bmatrix} s \\ j-k \end{Bmatrix}_{U} \begin{Bmatrix} r+s+N-1-j \\ s+k-j \end{Bmatrix}_{U} \begin{Bmatrix} s+N-1-j \\ s+k-j \end{Bmatrix}_{U}.$$

Specially, when the case p = 1, i.e. $U_n = F_n$, our results become valid for the usual Fibonomial coefficients.

4.2.2 Proofs

Now we will give the proofs of our main results.

To show $H = L \cdot U$, it is sufficient to prove that the following equation holds.

$$\sum_{0 \le d \le \min(k,j)} L_{kd} U_{dj} = H_{kj}.$$

Thus we need to show

$$\begin{split} &\sum_{0 \leq d \leq \min(k,j)} (-1)^{r(k+d)+d} \mathbf{i}^{(d+k)+(r-s)(k-d)} q^{\frac{1}{2}(k-d)(k-d+s-r)} \begin{bmatrix} r \\ k-d \end{bmatrix}_q \begin{bmatrix} k \\ k-d \end{bmatrix}_q \\ &\times \begin{bmatrix} s+k \\ k-d \end{bmatrix}_q^{-1} (-1)^{r(d+j)+j} \mathbf{i}^{d(1+r-s)+j(1-r+s)-r(1-s-r)} \\ &\times q^{\frac{1}{2}(d-j)(d-j-r+s)-\frac{1}{2}rs} \begin{bmatrix} s \\ j-d \end{bmatrix}_q \begin{bmatrix} r+s+d \\ r+j \end{bmatrix}_q \begin{bmatrix} s+d \\ j \end{bmatrix}_q^{-1} \\ &= (-1)^{r(k+j)+j} \mathbf{i}^{k(1+r-s)+j(1-r+s)-r(1-s-r)} q^{\frac{1}{2}(k-j)(k-j-r+s)-\frac{1}{2}rs} \begin{bmatrix} r+s \\ r+j-k \end{bmatrix}_q. \end{split}$$

After some simplifications, we have the following equation to prove

$$\begin{split} &\sum_{j-s \leq d \leq k} q^{-d(k+j)+d^2} \begin{bmatrix} r \\ k-d \end{bmatrix}_q \begin{bmatrix} k \\ k-d \end{bmatrix}_q \begin{bmatrix} s+k \\ k-d \end{bmatrix}_q^{-1} \begin{bmatrix} s \\ j-d \end{bmatrix}_q \begin{bmatrix} r+s+d \\ r+j \end{bmatrix}_q \begin{bmatrix} s+d \\ j \end{bmatrix}_q^{-1} \\ &= q^{-jk} \begin{bmatrix} r+s \\ r+j-k \end{bmatrix}_q. \end{split}$$

Let's denote the LHS of the above equation by SUM_k . Then the Mathematica package of the q-Zeilberger algorithm produces the recursion

$$\mathrm{SUM}_k = \frac{q^{-j}(1-q^{1+j-k+r})}{(1-q^{-j+k+s})}\mathrm{SUM}_{k-1}.$$

By going backward, we obtain

$$\mathtt{SUM}_k = \frac{q^{-jk}(1-q^{1+j-k+r})\cdots(1-q^{r+j})}{(1-q^{s-j+k})\cdots(1-q^{s-j+1})}\mathtt{SUM}_0,$$

where $SUM_0 = {r+s \brack r+j}_q$. After multiplying both the denominator and numerator of the above equation with $(q;q)_{j-k+r}$, we get

$$\mathrm{SUM}_k = q^{-jk} \begin{bmatrix} r+s \\ r+j-k \end{bmatrix}_q,$$

as claimed. So the proof of the LU-decomposition of the matrix H is completed.

Now we turn the inverse matrix L^{-1} . Since L and L^{-1} are lower triangular matrices, we only need to look at the entries indexed by (k, j) with $k \geq j$. So we should show that

$$\sum_{j \le d \le k} L_{kd} L_{dj}^{-1} = [k = j].$$

Then we have

$$\begin{split} & \sum_{j \leq d \leq k} L_{kd} L_{dj}^{-1} = (-1)^{r(k+j)} \mathbf{i}^{k+j+(k-j)(r-s)} q^{\frac{1}{2} \left(k^2 - j + (r-s)(j-k)\right)} \begin{bmatrix} s+j \\ j \end{bmatrix}_q \\ & \times \sum_{j < d \leq k} (-1)^d q^{\frac{1}{2} (d^2 + d) - kd} \begin{bmatrix} r \\ k - d \end{bmatrix}_q \begin{bmatrix} k \\ d \end{bmatrix}_q \begin{bmatrix} s+k \\ k - d \end{bmatrix}_q^{-1} \begin{bmatrix} d-j+r-1 \\ r-1 \end{bmatrix}_q \begin{bmatrix} s+d \\ d \end{bmatrix}_q^{-1}. \end{split}$$

The q-Zeilberger algorithm computes the sum on the RHS of the above equation as 0 when $k \neq j$ and $r \neq 0$. For the case r = 0, H is an upper triangular matrix so that the claim is clear. For the case k = j, it is easy to see that $L_{kk}L_{kk}^{-1} = 1$. So the proof is completed.

Since U and U^{-1} are upper triangular matrices, we just need to look at the entries indexed by (k, j) with $j \geq k$. Thus we have

$$\sum_{k \le d \le j} U_{kd} U_{dj}^{-1} = (-1)^{r(j+k+r-1)+j} \mathbf{i}^{(k-j)(r-s+1)} q^{\frac{1}{2}((r-s)(j-k)+k^2+j)} \begin{bmatrix} r+s+j \\ r \end{bmatrix}_q^{-1}$$

$$\times \sum_{k \le d \le j} (-1)^d q^{\frac{1}{2}(d^2-d)-kd} \begin{bmatrix} s \\ d-k \end{bmatrix}_q \begin{bmatrix} r+s+k \\ s-d+k \end{bmatrix}_q$$

$$\times \begin{bmatrix} s+k \\ d \end{bmatrix}_q^{-1} \begin{bmatrix} j-d+r-1 \\ r-1 \end{bmatrix}_q \begin{bmatrix} r+d \\ r \end{bmatrix}_q.$$

Similarly, the q-Zeilberger algorithm computes the sum on the RHS of the above equation as 0 when $k \neq j$ and $s \neq 0$. When we choose the number of superdiagonals of the matrix H as zero, that is the case s = 0, it is easy to check because the matrix H is a lower triangular matrix. If k = j, it is obvious that $U_{kk}U_{kk}^{-1} = 1$. Finally

$$\sum_{k \le d \le j} U_{kd} U_{dj}^{-1} = [k = j],$$

so the proof of Theorem 4.1 is completed.

For the inverse matrix H_N^{-1} , by using the fact that $H_N^{-1} = U_N^{-1} \cdot L_N^{-1}$, we can write

$$(H_N^{-1})_{kj} = \sum_{d=0}^{N-1} U_{kd}^{-1} L_{dj}^{-1}$$

$$\begin{split} &= \sum_{d=0}^{N-1} (-1)^{(k+d)(r+1)} \mathbf{i}^{(k-d-r)+(k+r-d)(r-s)} q^{\frac{1}{2}((k-d)(s-r-1)+rs)} \\ &\times \begin{bmatrix} d-k+s-1 \\ s-1 \end{bmatrix}_q \begin{bmatrix} r+k \\ k \end{bmatrix}_q \begin{bmatrix} r+s+d \\ r \end{bmatrix}_q^{-1} (-1)^{r(d+j)} \mathbf{i}^{(d-j)(r-1-s)} \\ &\times q^{\frac{1}{2}(d-j)(s-r+1)} \begin{bmatrix} d-j+r-1 \\ r-1 \end{bmatrix}_q \begin{bmatrix} s+j \\ j \end{bmatrix}_q \begin{bmatrix} s+d \\ d \end{bmatrix}_q^{-1}. \end{split}$$

After some straightforward simplifications, we obtain Theorem 4.3.

Since the matrix H is a Toeplitz matrix, Theorem 4.4 arises as a consequence of Proposition 4.1.

Thus the proofs of all theorems are completed for the general real parameter q. The proofs of all corollaries follow by choosing $q = \beta/\alpha$.

4.2.3 The Case Bandwidth r + s + 1 with the Binomial Coefficients

The results will be presented in this subsection are direct generalizations of the results of [1] with upper bandwidth s and lower bandwidth r. The results for the case r = s cover the results in [1].

For $r, s \geq 0$, we define non-symmetric Toeplitz band matrix $G = [G_{kj}]_{k,j\geq 0}$ via the binomial coefficients as

$$G_{kj} = (-1)^{k+j+r} \binom{r+s}{r+j-k}.$$

For example, when r = 2, s = 4 and N = 7, we have:

$$G_7 = \begin{bmatrix} 15 & -20 & 15 & -6 & 1 & 0 & 0 \\ -6 & 15 & -20 & 15 & -6 & 1 & 0 \\ 1 & -6 & 15 & -20 & 15 & -6 & 1 \\ 0 & 1 & -6 & 15 & -20 & 15 & -6 \\ 0 & 0 & 1 & -6 & 15 & -20 & 15 \\ 0 & 0 & 0 & 1 & -6 & 15 & -20 \\ 0 & 0 & 0 & 0 & 1 & -6 & 15 \end{bmatrix}.$$

We list the results related to the LU-decomposition, inverse matrices L^{-1} and U^{-1} and determinant of the matrix G, respectively.

Theorem 4.5. For $k, j \geq 0$,

$$L_{kj} = (-1)^{k+j} \binom{r}{k-j} \binom{k}{j} \binom{s+k}{k-j}^{-1},$$

$$U_{kj} = (-1)^{k+j+r} {s \choose j-k} {r+s+k \choose r+j} {s+k \choose j}^{-1},$$

$$L_{kj}^{-1} = {k-j+r-1 \choose r-1} {s+j \choose j} {s+k \choose k}^{-1}$$

and

$$U_{kj}^{-1} = (-1)^r \binom{j-k+s-1}{s-1} \binom{r+k}{k} \binom{r+s+j}{r}^{-1}.$$

Theorem 4.6. For $N \ge 1$,

$$\det G_N = (-1)^{rn} \prod_{d=0}^{N-1} \binom{r+s+d}{r+d} \binom{s+d}{d}^{-1}.$$

Similarly, we have the following result for the LU-decomposition of the inverse matrix G_N^{-1} .

Theorem 4.7. For $0 \le k, j \le N - 1$,

$$A_{kj} = \binom{k-j+r-1}{r-1} \binom{s+N-1-k}{s} \binom{s+N-1-j}{s}^{-1},$$

$$A_{kj}^{-1} = (-1)^{k+j} \binom{r}{k-j} \binom{N-1-j}{k-j} \binom{s+N-1-j}{k-j}^{-1},$$

$$B_{kj} = (-1)^r \binom{j-k+s-1}{s-1} \binom{r+N-1-j}{r} \binom{r+s+N-1-k}{r}^{-1}$$

and

$$B_{kj}^{-1} = (-1)^{k+j+r} \binom{s}{j-k} \binom{r+s+N-1-j}{s+k-j} \binom{s+N-1-j}{s+k-j}^{-1}.$$

The proofs of the all above theorems can be done by using Zeilberger's algorithm similar to the previous section. For example, for the LU-decomposition, we have

$$\sum_{d=0}^{\min(k,j)} L_{kd} U_{dj} = (-1)^{k+j+r} \sum_{d=0}^{\min(k,j)} {r \choose k-d} {k \choose d} {s+k \choose k-d}^{-1} \times {s \choose j-d} {r+s+d \choose r+j} {s+d \choose j}^{-1}.$$

Denote the sum on the RHS by SUM_k . Then Zeilberger's algorithm produces

$$SUM_{k+1} = \frac{j-k+r}{j-k-s-1}SUM_k.$$

After solving this recursion, we obtain the $SUM_k = \binom{r+s}{r+j-k}$, as claimed. Other proofs can be done similarly.

On the other hand, here we would like to present a different approach to prove them. If we perform the limit $q \to 1$ for the results in Section 4.2.1, then we get the results for the matrices including the usual binomial coefficients. When $q \to 1$, the matrix H takes the form

$$\hat{H}_{kj} = (-1)^{r(k+j)+j} \mathbf{i}^{k(1+r-s)+j(1-r+s)-r(1-s-r)} \binom{r+s}{r+j-k}.$$

So it is seen that

$$G_{kj} = (-1)^{r(k+j)+k+r} \mathbf{i}^{k(s-r-1)+j(r-s-1)+r(1-s-r)} \hat{H}_{kj}.$$

By performing the limit $q \to 1$ to the results in Section 4.2.1, we obtain the algebraic properties of the matrix \hat{H} . If we chose the sequences $\{s_n\}$ and $\{m_n\}$ as $\{(-1)^{n(r+1)}\mathbf{i}^{n(s-r-1)+r(1-s-r)}\}$ and $\{(-1)^{r(n+1)}\mathbf{i}^{n(r-s-1)}\}$, respectively and then apply Proposition 4.2 to the results for the matrix \hat{H} , then we obtain the results for the matrix G, as desired. This is a useful prototype to show the efficiency of Proposition 4.2.

Now we present some results about the infinity-norm of the matrix G_N^{-1} , which is the maximum value of the absolute row sum, that is,

$$\|G_N^{-1}\|_{\infty} = \max_k \left(\sum_{j=0}^{N-1} |G_{kj}^{-1}|, 0 \le k \le N-1\right).$$

Firstly, we have the following lemma:

Lemma 4.1. For $0 \le k \le N-1$, the kth row sum, denoted by S_k , of the matrix G_N^{-1} is

$$S_k = (-1)^r \binom{k+r}{r} \binom{N-k-1+s}{s} \binom{r+s}{r}^{-1}.$$

Proof. Let e_k be the unit vector of order N, where 1 is in the kth position and e be the vector of order N, where all entries consist of 1's. Then we may write

$$S_k = e_k^T G_N^{-1} e.$$

Since there is no closed formula for G_N^{-1} , we will use the fact $G^{-1} = U^{-1}L^{-1}$, where the matrices L^{-1} and U^{-1} were given in Theorem 4.5. Thus we should compute

$$S_k = (e_k^T U_N^{-1})(L_N^{-1} e).$$

Here the first parenthesis gives the kth row of the matrix of U_N^{-1} and the second parenthesis gives the row sum of the matrix L_N^{-1} . So the sum of kth row of the matrix L_N^{-1} , denoted by s_k , is

$$s_k = {\binom{s+k}{k}}^{-1} \sum_{j=0}^k {\binom{k-j+r-1}{r-1}} {\binom{s+j}{j}},$$

which, by a variant of the Vandermonde identity (for more details see the Eq. (5.26) in [13]), equals

$$\binom{s+k}{k}^{-1} \binom{k+r+s}{r+s} = \binom{k+r+s}{r} \binom{r+s}{r}^{-1}.$$

Consequently, we have that

$$(L_N^{-1}e) = [s_0, s_1, \dots, s_{N-1}]^T$$

and

$$(e_k^T U_N^{-1}) = [0, 0, \dots, U_{kk}^{-1}, U_{k,k+1}^{-1}, \dots, U_{k,N-1}^{-1}].$$

Finally, we obtain

$$S_k = \sum_{j=k}^{N-1} U_{kj}^{-1} s_j = (-1)^r {r+s \choose s}^{-1} {r+k \choose r} \sum_{j=k}^{N-1} {j-k+s-1 \choose s-1}$$
$$= (-1)^r {r+s \choose s}^{-1} {r+k \choose r} \sum_{j=0}^{N-k-1} {j+s-1 \choose j},$$

which by the formula $\sum_{k\leq n} {r+k \choose k} = {r+n+1 \choose n}$, equals

$$= (-1)^r \binom{r+s}{s}^{-1} \binom{r+k}{r} \binom{s+N-k-1}{s},$$

as claimed. \Box

Before going further, we would like to recall the definition of the unimodal sequence.

Definition 4.1. A unimodal sequence is a finite sequence which first increases and then decreases. That is, a sequence $\{a_1, a_2, \ldots, a_n\}$ is unimodal if there exists an integer $t \in \{2, 3, \ldots, n-1\}$ such that

$$a_1 \leq a_2 \leq \cdots \leq a_t \text{ and } a_t \geq a_{t+1} \geq \cdots \geq a_n.$$

In order to find the infinity-norm of the matrix G_N^{-1} , we need the maximum value of $|S_k|$. For this, we investigate the unimodality of $\{|S_k|\}_{n\geq 0}$.

Lemma 4.2. The sequence $\{|S_k|\}_{n>0}$ is unimodal.

Proof. Since the factor $\binom{r+s}{s}^{-1}$ is independent from the index k, we may show that

$$\{a_k\}_{n\geq 0} = \left\{ \binom{r+k}{r} \binom{s+N-k-1}{s} \right\}$$

is unimodal instead of showing the unimodality of the sequence $\{|S_k|\}_{n\geq 0}$. Consider

$$a_{k}^{2} = {r+k \choose r}^{2} {s+N-k-1 \choose s}^{2}$$

$$= \frac{(k+r)(k+1)(N-k-1+s)(N-k)}{(k+r+1)k(N-k+s)(N-k-1)} a_{k-1} a_{k+1}$$

$$= {1-\frac{1}{k+r+1}} {1+\frac{1}{k}} {1+\frac{1}{k}} {1-\frac{1}{N-k+s}} {1+\frac{1}{N-k-1}} a_{k-1} a_{k+1}$$

$$= {1+\frac{r}{k^{2}+kr+k}} {1+\frac{s}{(k-N+1)(k-N-s)}} a_{k-1} a_{k+1}$$

$$> a_{k-1} a_{k+1},$$

which gives that the sequence $\{a_k\}_{n\geq 0}$ is strictly log-concave that means $\{a_k\}_{n\geq 0}$ is unimodal (For more detail see [90]). Finally, the sequence $\{|S_k|\}_{n\geq 0}$ is unimodal, as well.

Since the sequence $\{|S_k|\}_{n\geq 0}$ is unimodal, it has a maximum value for some k, where $k\in\{1,2,\ldots,N-2\}$. Thus we can compute the $\|G_N^{-1}\|_{\infty}$:

Theorem 4.8. For $N \ge 1$,

$$\|G_N^{-1}\|_{\infty} = \frac{(r+t+1)^{\underline{r}}(s+N-t)^{\underline{s}}}{(r+s)!},$$

where $t = \left\lfloor \frac{Nr}{r+s} \right\rfloor$ and the falling factorial is defined as $x^{\underline{n}} = x(x-1)\cdots(x-n+1)$.

Proof. We know that there exist an integer $k \in \{1, 2, ..., N-2\}$ so that $|S_k|$ is maximum. We shall find this value of k. Similarly, we only consider the sequence $\{a_k\}_{n\geq 0} = \{\binom{r+k}{r}\binom{s+N-k-1}{s}\}$ instead of the sequence $\{S_k\}_{n\geq 0}$ because it is enough to consider the factors only depend on k. Consider

$$\frac{a_{k+1}}{a_k} = \frac{\binom{k+1+r}{r} \binom{N-k-2+s}{s}}{\binom{k+r}{r} \binom{N-k-1+s}{s}}$$

$$= \frac{(k+1+r)(N-k-1)}{(k+1)(N-k-1+s)}$$

$$= \left(1 + \frac{r}{k+1}\right) \left(1 - \frac{s}{N-k-1+s}\right)$$

$$= 1 + \frac{Nr - (k+1)(s+r)}{(k+1)(N-k-1+s)}.$$

If Nr - (k+1)(s+r) > 0, then $k < \frac{Nr}{r+s} - 1$ and so $\{a_k\}_{n \geq 0}$ is increasing for such k's. When $k > \frac{Nr}{r+s} - 1$, the sequence $\{a_k\}_{n \geq 0}$ is decreasing. Since k is an integer, the sequence $\{a_k\}_{n \geq 0}$ takes the maximum value at $k = \lfloor \frac{Nr}{r+s} \rfloor$, which completes the proof.

Denote the sum of the jth column entries of the matrix G_N^{-1} by \overline{S}_j . By Proposition 4.1(v), we can see that $S_k = \overline{S}_{N-1-k}$. So we derive the result

$$||G_N^{-1}||_1 = \overline{S}_{N-1-t} = S_t,$$

where $t = \left| \frac{Nr}{r+s} \right|$ and $\| \cdot \|_1$ is the maximum absolute column sum norm.

At the end of this subsection, we would like to mention the relationship between Toeplitz and Hankel matrices. Let T be a Toeplitz matrix. Then the matrix obtained by

$$M = T \cdot J$$

where J is a square matrix such that the entries outside of the skew main diagonal are all zero otherwise 1, is a Hankel matrix. So if we have a Toeplitz matrix then we obtain a Hankel matrix by the help of the matrix J, or vice versa (Note that $J^{-1} = J$).

Thus the matrix $H \cdot J$ is a Hankel matrix and by using the fact that $\det J_N = (-1)^{\binom{N}{2}}$ and the results in Section 4.2.1, we may compute the determinants of a new family of Hankel matrices and also obtain its Fibonomial analogue as well as the results including binomial coefficients. For example, for r = 3, s = 2 and N = 8 by the results of the

matrix G, we obtain

4.3 A Generalization of the Super Catalan Matrix

As mentioned in Section 3.2.2, in this section we will give the generalizations with two additional parameters of the results in [3]. Briefly, we study the matrices $M = [M_{kj}]$ and $T = [T_{kj}]$ defined by for nonnegative integers r and s and $k, j \ge 0$,

$$M_{kj} = {k+j \choose k} {2k+r \choose k}^{-1} {2j+s \choose j}^{-1}$$

and

$$T_{kj} = \binom{k+j}{k}^{-1} \binom{2k+r}{k} \binom{2j+s}{j},$$

respectively. These matrices are the reciprocals of each other. Clearly, the results in [3] are the case r = s = 0.

In order to obtain more general results, we introduce the matrices \mathcal{M} and \mathcal{T} which are the q-analogues of the matrices M and T, respectively. They are reasonably defined by for $k, j \geq 0$,

$$\mathcal{M}_{kj} = \begin{bmatrix} k+j \\ k \end{bmatrix}_q \begin{bmatrix} 2k+r \\ k \end{bmatrix}_q^{-1} \begin{bmatrix} 2j+s \\ j \end{bmatrix}_q^{-1}$$

and

$$\mathcal{T}_{kj} = \begin{bmatrix} k+j \\ k \end{bmatrix}_q^{-1} \begin{bmatrix} 2k+r \\ k \end{bmatrix}_q \begin{bmatrix} 2j+s \\ j \end{bmatrix}_q,$$

respectively.

For the both matrices, we derive explicit formulæ for the LU-decomposition, inverse matrices L^{-1} and U^{-1} , determinant and Cholesky decomposition when the matrix is symmetric, that is the case r = s. Unfortunately, the inverses do not have closed

formulæ. For this reason, we give explicit expressions for the LU-decomposition of the inverse matrices. Afterwards, when $q \to 1$, we get the results for the matrices M and T. The readers can also find our results in [4].

Similar to the previous section, in this section, our main approach is to guess relevant quantities and then we will use the q-Zeilberger algorithm and formulæ (2.9) and (2.10) to justify relevant equalities. All identities we will obtain hold for general q. One may also obtain Fibonomial analogues of these results by choosing $q = \beta/\alpha$ and the help of Proposition 4.2. But we don't prefer to present that results not only to avoid repetition but also they are a bit cumbersome.

4.3.1 The Matrix \mathcal{M}

For the matrix \mathcal{M} , we have the following theorem.

Theorem 4.9. For $k, j \geq 0$,

$$\mathcal{L}_{kj} = \begin{bmatrix} 2k+r \\ k \end{bmatrix}_q^{-1} \begin{bmatrix} 2j+r \\ j \end{bmatrix}_q \begin{bmatrix} k \\ j \end{bmatrix}_q,$$

$$\mathcal{L}_{kj}^{-1} = (-1)^{k+j} q^{\binom{k-j}{2}} \begin{bmatrix} 2k+r \\ k \end{bmatrix}_q^{-1} \begin{bmatrix} 2j+r \\ j \end{bmatrix}_q \begin{bmatrix} k \\ j \end{bmatrix}_q,$$

$$\mathcal{U}_{kj} = q^{k^2} \begin{bmatrix} 2k+r \\ k \end{bmatrix}_q^{-1} \begin{bmatrix} 2j+s \\ j \end{bmatrix}_q^{-1} \begin{bmatrix} j \\ k \end{bmatrix}_q$$

and

$$\mathcal{U}_{kj}^{-1} = (-1)^{k+j} q^{k(k+1)/2 - j(j+1)/2 - kj} \begin{bmatrix} 2k + s \\ k \end{bmatrix}_{q} \begin{bmatrix} 2j + r \\ j \end{bmatrix}_{q} \begin{bmatrix} j \\ k \end{bmatrix}_{q}.$$

For $0 \le k, j \le N - 1$,

$$\mathcal{A}_{kj} = (-1)^{k+j} q^{k(k+3)/2 - j(j+3)/2 - N(k-j)} \frac{1 - q^{2j+1}}{1 - q^{k+j+1}} \begin{bmatrix} N - j - 1 \\ k - j \end{bmatrix}_q \begin{bmatrix} 2k + s \\ k \end{bmatrix}_q$$

$$\times \begin{bmatrix} k + j \\ k \end{bmatrix}_q^{-1} \begin{bmatrix} 2j + s \\ s \end{bmatrix}_q^{-1} \begin{bmatrix} j + s \\ s \end{bmatrix}_q,$$

$$\mathcal{A}_{kj}^{-1} = q^{(k-j)(k-N+1)} \begin{bmatrix} k + j \\ k \end{bmatrix}_q \begin{bmatrix} N - j - 1 \\ k - j \end{bmatrix}_q \begin{bmatrix} 2j + s \\ j \end{bmatrix}_q^{-1} \begin{bmatrix} 2k + s \\ s \end{bmatrix}_q \begin{bmatrix} k + s \\ s \end{bmatrix}_q^{-1},$$

$$\mathcal{B}_{kj} = (-1)^{k+j} q^{(j+1)(j+2)/2 - N(k+j+1) + 3k(k+1)/2} \begin{bmatrix} 2j + r \\ j \end{bmatrix}_q \begin{bmatrix} N + k \\ k + j + 1 \end{bmatrix}_q \begin{bmatrix} j \\ k \end{bmatrix}_q$$

$$\times \begin{bmatrix} 2k + s \\ s \end{bmatrix}_q \begin{bmatrix} k + s \\ s \end{bmatrix}_q^{-1}$$

and

$$\mathcal{B}_{kj}^{-1} = q^{(k+j+1)(N-j-1)} \frac{1 - q^{2j+1}}{1 - q^{N-k}} \begin{bmatrix} 2k + r \\ k \end{bmatrix}_{q}^{-1} \begin{bmatrix} N+j \\ k+j \end{bmatrix}_{q}^{-1} \begin{bmatrix} j \\ k \end{bmatrix}_{q} \times \begin{bmatrix} 2j+s \\ s \end{bmatrix}_{q}^{-1} \begin{bmatrix} j+s \\ s \end{bmatrix}_{q}.$$

For $N \geq 1$,

$$\det \mathcal{M}_N = q^{N(N-1)(2N-1)/6} \prod_{d=0}^{N-1} {2d+r \brack d}_q^{-1} {2d+s \brack d}_q^{-1}.$$

Finally, when r = s, for $k, j \ge 0$,

$$C_{kj} = q^{j^2/2} \begin{bmatrix} 2k+r \\ k \end{bmatrix}_q^{-1} \begin{bmatrix} k \\ j \end{bmatrix}_q.$$

Now we shall give the proof of this theorem.

Proof. For \mathcal{L} and \mathcal{L}^{-1} .

$$\sum_{j \leq d \leq k} \mathcal{L}_{kd} \mathcal{L}_{dj}^{-1} = \sum_{j \leq d \leq k} (-1)^{d+j} q^{\binom{d-j}{2}} \begin{bmatrix} 2k+r \\ k \end{bmatrix}_q^{-1} \begin{bmatrix} 2d+r \\ d \end{bmatrix}_q \begin{bmatrix} k \\ d \end{bmatrix}_q$$

$$\times \begin{bmatrix} 2d+r \\ d \end{bmatrix}_q^{-1} \begin{bmatrix} d \\ j \end{bmatrix}_q \begin{bmatrix} 2j+r \\ j \end{bmatrix}_q$$

$$= \begin{bmatrix} 2k+r \\ k \end{bmatrix}_q^{-1} \begin{bmatrix} 2j+r \\ j \end{bmatrix}_q \begin{bmatrix} k \\ j \end{bmatrix}_q \sum_{d=0}^{k-j} \begin{bmatrix} k-j \\ d \end{bmatrix}_q (-1)^d q^{\binom{d}{2}}.$$

By Rothe's formula (2.10) if k > j then the last sum on the RHS of the above equation equals $(1;q)_{k-j} = 0$ and if k = j, then it equals 1. Thus we conclude

$$\sum_{j \le d \le k} \mathcal{L}_{kd} \mathcal{L}_{dj}^{-1} = [k = j],$$

as claimed.

For \mathcal{U} and \mathcal{U}^{-1} .

$$\sum_{k \le d \le j} \mathcal{U}_{kd} \mathcal{U}_{dj}^{-1} = q^{k^2 - \binom{j+1}{2}} {2k+r \brack k}_q^{-1} {2j+r \brack j}_q {j \brack k}_q \times q^{k(2j+k+1)/2} (-1)^{k+j} \sum_{0 \le d \le j-k} {j-k \brack d}_q (-1)^d q^{\binom{d+1}{2}+d(k-j)}.$$

By the Cauchy binomial theorem (2.9), if j > k then the last sum on the RHS of the above equation equals $\prod_{d=1}^{j-k} (1-q^{(k-j)+d}) = 0$. The case k = j can be easily computed as 1. So we have

$$\sum_{k \le d \le j} \mathcal{U}_{kd} \mathcal{U}_{dj}^{-1} = [k = j],$$

as desired.

For the LU-decomposition, we should show that

$$\sum_{0 \leq d \leq \min(k,j)} \mathcal{L}_{kd} \mathcal{U}_{dj} = \mathcal{M}_{kj}.$$

Firstly, we can assume that $k \leq j$. Consider,

$$\sum_{0 \le d \le \min(k,j)} \mathcal{L}_{kd} \mathcal{U}_{dj} = \begin{bmatrix} 2k+r \\ k \end{bmatrix}_{q}^{-1} \begin{bmatrix} 2j+s \\ j \end{bmatrix}_{q}^{-1} (q;q)_{k} (q;q)_{j}$$

$$\times \sum_{0 \le d \le k} q^{d^{2}} \frac{1}{(q;q)_{d}^{2} (q;q)_{k-d} (q;q)_{j-d}}.$$
(4.4)

Denote the sum in (4.4) by SUM_k . The Mathematica package of the q-Zeilberger algorithm produces the recursion

$$\mathtt{SUM}_k = \frac{1 - q^{j+k}}{(1 - q^k)^2} \mathtt{SUM}_{k-1}.$$

Since $SUM_0 = (q;q)_k^{-1}(q;q)_j^{-1}$, we obtain

$$\mathrm{SUM}_k = (q;q)_k^{-1}(q;q)_j^{-1} \begin{bmatrix} k+j \\ k \end{bmatrix}_q.$$

Since the sum in (4.4) is symmetric with respect to k and j, the case j < k follows likewise. Eventually, we get that

$$\sum_{0 \leq d \leq \min(k,j)} \mathcal{L}_{kd} \mathcal{U}_{dj} = \mathcal{M}_{kj},$$

which completes the proof of the LU-decomposition of the matrix \mathcal{M} .

For \mathcal{A} and \mathcal{A}^{-1} , consider

$$\begin{split} \sum_{j \leq d \leq k} \mathcal{A}_{kd} \mathcal{A}_{dj}^{-1} &= (-1)^k q^{k(k+3)/2 - j + N(j-k)} \frac{(q;q)_{N-j-1}}{(q;q)_{N-k-1}} \\ &\times \begin{bmatrix} 2k + s \\ k \end{bmatrix}_q \begin{bmatrix} 2j + s \\ j \end{bmatrix}_q^{-1} \begin{bmatrix} k \\ j \end{bmatrix}_q \\ &\times \sum_{j \leq d \leq k} \begin{bmatrix} k - j \\ d - j \end{bmatrix}_q (-1)^d q^{d(d-1)/2 - jd} \frac{(q;q)_{d+j}}{(q;q)_{d-j}} \frac{1 - q^{2d+1}}{1 - q^{k+d+1}}. \end{split}$$

For the sum on the last line of the above equation, we get that it is equal to 0 provided that $k \neq j$ by the q-Zeilberger algorithm. If k = j, it is obvious that $\mathcal{A}_{kk}\mathcal{A}_{kk}^{-1} = 1$. Thus

$$\sum_{j \le d \le k} \mathcal{A}_{kd} \mathcal{A}_{dj}^{-1} = [k = j],$$

as claimed. Similarly, by the q-Zeilberger algorithm, we compute

$$\sum_{k \le d \le j} \mathcal{B}_{kd} \mathcal{B}_{dj}^{-1} = [k = j].$$

When r = s, then we have a symmetric matrix. Thus by Corollary 4.1, the Cholesky decomposition can be computed as

$$C_{kj} = \mathcal{U}_{jk} \mathcal{U}_{jj}^{-\frac{1}{2}}$$

$$= q^{j^2} \begin{bmatrix} 2j+r \\ j \end{bmatrix}_a^{-1} \begin{bmatrix} 2k+r \\ k \end{bmatrix}_a^{-1} \begin{bmatrix} k \\ j \end{bmatrix}_a q^{-\frac{1}{2}j^2} \begin{bmatrix} 2j+r \\ j \end{bmatrix}_a = q^{j^2/2} \begin{bmatrix} 2k+r \\ k \end{bmatrix}_a^{-1} \begin{bmatrix} k \\ j \end{bmatrix}_a,$$

as claimed.

For the LU-decomposition of \mathcal{M}^{-1} , we should show that $\mathcal{M}^{-1} = \mathcal{A} \cdot \mathcal{B}$ which is the same as $\mathcal{M} = \mathcal{B}^{-1} \cdot \mathcal{A}^{-1}$. So it is sufficient to show that

$$\sum_{\max(k,j) \le d \le N-1} \mathcal{B}_{kd}^{-1} \mathcal{A}_{dj}^{-1} = \mathcal{M}_{kj}.$$

After some arrangements, we have

$$\sum_{\max(k,j) \le d \le N-1} \mathcal{B}_{kd}^{-1} \mathcal{A}_{dj}^{-1} = \begin{bmatrix} 2k+r \\ k \end{bmatrix}_{q}^{-1} \begin{bmatrix} 2j+s \\ j \end{bmatrix}_{q}^{-1} \sum_{j \le d \le N-1} q^{(j+k+1)(N-1-d)}$$

$$\times \frac{1-q^{2d+1}}{1-q^{N-k}} \begin{bmatrix} d \\ k \end{bmatrix}_{q} \begin{bmatrix} N+d \\ k+d \end{bmatrix}_{q}^{-1} \begin{bmatrix} d+j \\ d \end{bmatrix}_{q} \begin{bmatrix} N-j-1 \\ d-j \end{bmatrix}_{q}.$$

By replacing (N-1) with N, the sum on the RHS of the above equation equals

$$\sum_{j \le d \le N} q^{(j+k+1)(N-d)} \frac{1 - q^{2d+1}}{1 - q^{N+1-k}} \begin{bmatrix} d \\ k \end{bmatrix}_q \begin{bmatrix} N+1+d \\ k+d \end{bmatrix}_q^{-1} \begin{bmatrix} d+j \\ d \end{bmatrix}_q \begin{bmatrix} N-j \\ d-j \end{bmatrix}_q.$$

Denote this sum by SUM_N . The q-Zeilberger algorithm gives the following recursion provided that $k \neq N$ and $j \neq N$

$$SUM_N = SUM_{N-1}$$
.

So $SUM_N = SUM_j = {k+j \brack k}_q$, which completes the proof except the case (k,j) = (N-1,N-1). This case could be easily checked by hand.

Thus the proof is completed.

4.3.2 The Matrix \mathcal{T}

For the matrix \mathcal{T} , we have the following result.

Theorem 4.10. For $k, j \ge 0$,

$$\mathcal{L}_{kj} = \begin{bmatrix} 2k+r \\ k+j \end{bmatrix}_q \begin{bmatrix} k \\ j \end{bmatrix}_q \begin{bmatrix} k+r \\ j \end{bmatrix}_q^{-1} \begin{bmatrix} 2j+r \\ r \end{bmatrix}_q^{-1} \begin{bmatrix} j+r \\ r \end{bmatrix}_q.$$

For $j \geq 1$ and $k \geq 0$,

$$\mathcal{L}_{kj}^{-1} = (-1)^{k+j} q^{\binom{k-j}{2}} \frac{1-q^{2k}}{1-q^{k+j}} {k+j \brack k-j}_q {2k+r \brack r}_q {k+r \brack r}_q {2j+r \brack r}_q^{-1} {j+r \brack r}_q,$$

for $k \geq 1$,

$$\mathcal{L}_{k0}^{-1} = (-1)^k (1+q^k) q^{\binom{k}{2}} \begin{bmatrix} 2k+r \\ r \end{bmatrix}_a \begin{bmatrix} k+r \\ r \end{bmatrix}_a^{-1}$$

and $\mathcal{L}_{00}^{-1} = 1$. For $k \ge 1$ and $j \ge 0$,

$$\mathcal{U}_{kj} = (-1)^k q^{k(3k-1)/2} (1+q^k) \begin{bmatrix} 2j+s \\ k+j \end{bmatrix}_q \begin{bmatrix} 2k+r \\ r \end{bmatrix}_q \begin{bmatrix} j-k+s \\ s \end{bmatrix}_q$$

$$\times \begin{bmatrix} k+r \\ r \end{bmatrix}_q^{-1} \begin{bmatrix} j+s \\ s \end{bmatrix}_q^{-1}$$

and for $j \geq 0$,

$$\mathcal{U}_{0j} = \begin{bmatrix} 2j+s \\ j \end{bmatrix}_{a}.$$

For $k, j \geq 0$,

$$\mathcal{U}_{kj}^{-1} = (-1)^k q^{k(k+1)/2 - j(k+j)} \frac{1 - q^j}{1 - q^{k+j}} \begin{bmatrix} k+j \\ j-k \end{bmatrix}_q \begin{bmatrix} 2j+r \\ r \end{bmatrix}_q^{-1} \begin{bmatrix} j+r \\ r \end{bmatrix}_q$$

$$\times \begin{bmatrix} 2k+s \\ s \end{bmatrix}_q^{-1} \begin{bmatrix} k+s \\ s \end{bmatrix}_q.$$

For $0 \le k, j \le N - 1$,

$$\mathcal{A}_{kj} = (-1)^{k+j} q^{(k+1)(k+2)/2 - (j+1)(j+2)/2 + N(j-k)} \begin{bmatrix} k \\ j \end{bmatrix}_q \begin{bmatrix} N+k-1 \\ 2k \end{bmatrix}_q$$

$$\times \begin{bmatrix} N+j-1 \\ 2j \end{bmatrix}_q^{-1} \begin{bmatrix} k+s \\ s \end{bmatrix}_q \begin{bmatrix} 2k+s \\ s \end{bmatrix}_q^{-1} \begin{bmatrix} j+s \\ s \end{bmatrix}_q^{-1} \begin{bmatrix} 2j+s \\ s \end{bmatrix}_q,$$

$$\mathcal{A}_{kj}^{-1} = q^{(k-j)(k-N+1)} \begin{bmatrix} k \\ j \end{bmatrix}_q \begin{bmatrix} N+k-1 \\ 2k \end{bmatrix}_q \begin{bmatrix} N+j-1 \\ 2j \end{bmatrix}_q^{-1} \begin{bmatrix} k+s \\ s \end{bmatrix}_q \begin{bmatrix} 2k+s \\ s \end{bmatrix}_q^{-1} \times \begin{bmatrix} j+s \\ s \end{bmatrix}_q^{-1} \begin{bmatrix} 2j+s \\ s \end{bmatrix}_q,$$

$$\mathcal{B}_{kj} = q^{(j+1)(j+2)/2 - N(N-1)/2 - jN + k^2 - 1} \begin{bmatrix} N+j-1\\2j \end{bmatrix}_q \begin{bmatrix} j\\k \end{bmatrix}_q \begin{bmatrix} 2k+s\\k \end{bmatrix}_q^{-1} \times \begin{bmatrix} j+r\\r \end{bmatrix}_q \begin{bmatrix} 2j+r\\r \end{bmatrix}_q^{-1}$$

and

$$\mathcal{B}_{kj}^{-1} = (-1)^{N+j+1} q^{k-kj-j(j+1)/2+kN+N(N-1)/2} \begin{bmatrix} j \\ k \end{bmatrix}_q \begin{bmatrix} N+k-1 \\ 2k \end{bmatrix}_q^{-1} \begin{bmatrix} 2j+s \\ s \end{bmatrix}_q \times \begin{bmatrix} 2k+r \\ r \end{bmatrix}_q \begin{bmatrix} k+r \\ r \end{bmatrix}_q^{-1}.$$

For $N \geq 1$,

$$\det \mathcal{T}_N = (-1)^{N(N-1)/2} \prod_{d=1}^{N-1} q^{d(3d-1)/2} \begin{bmatrix} 2d+s \\ s \end{bmatrix}_q \begin{bmatrix} 2d+r \\ r \end{bmatrix}_q \begin{bmatrix} d+r \\ r \end{bmatrix}_q^{-1} \begin{bmatrix} d+s \\ s \end{bmatrix}_q^{-1}.$$

Finally, when r = s, for $j \ge 1$ and $k \ge 0$,

$$C_{kj} = \mathbf{i}^{j} (1+q)^{j/2} q^{j(3j-1)/4} \begin{bmatrix} 2k+r \\ k+j \end{bmatrix}_{q} \begin{bmatrix} k+r \\ r \end{bmatrix}_{q}^{-1} \begin{bmatrix} k-j+r \\ r \end{bmatrix}_{q}$$

and for $k \geq 0$,

$$C_{k0} = \begin{bmatrix} 2k+r \\ k \end{bmatrix}_{a}.$$

Proof. By the definitions of the matrices \mathcal{L} and \mathcal{L}^{-1} , for the case j=0, we have

$$\sum_{0 \leq d \leq k} \mathcal{L}_{kd} \mathcal{L}_{d0}^{-1} = \mathcal{L}_{kd} \mathcal{L}_{00}^{-1} + \sum_{1 \leq d \leq k} \mathcal{L}_{kd} \mathcal{L}_{d0}^{-1}.$$

If k = 0, we get 1 as (0,0)th entry of the multiplication $\mathcal{L} \cdot \mathcal{L}^{-1}$. If k > 0, after some rearrangements, we have

$$\begin{split} \sum_{1 \leq d \leq k} \mathcal{L}_{kd} \mathcal{L}_{d0}^{-1} &= \sum_{0 \leq d \leq n} \mathcal{L}_{n+1,d+1} \mathcal{L}_{d+1,0}^{-1} \\ &= \sum_{0 \leq d \leq n} (-1)^{d+1} (1+q^{d+1}) q^{(d^2+d)/2} {2n+2+r \brack n+d+2}_q \\ &\times {n+1 \brack d+1}_q {n+1+r \brack d+1}_q^{-1}. \end{split}$$

The q-Zeilberger algorithm compute the sum on the RHS of the above equation as $-\begin{bmatrix}2^{n+2+r}\\n+1\end{bmatrix}_q$. By changing n+1 with k again, we get $-\begin{bmatrix}2^{k+r}\\k\end{bmatrix}_q$. Finally, if k>0,

$$\sum_{0 \le d \le k} \mathcal{L}_{kd} \mathcal{L}_{d0}^{-1} = \begin{bmatrix} 2k+r \\ k \end{bmatrix}_q + \sum_{1 \le d \le k} \mathcal{L}_{kd} \mathcal{L}_{d0}^{-1}$$

$$= \begin{bmatrix} 2k+r \\ k \end{bmatrix}_{a} - \begin{bmatrix} 2k+r \\ k \end{bmatrix}_{a} = 0,$$

as desired. For the case j > 0, we have

$$\sum_{j \le d \le k} \mathcal{L}_{kd} \mathcal{L}_{dj}^{-1} = \sum_{j \le d \le k} (-1)^{d+j} q^{\binom{d-j}{2}} \frac{1 - q^{2d}}{1 - q^{d+j}} {2k + r \brack k + d}_q {k \brack d}_q$$

$$\times {k + r \brack d}_q {d+j \brack d - j}_q {2j + r \brack r}_q^{-1} {j+r \brack r}_q.$$

Again by the q-Zeilberger algorithm, we obtain that it is equal to 0 provided that $k \neq j$. The case k = j could be easily computed as 1. Finally,

$$\sum_{j \le d \le k} \mathcal{L}_{kd} \mathcal{L}_{dj}^{-1} = [k = j],$$

as claimed. Verification of the inverse of \mathcal{U} can be similarly done by the help of the q-Zeilberger algorithm. Inverses of the matrices \mathcal{A} and \mathcal{B} may be shown as in the proof of Theorem 4.9. We would prefer to omit them due to the similarities.

For the LU-decomposition, we have to prove that

$$\sum_{0 \leq d \leq \min(k,j)} \mathcal{L}_{kd} \mathcal{U}_{dj} = \mathcal{T}_{kj}.$$

The cases $k=0,\,j\geq 0$ and $j=0,\,k\geq 0$ could be easily seen. For the other cases, consider

$$\sum_{0 \le d \le \min(k,j)} \mathcal{L}_{kd} \mathcal{U}_{dj} = \mathcal{L}_{k0} \mathcal{U}_{0j} + \sum_{1 \le d \le \min(k,j)} \mathcal{L}_{kd} \mathcal{U}_{dj} = \begin{bmatrix} 2k+r \\ k \end{bmatrix}_q \begin{bmatrix} 2j+s \\ j \end{bmatrix}_q$$

$$+ \sum_{1 \le d \le \min(k,j)} (-1)^d (1+q^d) q^{(3d-1)d/2} \begin{bmatrix} 2k+r \\ k+d \end{bmatrix}_q \begin{bmatrix} k \\ d \end{bmatrix}_q$$

$$\times \begin{bmatrix} k+r \\ d \end{bmatrix}_q \begin{bmatrix} 2j+s \\ j+d \end{bmatrix}_q \begin{bmatrix} j-d+s \\ s \end{bmatrix}_q \begin{bmatrix} j+s \\ s \end{bmatrix}_q^{-1}$$

$$= \begin{bmatrix} 2k+r \\ k \end{bmatrix}_q \begin{bmatrix} 2j+s \\ j \end{bmatrix}_q + \begin{bmatrix} 2k+r \\ k \end{bmatrix}_q \begin{bmatrix} 2j+s \\ j \end{bmatrix}_q$$

$$\times \begin{bmatrix} 2k \\ k \end{bmatrix}_q^{-1} \begin{bmatrix} 2j \\ j \end{bmatrix}_q^{-1} \sum_{1 \le d \le \min(k,j)} (-1)^d (1+q^d) q^{(3d-1)d/2}$$

$$\times \begin{bmatrix} 2k \\ k+d \end{bmatrix}_q \begin{bmatrix} 2j \\ j+d \end{bmatrix}_q.$$

Without loss of generality, we may assume that $k \leq j$. So consider the sum

$$\mathrm{SUM}_k = \sum_{-k \leq d \leq k} (-1)^d (1+q^d) q^{(3d-1)d/2} {2k \brack k+d}_q {2j \brack j+d}_q.$$

Then the q-Zeilberger algorithm gives the following recurrence relation for the sum SUM_k .

$${\rm SUM}_k = \frac{(1+q^k)(1-q^{2k-1})}{(1-q^{j+k})} {\rm SUM}_{k-1}.$$

Since $SUM_0 = 2 {2j \brack j}_q$, we obtain

$$\mathrm{SUM}_k = 2 \begin{bmatrix} 2k \\ k \end{bmatrix}_a \begin{bmatrix} 2j \\ j \end{bmatrix}_a \begin{bmatrix} k+j \\ k \end{bmatrix}_a^{-1}.$$

Since

$$\begin{split} \mathrm{SUM}_k &= \sum_{-k \leq d \leq -1} (-1)^d (1+q^d) q^{(3d-1)d/2} {2k \brack k+d}_q {2j \brack j+d}_q + 2 {2k \brack k}_q {2j \brack j}_q \\ &+ \sum_{1 \leq d \leq k} (-1)^d (1+q^d) q^{(3d-1)d/2} {2k \brack k+d}_q {2j \brack j+d}_q \\ &= 2 {2k \brack k}_q {2j \brack j}_q + 2 \sum_{1 \leq d \leq k} (-1)^d (1+q^d) q^{(3d-1)d/2} {2k \brack k+d}_q {2j \brack j+d}_q, \end{split}$$

then we have

$$\sum_{0 \le d \le k} \mathcal{L}_{kd} \mathcal{U}_{dj} = \begin{bmatrix} 2k+r \\ k \end{bmatrix}_q \begin{bmatrix} 2j+s \\ j \end{bmatrix}_q + \begin{bmatrix} 2k+r \\ k \end{bmatrix}_q \begin{bmatrix} 2j+s \\ j \end{bmatrix}_q$$

$$\times \begin{bmatrix} 2k \\ k \end{bmatrix}_q^{-1} \begin{bmatrix} 2j \\ j \end{bmatrix}_q^{-1} \left(\frac{1}{2} \text{SUM}_k - \begin{bmatrix} 2k \\ k \end{bmatrix}_q \begin{bmatrix} 2j \\ j \end{bmatrix}_q \right)$$

$$= \begin{bmatrix} 2k+r \\ k \end{bmatrix}_q \begin{bmatrix} 2j+s \\ j \end{bmatrix}_q \begin{bmatrix} k+j \\ k \end{bmatrix}_q^{-1} = \mathcal{T}_{kj},$$

as desired.

The proof of the LU-decomposition of the inverse matrix \mathcal{T}^{-1} could be similarly done as in the proof of Theorem 4.9. Similarly, when r = s, the Cholesky decomposition follows by Corollary 4.1.

4.3.3 The Matrices M and T

Obviously, we have

$$\lim_{q\to 1} \mathcal{M} = M$$
 and $\lim_{q\to 1} \mathcal{T} = T$.

Thus the related results for the matrices M and T can be easily obtained by performing the limit $q \to 1$ to the results in Sections 4.3.1 and 4.3.2, respectively. Without any efforts, we obtain new results including binomial coefficients. This is one of the greatest advantages of studying q-analogues.

Now we list the results for the matrices M and T, respectively.

Theorem 4.11. For the matrix M, we have for $k, j \geq 0$,

$$L_{kj} = {2k+r \choose k}^{-1} {2j+r \choose j} {k \choose j},$$

$$L_{kj}^{-1} = (-1)^{k+j} {2k+r \choose k}^{-1} {2j+r \choose j} {k \choose j},$$

$$U_{kj} = {2k+r \choose k}^{-1} {2j+s \choose j}^{-1} {j \choose k}$$

and

$$U_{kj}^{-1} = \left(-1\binom{k+j}{2k+s}\binom{2k+s}{j}\binom{2j+r}{j}\binom{j}{k}\right).$$

For $0 \le k, j \le N - 1$,

$$A_{kj} = (-1)^{k+j} \frac{1+2j}{k+j+1} \binom{N-j-1}{k-j} \binom{2k+s}{k} \binom{k+j}{k}^{-1} \binom{2j+s}{s}^{-1} \binom{j+s}{s},$$

$$A_{kj}^{-1} = \binom{k+j}{k} \binom{N-j-1}{k-j} \binom{2j+s}{j}^{-1} \binom{2k+s}{s} \binom{k+s}{s}^{-1},$$

$$B_{kj} = (-1)^{k+j} \binom{2j+r}{j} \binom{N+k}{k+j+1} \binom{j}{k} \binom{2k+s}{s} \binom{k+s}{s}^{-1}$$

and

$$B_{kj}^{-1} = \frac{2j+1}{N-k} {2k+r \choose k}^{-1} {N+j \choose k+j}^{-1} {j \choose k} {2j+s \choose s}^{-1} {j+s \choose s}.$$

For $N \geq 1$,

$$\det \mathcal{M}_N = \prod_{k=0}^{N-1} {2k+r \choose k}^{-1} {2k+s \choose k}^{-1}$$

and finally, when r = s, for $k, j \ge 0$,

$$C_{kj} = \binom{2k+r}{k}^{-1} \binom{k}{j}.$$

Theorem 4.12. For the matrix T, we have for $k, j \geq 0$,

$$L_{kj} = {2k+r \choose k+j} {k \choose j} {k+r \choose j}^{-1} {2j+r \choose r}^{-1} {j+r \choose r},$$

for $j \ge 1$ and $k \ge 0$,

$$L_{kj}^{-1} = (-1)^{k+j} \frac{2k}{k+j} {k+j \choose k-j} {2k+r \choose r} {k+r \choose r}^{-1} {2j+r \choose r}^{-1} {j+r \choose r},$$

$$L_{k0}^{-1} = 2(-1)^k {2k+r \choose r} {k+r \choose r}^{-1}$$

and $L_{00}^{-1} = 1$. For $k \ge 1$ and $j \ge 0$,

$$U_{kj} = (-1)^k 2 \binom{2j+s}{k+j} \binom{2k+r}{r} \binom{j-k+s}{s} \binom{k+r}{r}^{-1} \binom{j+s}{s}^{-1}$$

and
$$U_{0j} = {2j+s \choose j}$$
. For $k, j \ge 0$,

$$U_{kj}^{-1} = (-1)^k \frac{j}{k+j} \binom{k+j}{j-k} \binom{2j+r}{r}^{-1} \binom{j+r}{r} \binom{2k+s}{s}^{-1} \binom{k+s}{s}.$$

For $0 \le k, j \le N - 1$,

$$A_{kj} = (-1)^{k+j} \binom{k}{j} \binom{N+k-1}{2k} \binom{N+j-1}{2j}^{-1} \binom{2k+s}{s}^{-1} \binom{k+s}{s}$$
$$\times \binom{2j+s}{s} \binom{j+s}{s}^{-1},$$

$$A_{kj}^{-1} = \binom{k}{j} \binom{N+k-1}{2k} \binom{N+j-1}{2j}^{-1} \binom{2k+s}{s}^{-1} \binom{k+s}{s} \binom{2j+s}{s} \binom{j+s}{s}^{-1},$$

$$B_{kj} = \binom{N+j-1}{2j} \binom{j}{k} \binom{2k+s}{s}^{-1} \binom{j+r}{r} \binom{2j+r}{r}^{-1}$$

and

$$B_{kj}^{-1} = (-1)^{N+j+1} {j \choose k} {N+k-1 \choose 2k}^{-1} {2j+s \choose s} {2k+r \choose r} {k+r \choose r}^{-1}.$$

For $N \geq 1$,

$$\det T_N = (-1)^{N(N-1)/2} \prod_{d=1}^{N-1} {2d+s \choose s} {2d+r \choose r} {d+r \choose r}^{-1} {d+s \choose s}^{-1}.$$

Finally, when r = s, for $j \ge 1$ and $k \ge 0$,

$$C_{kj} = (-2)^{j/2} {2k+r \choose k+j} {k+r \choose r}^{-1} {k-j+r \choose r}$$

and for $k \geq 0$,

$$C_{k0} = \binom{2k+r}{k}.$$

We want to finish this section by giving a conclusion remark. We will show how one can obtain results for the matrices M and T or similar kind of matrices without the help of the q-analogues.

For example, we shall take the matrix M. We can consider the entries of it as

$$M_{kj} = \binom{k+j}{k} s_k m_j,$$

where $s_k = {2k+r \choose k}^{-1}$ and $m_j = {2j+s \choose j}^{-1}$. Thus the entries of the matrix M is separable with respect to indexes. So we can apply Proposition 4.2 if we know the properties of the matrix ${k+j \choose k}_{k,j>0}$. Fortunately, the algebraic properties of this symmetric Pascal

matrix (also its reciprocal analogue) are studied in [58]. For example, the (k, j)th entry of the L matrix coming from the LU-decomposition of this symmetric Pascal matrix is $\binom{k}{j}$. So if we apply Proposition 4.2 for the matrix M, we obtain its L matrix as

$$L_{kj} = \binom{k}{j} \binom{2k+r}{k}^{-1} \binom{2j+r}{j},$$

which is the exactly same with the quantity given in Theorem 4.11. Likewise, all algebraic properties can be obtained in this way. Furthermore, since $\binom{k+j}{k} = \frac{(k+j)!}{k!j!}$, by choosing $s_k = k!$ and $m_j = j!$ in Proposition 4.2, we get new results for the matrix $[(k+j)!]_{k,j\geq 0}$, which is a Hankel matrix, as well. More generally, by Proposition 4.2, one can easily extend our results by considering the matrix with entries for $k, j \geq 0$,

$$\binom{k+j}{k} \prod_{l=1}^{m_1} \binom{a_l k + p_l}{k} \prod_{l=1}^{m_2} \binom{b_l k + r_l}{k}^{-1} \prod_{l=1}^{m_3} \binom{c_l j + s_l}{j} \prod_{l=1}^{m_4} \binom{d_l j + t_l}{j}^{-1},$$

as well as its reciprocal analogue, where the parameters m_1 , m_2 , m_3 , m_4 , p_l 's, r_l 's, s_l 's and t_l 's are nonnegative integers and a_l 's, b_l 's, c_l 's and d_l 's are positive integers.

Thus we see that Proposition 4.2 is very useful to derive new results as well as to prove existing identities.

4.4 General Family of the Max and Min Matrices

As mentioned in Section 3.2.3, recently the authors [5] studied the matrices of order N defined by $[\max(a_k, a_j)]_{1 \leq k, j \leq N}$ and $[\min(a_k, a_j)]_{1 \leq k, j \leq N}$ over the set $\{a_1, a_2, \ldots, a_N\}$, such that $a_1 \leq a_2 \leq \cdots \leq a_N$. They obtained some algebraic properties of these matrices. Note that their approach only works for the increasing sequence. Their method based on another auxiliary family of the matrices which called "meet and join matrices". They also indicated that the elementary tools are difficult to derive such results.

In this section, we generalize their results for an arbitrary sequence $\{a_n\}$ by defining the matrices conveniently. In other words, we define two new families of the matrices, which called Max and Min matrices, whose entries run in left-reversed and up-reversed L-shaped pattern, respectively. By any given sequence $\{a_n\}$, we define the matrices M_1 , M_2 as

$$(M_1)_{kj} = a_{\max(k,j)}, \qquad (M_2)_{kj} = a_{\min(k,j)}$$

and their reciprocal analogues \mathcal{M}_1 and \mathcal{M}_2 as

$$(\mathcal{M}_1)_{kj} = \frac{1}{a_{\max(k,j)}}, \quad (\mathcal{M}_2)_{kj} = \frac{1}{a_{\min(k,j)}}.$$

Here note that the size of the matrices does not matter, as well. That means we can consider them as infinite matrices. Clearly, the matrices M_1 and M_2 are of the forms

$$M_{1} = \begin{bmatrix} a_{1} & a_{2} & a_{3} & \cdots & a_{n} & \cdots \\ a_{2} & a_{2} & a_{3} & \cdots & a_{n} & \cdots \\ a_{3} & a_{3} & a_{3} & \cdots & a_{n} & \cdots \\ \vdots & \vdots & \vdots & \ddots & \vdots & \ddots \\ a_{n} & a_{n} & a_{n} & \cdots & a_{n} & \cdots \\ \vdots & \vdots & \vdots & \ddots & \ddots & \cdots \end{bmatrix} \text{ and } M_{2} = \begin{bmatrix} a_{1} & a_{1} & a_{1} & \cdots & a_{1} & \cdots \\ a_{1} & a_{2} & a_{2} & \cdots & a_{2} & \cdots \\ a_{1} & a_{2} & a_{3} & \cdots & a_{3} & \cdots \\ \vdots & \vdots & \vdots & \ddots & \vdots & \ddots \\ a_{1} & a_{2} & a_{3} & \cdots & a_{n} & \ddots \\ \vdots & \vdots & \vdots & \ddots & \ddots & \ddots \end{bmatrix},$$

respectively.

It is worthwhile to note that if the sequence $\{a_n\}$ is increasing, then $a_{\max(k,j)} = \max(a_k, a_j)$ and $a_{\min(k,j)} = \min(a_k, a_j)$. Conversely, if the sequence $\{a_n\}$ is decreasing, then $a_{\max(k,j)} = \min(a_k, a_j)$ and $a_{\min(k,j)} = \max(a_k, a_j)$. Thus the results of [5] will be the special cases of our results.

We will study various properties of the matrices M_1 , M_2 , \mathcal{M}_1 and \mathcal{M}_2 , such as LUdecomposition, inverse, Cholesky decomposition, etc. by using elementary tools which
are simpler and more convenient way than the method used in [5]. In Section 4.4.1,
we go around the matrices M_1 and \mathcal{M}_1 . Afterwards, in Section 4.4.2, we examine the
matrices M_2 and \mathcal{M}_2 . Finally, we give some further applications of our main results.
For example, as a consequence of our results, we will give an idea how we can obtain
a generalization of the Lehmer matrix and its reciprocal analogue.

We have the following lemma for later use.

Lemma 4.3. Let $\{a_n\}$ be any real sequence. Then for all k, j > 0, we have

$$a_{\max(k,j)}a_{\min(k,j)} = a_k a_j$$
.

Proof. For all cases k = j, k > j and j > k, it is obviously seen.

Throughout this section, we assume that $\{a_n\}$ is any sequence such that $a_n \neq 0$ and $a_n \neq a_{n+1}$ for all $n \geq 1$, otherwise the matrices M_1 and M_2 become singular.

4.4.1 Max Matrices and Their Reciprocal Analogues

We derive the LU-decompositions, inverses, Cholesky decompositions and LU-decompositions of the inverses of the matrices M_1 and \mathcal{M}_1 , respectively.

For the matrix M_1 , we have the following results.

We start with the LU-decomposition:

Theorem 4.13. *For* $k, j \ge 1$,

$$L_{kj} = \begin{cases} \frac{a_k}{a_j} & \text{if } k \ge j, \\ 0 & \text{otherwise} \end{cases}$$

and

$$U_{kj} = \begin{cases} a_j & \text{if } k = 1, \\ \frac{a_j (a_{k-1} - a_k)}{a_{k-1}} & \text{if } j \ge k > 1, \\ 0 & \text{otherwise.} \end{cases}$$

Now we shall give the inverse matrices L^{-1} and U^{-1} by the following theorem.

Theorem 4.14. For $k, j \geq 1$,

$$L_{kj}^{-1} = \begin{cases} (-1)^{k+j} \frac{a_k}{a_j} & \text{if } 0 \le k-j \le 1, \\ 0 & \text{otherwise} \end{cases}$$

and

$$U_{kj}^{-1} = \begin{cases} (-1)^{k+j} \frac{a_{j-1}}{a_k (a_{j-1} - a_j)} & \text{if } 0 \le j - k \le 1 \text{ and } j \ne 1, \\ \frac{1}{a_1} & \text{if } k = j = 1, \\ 0 & \text{otherwise.} \end{cases}$$

Now we compute the inverse matrix $(M_1)_N^{-1}$ as follows.

Theorem 4.15. For $1 \leq k, j \leq N$, $(M_1)_N^{-1}$ is the symmetric tridiagonal matrix defined

by

$$(M_1^{-1})_{kj} = \begin{cases} \frac{1}{a_1 - a_2} & \text{if } k = j = 1, \\ \frac{a_{k-1} - a_{k+1}}{(a_{k+1} - a_k)(a_k - a_{k-1})} & \text{if } 1 \neq k = j \neq N, \\ \frac{a_{N-1}}{a_N(a_{N-1} - a_N)} & \text{if } k = j = N, \\ \frac{1}{a_k - a_{k-1}} & \text{if } k = j + 1. \end{cases}$$

For the Cholesky decomposition, we have the following result.

Theorem 4.16. For $k, j \geq 1$, C is the lower triangular matrix defined by

$$C_{kj} = \begin{cases} \frac{a_k}{\sqrt{a_1}} & \text{if } j = 1, \\ \frac{a_k}{a_j a_{j-1}} \sqrt{a_j a_{j-1} (a_{j-1} - a_j)} & \text{if } j > 1. \end{cases}$$

We will give the LU-decomposition of $(M_1)_N^{-1}$ and the inverses of these factor matrices by the following theorems.

Theorem 4.17. For $1 \le k, j \le N$,

$$A_{kj} = \begin{cases} (-1)^{k+j} & \text{if } 0 \le k-j \le 1, \\ 0 & \text{otherwise} \end{cases}$$

and

$$B_{kj} = \begin{cases} \frac{1}{a_N} & \text{if } k = j = N, \\ (-1)^{k+j} \frac{1}{a_k - a_{k+1}} & \text{if } 0 \le j - k \le 1 \text{ and } k \ne N, \\ 0 & \text{otherwise.} \end{cases}$$

Theorem 4.18. For $1 \le k, j \le N$,

$$A_{kj}^{-1} = \begin{cases} 1 & if \ k \ge j, \\ 0 & otherwise \end{cases}$$

and

$$B_{kj}^{-1} = \begin{cases} a_N & \text{if } j = N, \\ a_j - a_{j+1} & \text{if } k \le j < N, \\ 0 & \text{otherwise.} \end{cases}$$

Proof. In order to prove $M_1 = L \cdot U$, it is sufficient to show that

$$\sum_{d=1}^{\min(k,j)} L_{kd} U_{dj} = a_{\max(k,j)}.$$

Consider

$$\begin{split} \sum_{d=1}^{\min(k,j)} L_{kd} U_{dj} &= \frac{a_k}{a_1} a_j + \sum_{d=2}^{\min(k,j)} \frac{a_k}{a_d} \frac{a_j (a_{d-1} - a_d)}{a_{d-1}} \\ &= a_k a_j \left[\frac{1}{a_1} + \sum_{d=2}^{\min(k,j)} \left(\frac{1}{a_d} - \frac{1}{a_{d-1}} \right) \right] = \frac{a_k a_j}{a_{\min(k,j)}}, \end{split}$$

which, by Lemma 4.3, equals $a_{\max(k,j)}$, as expected.

Define the matrix $T = [T_{kj}]$ with

$$T_{kj} = \begin{cases} 1 & \text{if } k \ge j, \\ 0 & \text{otherwise.} \end{cases}$$

It is easy to see that

$$T_{kj}^{-1} = \begin{cases} (-1)^{k+j} & \text{if } 0 \le k-j \le 1, \\ 0 & \text{otherwise.} \end{cases}$$

Thus the proofs related to inverse matrices L^{-1} , U^{-1} , A_N^{-1} and B_N^{-1} follow from Proposition 4.2.

In order to prove the LU-decomposition of $(M_1)_N^{-1}$, it is sufficient to show that $(M_1)_N = B_N^{-1} \cdot A_N^{-1}$. Consider

$$\sum_{d=\max(k,j)}^{N} B_{kd}^{-1} A_{dj}^{-1} = \sum_{d=\max(k,j)}^{N-1} (a_d - a_{d+1}) + a_N = a_{\max(k,j)},$$

as desired.

For the Cholesky decomposition, consider

$$\sum_{d=1}^{\min(k,j)} C_{kd} C_{jd} = \frac{a_k a_j}{a_1} + \sum_{d=2}^{\min(k,j)} \frac{a_k a_j}{a_d a_{d-1}} (a_{d-1} - a_d) = a_{\max(k,j)},$$

which completes the proof (Note that it can be also derived by the help of Corollary 4.1).

Finally, in order to prove the inverse matrix, we have three cases: j = 1, 1 < j < N and j = N. For these cases, consider the following equalities, respectively.

$$\sum_{d=1}^{N} (M_1)_{kd} (M_1^{-1})_{d1} = \frac{a_{\max(k,1)}}{a_1 - a_2} + \frac{a_{\max(k,2)}}{a_2 - a_1} = [k = 1],$$

$$\sum_{d=1}^{N} (M_1)_{kd} (M_1^{-1})_{dj} = \frac{a_{\max(k,j-1)}}{a_j - a_{j-1}} + \frac{a_{\max(k,j)} (a_{j-1} - a_{j+1})}{(a_{j+1} - a_j)(a_j - a_{j-1})} + \frac{a_{\max(k,j+1)}}{a_{j+1} - a_j} = [k = j],$$

$$\sum_{d=1}^{N} (M_1)_{kd} (M_1^{-1})_{dN} = \frac{a_{\max(k,N-1)}}{a_N - a_{N-1}} + \frac{a_{N-1} a_{\max(k,N)}}{a_N (a_{N-1} - a_N)} = [k = N].$$

By all of them, the proof is complete.

Corollary 4.5. For $N \geq 2$,

$$\det(M_1)_N = a_N \prod_{d=1}^{N-1} (a_d - a_{d+1})$$

and $\det(M_1)_1 = a_1$.

Proof. Since
$$\det(M_1)_N = \prod_{d=1}^N U_{dd}$$
, it is immediately seen.

For example, let T_1 be the matrix defined by $[\max(k,j)]_{1 \le k,j \le N}$. Then

$$\det T_1 = (-1)^{N-1} N.$$

One can easily obtain many special and nice examples. The evaluation of the determinants of these kinds of matrices by using other methods needs more effort.

Remark 4.1. If the sequence $\{a_n\}$ is positive and decreasing, then the matrix M_1 is a positive definite matrix, which can be easily seen by Corollary 4.5. On the other hand, the sequence $\{a_n\}$ is negative and increasing, then the matrix M_1 is a negative definite matrix.

Now we shall give the results for the reciprocal Max matrix \mathcal{M}_1 without proof because all of them could be seen by choosing reciprocal term in the results for the matrix M_1 . Studying with reciprocals could sometimes be more challenging. For this reason, we list the results for the quick access. We have the following results for the LU-decomposition, inverse matrix, Cholesky decomposition and LU-decomposition of the inverse matrix \mathcal{M}_1^{-1} , respectively.

Corollary 4.6. For $k, j \geq 1$,

$$\mathcal{L}_{kj} = \begin{cases} \frac{a_j}{a_k} & \text{if } k \ge j, \\ 0 & \text{otherwise} \end{cases}$$

and

$$\mathcal{U}_{kj} = \begin{cases} \frac{1}{a_j} & \text{if } k = 1, \\ \frac{a_k - a_{k-1}}{a_j a_k} & \text{if } j \ge k > 1, \\ 0 & \text{otherwise.} \end{cases}$$

For $1 \leq k, j \leq N$, $(\mathcal{M}_1^{-1})_N$ is the symmetric tridiagonal matrix defined by

$$(\mathcal{M}_{1}^{-1})_{kj} = \begin{cases} \frac{a_{1}a_{2}}{a_{2} - a_{1}} & \text{if } k = j = 1, \\ \frac{a_{k}^{2} (a_{k+1} - a_{k-1})}{(a_{k+1} - a_{k}) (a_{k} - a_{k-1})} & \text{if } 1 \neq k = j \neq N, \\ \frac{a_{N}^{2}}{a_{N} - a_{N-1}} & \text{if } k = j = N, \\ \frac{a_{k}a_{j}}{a_{k-1} - a_{k}} & \text{if } k = j + 1. \end{cases}$$

For $k, j \geq 1$, C is the lower triangular matrix defined by

$$C_{kj} = \begin{cases} \frac{\sqrt{a_1}}{a_k} & \text{if } j = 1, \\ \frac{\sqrt{a_j - a_{j-1}}}{a_k} & \text{if } j > 1. \end{cases}$$

For $1 \le k, j \le N$,

$$\mathcal{A}_{kj} = \begin{cases} (-1)^{k+j} & \text{if } 0 \le k-j \le 1, \\ 0 & \text{otherwise} \end{cases}$$

and

$$\mathcal{B}_{kj} = \begin{cases} (-1)^{k+j} \frac{a_{k+1} a_k}{a_{k+1} - a_k} & \text{if } 0 \le j - k \le 1, \\ a_N & \text{if } k = j = N, \\ 0 & \text{otherwise.} \end{cases}$$

4.4.2 Min Matrices and Their Reciprocal Analogues

Firstly, we list the LU-decomposition, inverse matrix, Cholesky decomposition of the matrix M_2 and LU-decomposition of the inverse matrix M_2^{-1} , respectively. We omit the results related to L^{-1} , U^{-1} , A^{-1} and B^{-1} here. They could be easily obtained as in the proof of the matrix M_1 by the help of Proposition 4.2.

Theorem 4.19. Take the matrix M_2 , for $k, j \geq 1$,

$$L_{kj} = \begin{cases} 1 & \text{if } k \geq j, \\ 0 & \text{otherwise}, \end{cases}$$

$$U_{kj} = \begin{cases} a_1 & \text{if } k = 1, \\ a_k - a_{k-1} & \text{if } j \geq k > 1, \\ 0 & \text{otherwise}, \end{cases}$$

$$(M_2^{-1})_{kj} = \begin{cases} \frac{a_2}{a_1(a_2 - a_1)} & \text{if } k = j = 1, \\ \frac{(a_{k+1} - a_{k-1})}{(a_{k+1} - a_k)(a_k - a_{k-1})} & \text{if } 1 \neq k = j \neq N, \end{cases}$$

$$\frac{1}{a_N - a_{N-1}} & \text{if } k = j = N,$$

$$\frac{1}{a_{k-1} - a_k} & \text{if } k = j + 1, \end{cases}$$

$$C_{kj} = \begin{cases} \sqrt{a_1} & \text{if } j = 1, \\ \sqrt{a_j - a_{j-1}} & \text{if } j > 1, \\ 0 & \text{otherwise}, \end{cases}$$

$$A_{kj} = \begin{cases} (-1)^{k+j} \frac{a_j}{a_k} & \text{if } 0 \leq k - j \leq 1, \\ 0 & \text{otherwise}, \end{cases}$$

$$B_{kj} = \begin{cases} (-1)^{k+j} \frac{a_{k+1}}{a_j(a_{k+1} - a_k)} & \text{if } 0 \leq j - k \leq 1, \\ \frac{1}{a_N} & \text{if } k = j = N, \end{cases}$$

$$0 & \text{otherwise}.$$

Note that the inverse matrix M_2^{-1} is a symmetric tridiagonal matrix of order N.

Corollary 4.7. Take the matrix \mathcal{M}_2 , for $k, j \geq 1$, we have

$$\mathcal{L}_{kj} = \begin{cases} 1 & if \ k \ge j, \\ 0 & otherwise, \end{cases}$$

$$\mathcal{U}_{kj} = \begin{cases} \frac{1}{a_1} & if \ k = 1, \\ \frac{a_{k-1} - a_k}{a_k a_{k-1}} & if \ j \ge k > 1, \\ 0 & otherwise, \end{cases}$$

$$(\mathcal{M}_{2}^{-1})_{kj} = \begin{cases} \frac{a_{1}^{2}}{a_{1} - a_{2}} & \text{if } k = j = 1, \\ \frac{a_{k}^{2} \left(a_{k-1} - a_{k+1}\right)}{\left(a_{k+1} - a_{k}\right) \left(a_{k} - a_{k-1}\right)} & \text{if } 1 \neq k = j \neq N, \\ \frac{a_{N}a_{N-1}}{a_{N-1} - a_{N}} & \text{if } k = j = N, \\ \frac{a_{k}a_{j}}{a_{k} - a_{k-1}} & \text{if } k = j + 1, \end{cases}$$

$$C_{kj} = \begin{cases} \frac{1}{a_{j}a_{j-1}} \sqrt{a_{j}a_{j-1} \left(a_{j-1} - a_{j}\right)} & \text{if } j = 1, \\ 0 & \text{otherwise}, \end{cases}$$

$$A_{kj} = \begin{cases} (-1)^{k+j} \frac{a_{k}}{a_{j}} & \text{if } 0 \leq k - j \leq 1, \\ 0 & \text{otherwise}, \end{cases}$$

$$\mathcal{B}_{kj} = \begin{cases} (-1)^{k+j} \frac{a_{k}a_{j}}{a_{j} \left(a_{k} - a_{k+1}\right)} & \text{if } 0 \leq j - k \leq 1, \\ a_{N} & \text{if } k = j = N, \\ 0 & \text{otherwise}. \end{cases}$$

Similarly, note that the inverse matrix \mathcal{M}_2^{-1} is a symmetric tridiagonal matrix of order N.

Proof. By Lemma 4.3, we can write

$$a_{\min(k,j)} = \frac{a_k a_j}{a_{\max(k,j)}} \text{ and } \frac{1}{a_{\min(k,j)}} = \frac{a_{\max(k,j)}}{a_k a_j}.$$

So all claimed results follow by Proposition 4.2 and the results for the matrices M_1 and \mathcal{M}_1 .

By the LU-decomposition of the matrix M_2 , we have the following corollary.

Corollary 4.8. For $N \geq 1$,

$$\det(M_2)_N = a_1 \prod_{d=1}^{N-1} (a_{d+1} - a_d).$$

Remark 4.2. By the above corollary, it is seen that if a_1 is a positive real number and the sequence $\{a_n\}$ is increasing, then the matrix M_2 is a positive definite matrix. Conversely, if a_1 is a negative real number and the sequence $\{a_n\}$ is decreasing, then the matrix M_2 is a negative definite matrix.

4.4.3 Further Applications

Recall the Lehmer matrix M defined by

$$M_{kj} = \frac{\min(k, j)}{\max(k, j)}.$$

By Lemma 4.3, one can write the (k, j)th entry of it as:

$$\frac{\min(k,j)}{\max(k,j)} = \frac{k \times j}{(\max(k,j))^2} = \frac{k \times j}{\max(k^2,j^2)}.$$

By using Proposition 4.2 and the results for the matrix \mathcal{M}_1 for $a_n = n^2$, i.e. $\max(k^2, j^2) = a_{\max(k,j)}$, it is easily rediscovered the LU-decomposition, inverse matrix and Cholesky decomposition of the Lehmer matrix. Also the results of [91, 92] for some recursive analogues of the Lehmer matrix can be retrieved by using similar approach.

Moreover, our results give us an idea to find a sequential generalization of the Lehmer matrix. For example, we define the matrix $H = [H_{kj}]$ for any positive and strictly increasing sequence $\{a_n\}$ by

$$H_{kj} = \frac{\min(a_k, a_j)}{\max(a_k, a_j)} = \frac{a_k a_j}{\max(a_k^2, a_j^2)}.$$

Thus by our results for the matrix \mathcal{M}_1 with the sequence $\{a_n^2\}$ and Proposition 4.2, the LU-decomposition, inverse and Cholesky decomposition of the matrix H could be derived but we omit the details here due to the similarities with the following corollary. Note that if the sequence $\{a_n\}$ is decreasing than we may also obtain sequential generalization of the Lehmer matrix by using the results for the matrix \mathcal{M}_2 .

Up to now, any reciprocal analogue of the Lehmer matrix has not been studied yet. The following corollary will be the first reciprocal-sequential generalization of the Lehmer matrix.

Corollary 4.9. Let $\{a_n\}$ be a positive and strictly increasing sequence and $\mathcal{H} = [\mathcal{H}_{kj}]$ be the matrix defined by

$$\mathcal{H}_{kj} = \frac{\max(a_k, a_j)}{\min(a_k, a_j)}.$$

Then

$$\mathcal{L}_{kj} = \begin{cases} \frac{a_k}{a_j} & \text{if } k \ge j, \\ 0 & \text{otherwise,} \end{cases}$$

$$\mathcal{U}_{kj} = \begin{cases} \frac{a_j}{a_1} & \text{if } k = 1, \\ \frac{a_j (b_{k-1} - b_k)}{a_k b_{k-1}} & \text{if } j \ge k > 1, \\ 0 & \text{otherwise,} \end{cases}$$

$$C_{kj} = \begin{cases} \frac{a_k}{b_j b_{j-1}} & \text{if } j = 1, \\ \frac{a_k}{b_j b_{j-1}} \sqrt{b_j b_{j-1} (b_{j-1} - b_j)} & \text{if } j > 1, \\ 0 & \text{otherwise,} \end{cases}$$

$$\mathcal{H}_{kj}^{-1} = \begin{cases} \frac{b_1}{b_1 - b_2} & \text{if } k = j = 1, \\ \frac{b_k (b_{k-1} - b_{k+1})}{(b_{k+1} - b_k) (b_k - b_{k-1})} & \text{if } 1 \ne k = j \ne N, \\ \frac{b_{N-1}}{b_{N-1} - b_N} & \text{if } k = j = N, \\ \frac{a_k a_j}{b_k - b_{k-1}} & \text{if } k = j + 1, \end{cases}$$
A symmetric tridiagonal matrix of order N and $b_n = a^2$

where \mathcal{H}^{-1} is a symmetric tridiagonal matrix of order N and $b_n = a_n^2$.

Proof. Since $\{a_n\}$ is a positive and strictly increasing, by Lemma 4.3, we have

$$\mathcal{H}_{kj} = \frac{\max(a_k, a_j)}{\min(a_k, a_j)} = \frac{a_k a_j}{b_{\min(k, j)}}.$$

So the proof follows by the results given for the matrix \mathcal{M}_2 with the sequence $\{b_n\}$ $\{a_n^2\}$ and Proposition 4.2.

Note that when $a_n = n$, we get the reciprocal analogue of the usual Lehmer matrix. Besides, one can also obtain related results when the sequence $\{a_n\}$ is decreasing by the help of the results for the matrix \mathcal{M}_1 .

Now we give an interesting and useful idea. Although, one can't directly use our main results to derive some results related to some certain kind of matrices, our results allow us to guess these results with less effort. We shall give an example to show this idea.

Corollary 4.10. For a positive integer r, define the matrix $\mathcal{F} = [\mathcal{F}_{kj}]$ by

$$\mathcal{F}_{kj} = \begin{cases} a_{\max(k,j)} & \text{if } k \ge j - r, \\ 0 & \text{otherwise.} \end{cases},$$

for a given sequence $\{a_n\}$. Then for $k, j \geq 1$, the LU-decomposition of the matrix \mathcal{F} is

$$\mathcal{L}_{kj} = \begin{cases} \frac{a_k}{a_j} & \text{if } k \ge j, \\ 0 & \text{otherwise} \end{cases}$$

and

$$\mathcal{U}_{kj} = \begin{cases} a_j & \text{if } k = 1 \text{ and } j \leq r+1, \\ a_j & \text{if } j > r+1 \text{ and } k = j-r, \\ \\ \frac{a_j (a_{k-1} - a_k)}{a_{k-1}} & \text{if } k+r-1 > j \geq k > 1, \\ 0 & \text{otherwise.} \end{cases}$$

Clearly, for N=8 and r=3, the matrix \mathcal{F} takes the form

$$\mathcal{F}_8 = \begin{bmatrix} a_1 & a_2 & a_3 & a_4 & 0 & 0 & 0 & 0 \\ a_2 & a_2 & a_3 & a_4 & a_5 & 0 & 0 & 0 \\ a_3 & a_3 & a_3 & a_4 & a_5 & a_6 & 0 & 0 \\ a_4 & a_4 & a_4 & a_4 & a_5 & a_6 & a_7 & 0 \\ a_5 & a_5 & a_5 & a_5 & a_5 & a_6 & a_7 & a_8 \\ a_6 & a_6 & a_6 & a_6 & a_6 & a_6 & a_7 & a_8 \\ a_7 & a_7 & a_7 & a_7 & a_7 & a_7 & a_7 & a_8 \\ a_8 & a_8 & a_8 & a_8 & a_8 & a_8 & a_8 & a_8 \end{bmatrix}.$$

As seen, the matrix \mathcal{F} is obtained from the matrix M_1 by deleting the entries after rth superdiagonal. (Note that similar example can be obtained for the matrix which is obtained by applying the same process to the matrix M_2).

Proof. We should show that

$$\mathcal{F}_{kj} = \sum_{d=1}^{\min(k,j)} \mathcal{L}_{kd} \mathcal{U}_{dj}.$$

The proof for the case $j \leq r+1$ can be similarly done as in the proof of the results for the matrix M_1 . Now consider for j > r+1 and k > j-r,

$$\sum_{d=1}^{\min(k,j)} \mathcal{L}_{kd} \mathcal{U}_{dj} = \frac{a_k a_j}{a_{j-r}} + \sum_{d=j-r+1}^{\min(k,j)} \mathcal{L}_{kd} \mathcal{U}_{dj} = \frac{a_k a_j}{a_{j-r}} + a_k a_j \sum_{d=j-r+1}^{\min(k,j)} \left(\frac{1}{a_d} - \frac{1}{a_{d-1}}\right)$$
$$= \frac{a_k a_j}{a_{\min(k,j)}} = a_{\max(k,j)}.$$

And the final case j > r+1 and $k \le j-r$ can be easily computed as 0, which completes the proof.

It is possible to obtain similar results for the matrix which is derived by deleting the entries of Max matrix (or Min matrix) after rth subdiagonal by the help of our results. In the light of all these, our results are very useful to define new families of combinatorial matrices.

As a conclusion remark, our results cover the results which will be valid for the matrices $[\max(a_k, a_j)]_{k,j\geq 1}$ and $[\min(a_k, a_j)]_{k,j\geq 1}$ when the sequence $\{a_n\}$ is increasing or decreasing. Note that the increasing case is given in [5] for only finite order matrices with a different approach. Unfortunately, if a sequence $\{c_n\}$ is neither increasing nor decreasing, such as unimodal sequences, then our results don't work for the matrices $[\max(c_k, c_j)]_{k,j\geq 1}$ and $[\min(c_k, c_j)]_{k,j\geq 1}$ and we could not find explicit results for such matrices.

4.5 A Non-symmetric Variant of the Filbert Matrix

As mentioned in Section 3.2.4, in this section we will study a new non-symmetric variant of Filbert matrix. We define our main matrix $M = [M_{kj}]_{k,j \geq 0}$ by

$$M_{kj} = \frac{1 - xq^{\lambda k - \mu j}}{1 - xq^{\lambda k + \mu j}},$$

where λ and μ are positive integers and x is a real number such that $x \neq q^{-\lambda k - \mu j}$ for all $k, j \geq 0$. Here we would like to point out indexes of the entries of the matrix M start from 0. Otherwise related matrix would be singular that makes no sense to study.

We will derive explicit formulæ for the LU-decomposition, inverse matrices L^{-1} and U^{-1} , and inverse of the matrix M. Our approach is mainly to guess the relevant quantities. Afterwards, we will provide proofs of these formulæ. It is worthwhile to note that, although all the sum identities we need to prove seem to be Gosper-summable, the q-Zeilberger algorithm does not work for the general parameters λ and μ . The algorithm can only compute the specialized sums for some fixed special numerical values of λ and μ . But it is not enough to prove general results. For this reason, we will use some traditional ways which cause to guess some new sum identities with one additional parameter. Finally, as applications, we will give some particular results related to the generalized Fibonacci and Lucas numbers as non-symmetric variants of Filbert and Lilbert matrices for the special choices of q and x.

4.5.1 Main Results

Here, we will list the LU-decomposition of the matrix M, inverse matrices L^{-1} , U^{-1} , determinant and inverse matrix M^{-1} . In Section 4.5.2, we will provide the proofs of these results. Recall that the matrix M would be never a symmetric matrix for any choice of the parameters. For this reason, we can not talk about the Cholesky decomposition.

Theorem 4.20. For $k, j \ge 0$,

$$L_{kj} = \frac{(xq^{\lambda j + \mu}; q^{\mu})_j (q^{\lambda(k-j+1)}; q^{\lambda})_j}{(xq^{\lambda k + \mu}; q^{\mu})_j (q^{\lambda}; q^{\lambda})_j}$$

and

$$U_{kj} = \begin{cases} \frac{1 - xq^{-\mu j}}{1 - xq^{\mu j}} & \text{if } k = 0, \\ q^{-\mu j + (\lambda + \mu)\binom{k}{2}} x^k (1 + q^{\mu j}) \frac{(q^{\mu(j-k+1)}; q^{\mu})_k (q^{\lambda}; q^{\lambda})_k}{(xq^{\mu j}; q^{\lambda})_{k+1} (xq^{\lambda k + \mu}; q^{\mu})_{k-1}} & \text{if } k > 0. \end{cases}$$

Theorem 4.21. For N > 1,

$$\det M_N = x^{\binom{N}{2}} q^{-\mu \binom{N}{2}} \frac{(-q^{\mu}; q^{\mu})_{N-1} (xq^{\lambda}; q^{\lambda})_{N-1}}{(xq^{\lambda+\mu}; q^{\lambda+\mu})_{N-1}} \prod_{d=1}^{N-1} q^{(\lambda+\mu)\binom{d}{2}} \frac{(q^{\mu}; q^{\mu})_d (q^{\lambda}; q^{\lambda})_d}{(xq^{\mu d}; q^{\lambda})_d (xq^{\lambda d}; q^{\mu})_d}$$

and $\det M_1 = 1$.

Theorem 4.22. For $k, j \geq 0$,

$$L_{kj}^{-1} = (-1)^{k+j} q^{\lambda \binom{k-j}{2}} \frac{(xq^{\lambda j+\mu}; q^{\mu})_{k-1} (q^{\lambda (k-j+1)}; q^{\lambda})_j}{(xq^{\lambda k+\mu}; q^{\mu})_{k-1} (q^{\lambda}; q^{\lambda})_j}$$

and

$$U_{kj}^{-1} = \begin{cases} 1 & \text{if } k = j = 0, \\ \frac{q^{-\lambda\binom{j}{2}}(-1)^{j+1}(xq^{\lambda j+\mu};q^{\mu})_{j}}{x^{j}(q^{\lambda};q^{\lambda})_{j}} & \text{if } j \geq 1 \text{ and } k = 0, \\ \times \sum_{t=1}^{j} \frac{q^{\mu(\binom{t+1}{2})+t-tj}(-1)^{t}(1-xq^{-\mu t})(xq^{\mu t};q^{\lambda})_{j}}{(1-xq^{\mu t})(1-q^{2\mu t})(q^{\mu};q^{\mu})_{j-t}(q^{\mu};q^{\mu})_{t-1}} & \text{if } j \geq 1 \text{ and } k = 0, \\ (-1)^{k+j} \frac{q^{-\lambda\binom{j}{2}+\mu(\binom{k+1}{2})+k-kj}}{x^{j}(1-q^{2\mu k})} \frac{(xq^{\mu k};q^{\lambda})_{j}(xq^{\lambda j+\mu};q^{\mu})_{j}}{(q^{\lambda};q^{\lambda})_{j}(q^{\mu};q^{\mu})_{j-k}(q^{\mu};q^{\mu})_{k-1}} & \text{if } j \geq k \geq 1, \\ 0 & \text{otherwise.} \end{cases}$$

For the inverse matrix M_N^{-1} , we have the following result.

Theorem 4.23. For $1 \le k < N \text{ and } 0 \le j < N$,

$$\begin{split} M_{kj}^{-1} &= \frac{(-1)^{k+j}}{x^{N-1}} \frac{q^{\lambda \binom{j}{2} + \mu \binom{k+1}{2} - (N-2)(\lambda j + \mu k)}}{(1 - xq^{\mu k + \lambda j})(1 - q^{2\mu k})} \\ &\times \frac{(xq^{\lambda j + \mu}; q^{\mu})_{N-1} (xq^{\mu k}; q^{\lambda})_{N}}{(q^{\mu}; q^{\mu})_{N-k-1} (q^{\lambda}; q^{\lambda})_{N-j-1} (q^{\lambda}; q^{\lambda})_{j} (q^{\mu}; q^{\mu})_{k-1}} \end{split}$$

and for $0 \le j < N$,

$$\begin{split} M_{0j}^{-1} &= [j=0] + (-1)^{j+1} q^{\lambda \binom{j}{2} - \lambda(N-2)j} \frac{x^{N-1} (xq^{\lambda j + \mu}; q^{\mu})_{N-1}}{(q^{\lambda}; q^{\lambda})_{N-j-1} (q^{\lambda}; q^{\lambda})_{j}} \\ &\times \sum_{t=1}^{N-1} \frac{1 - xq^{-\mu t}}{1 - xq^{\mu t}} \frac{(-1)^{t} q^{\mu \binom{t+1}{2} - \mu(N-2)t}}{1 - q^{2\mu t}} \frac{(xq^{\mu t}; q^{\lambda})_{N}}{(1 - xq^{\mu t + \lambda j}) (q^{\mu}; q^{\mu})_{N-t-1} (q^{\mu}; q^{\mu})_{t-1}}. \end{split}$$

Here note that the entries except the first row of the matrices U^{-1} and M^{-1} can be nicely factorized but the first row of these matrices can't be factorized. The guessing procedure of these entries is extremely time consuming.

Since our matrix is not symmetric, the results related to its transposed matrix $M^T = \left[\frac{1-xq^{\lambda j-\mu k}}{1-xq^{\lambda j+\mu k}}\right]_{k,j\geq 0}$ may yield new results for a new matrix family. By Propositions 4.2 and 4.3 and the fact $(M^T)^{-1}=(M^{-1})^T$, we have the following theorem for the matrix M^T .

Theorem 4.24. For $k, j \geq 0$

$$L_{kj} = \begin{cases} \frac{1 - xq^{-\mu k}}{1 - xq^{\mu k}} & \text{if } j = 0, \\ q^{\mu(j-k)} \frac{(1 + q^{\mu k})}{(1 + q^{\mu j})} \frac{(q^{\mu(k-j+1)}; q^{\mu})_j (xq^{\mu j}; q^{\lambda})_{j+1}}{(xq^{\mu k}; q^{\lambda})_{j+1} (q^{\mu}; q^{\mu})_j} & \text{if } j > 0 \end{cases}$$

and

$$U_{kj} = q^{-\mu k + (\lambda + \mu)\binom{k}{2}} x^k (1 + q^{\mu k}) \frac{(q^{\mu}; q^{\mu})_k (q^{\lambda(j-k+1)}; q^{\lambda})_k}{(xq^{\lambda j + \mu}; q^{\mu})_k (xq^{\mu k}; q^{\lambda})_k}.$$

For $k, j \geq 0$,

$$L_{kj}^{-1} = \begin{cases} 1 & \text{if } k = j = 0, \\ \frac{q^{-\mu k + \mu\binom{k}{2}}(-1)^{k+1}(1 + q^{\mu k})(q^{\mu}; q^{\mu})_k}{(xq^{\mu k}; q^{\lambda})_k} & \text{if } k \geq 1 \text{ and } j = 0, \\ \times \sum_{t=1}^k \frac{q^{\mu(\binom{t+1}{2}) + t - tk}(-1)^t(1 - xq^{-\mu t})(xq^{\mu t}; q^{\lambda})_k}{(1 - xq^{\mu t})(1 - q^{2\mu t})(q^{\mu}; q^{\mu})_{k-t}(q^{\mu}; q^{\mu})_{t-1}} & \text{if } k \geq 1 \text{ and } j = 0, \\ \frac{(-1)^{k+j}(1 + q^{\mu k})}{q^{-\mu\binom{k}{2} - \mu(\binom{j+1}{2}) + j - k - kj}(1 - q^{2\mu j})} \frac{(q^{\mu(k-j+1)}; q^{\mu})_j(xq^{\lambda k + \mu}; q^{\mu})_k}{(q^{\mu}; q^{\mu})_{j-1}(xq^{\lambda j + \mu}; q^{\mu})_{k-1}} & \text{if } k \geq j \geq 1, \\ 0 & \text{otherwise} \end{cases}$$

and for $j \geq k$,

$$U_{kj}^{-1} = \frac{(-1)^{k+j} q^{\mu j - \mu\binom{j}{2} + \lambda\binom{k+1}{2} - \lambda kj}}{x^j (1 + q^{\mu j})} \frac{(xq^{\mu j}; q^{\lambda})_{j+1} (xq^{\lambda k + \mu}; q^{\mu})_{j-1}}{(q^{\mu}; q^{\mu})_j (q^{\lambda}; q^{\lambda})_{j-k} (q^{\lambda}; q^{\lambda})_k}$$

and 0 otherwise. Finally, for $1 \le j < N$ and $0 \le k < N$,

$$(M^{T})_{kj}^{-1} = \frac{(-1)^{k+j}}{x^{N-1}} \frac{q^{\lambda \binom{k}{2} + \mu \binom{j+1}{2} - (N-2)(\lambda k + \mu j)}}{(1 - xq^{\mu j + \lambda k})(1 - q^{2\mu j})} \times \frac{(xq^{\lambda k + \mu}; q^{\mu})_{N-1} (xq^{\mu j}; q^{\lambda})_{N}}{(q^{\mu}; q^{\mu})_{N-j-1} (q^{\lambda}; q^{\lambda})_{N-k-1} (q^{\lambda}; q^{\lambda})_{k} (q^{\mu}; q^{\mu})_{j-1}}$$

and for $0 \le k < N$,

$$\begin{split} (M^T)_{k0}^{-1} &= [k=0] + (-1)^{k+1} q^{\lambda {k \choose 2} - \lambda (N-2)k} \frac{x^{N-1} (xq^{\lambda k + \mu}; q^{\mu})_{N-1}}{(q^{\lambda}; q^{\lambda})_{N-k-1} (q^{\lambda}; q^{\lambda})_k} \\ &\times \sum_{t=1}^{N-1} \frac{1 - xq^{-\mu t}}{1 - xq^{\mu t}} \frac{(-1)^t q^{\mu {t+1 \choose 2} - \mu (N-2)t}}{1 - q^{2\mu t}} \frac{(xq^{\mu t}; q^{\lambda})_N}{(1 - xq^{\mu t + \lambda k}) (q^{\mu}; q^{\mu})_{N-t-1} (q^{\mu}; q^{\mu})_{t-1}}. \end{split}$$

Now we can proceed with the proofs of the above theorems.

4.5.2 Proofs

Define the following four sums:

$$S_{1}(K) = \sum_{d=K}^{\min(k,j)} q^{(\lambda+\mu)\binom{d}{2}} x^{d} (1 - xq^{d(\lambda+\mu)}) \frac{(q^{\lambda(k-d+1)}; q^{\lambda})_{d} (q^{\mu(j-d+1)}; q^{\mu})_{d-1}}{(xq^{\lambda k+\mu}; q^{\mu})_{d} (xq^{\mu j}; q^{\lambda})_{d+1}},$$

$$S_{2}(K) = \sum_{d=j}^{K} (-1)^{d} q^{\lambda\binom{d-j}{2}} (1 - xq^{d(\lambda+\mu)}) \frac{(xq^{\lambda j+\mu}; q^{\mu})_{d-1} (q^{\lambda(k-d)+\lambda}; q^{\lambda})_{d}}{(xq^{\lambda k+\mu}; q^{\mu})_{d} (q^{\lambda}; q^{\lambda})_{d-j}},$$

$$S_{3}(K) = \sum_{d=k}^{K} (-1)^{d} q^{\mu\binom{d}{2}-\mu k d} (1 - xq^{d(\lambda+\mu)}) \frac{(xq^{\mu k}; q^{\lambda})_{d} (q^{\mu(j-d)+\mu}; q^{\mu})_{d}}{(xq^{\mu j}; q^{\lambda})_{d+1} (q^{\mu}; q^{\mu})_{d-k}}$$

and

$$S_4(K) = \sum_{d=\max(k,j)}^{K} q^{-\mu k d - \lambda j d} x^{-d} (1 - x q^{d(\lambda + \mu)}) \frac{(x q^{\mu k}; q^{\lambda})_d (x q^{\lambda j + \mu}; q^{\mu})_{d-1}}{(q^{\mu}; q^{\mu})_{d-k} (q^{\lambda}; q^{\lambda})_{d-j}}.$$

We provide the following lemmas for later use.

Lemma 4.4.

$$S_1(K) = x^K q^{(\lambda+\mu)\binom{K}{2}} \frac{(q^{\lambda(k-K+1)}; q^{\lambda})_K (q^{\mu(j-K+1)}; q^{\mu})_{K-1}}{(1 - xq^{\lambda k+\mu j})(xq^{\mu j}; q^{\lambda})_K (xq^{\lambda k+\mu}; q^{\mu})_{K-1}}.$$

Proof. We will use the backward induction method. Let us denote the summand term by s_d for brevity.

Firstly, assume that $k \geq j$ so when K = j the claim is obvious. Similarly for the case j > k, the initial claim is clear.

The backward induction step amounts to show that

$$S_{1}(K-1)$$

$$= S_{1}(K) + s_{K-1}$$

$$= x^{K} q^{(\lambda+\mu)\binom{K}{2}} \frac{(q^{\lambda(k-K+1)}; q^{\lambda})_{K} (q^{\mu(j-K+1)}; q^{\mu})_{K-1}}{(1 - xq^{\lambda k + \mu j})(xq^{\mu j}; q^{\lambda})_{K} (xq^{\lambda k + \mu}; q^{\mu})_{K-1}}$$

$$+ q^{(\lambda+\mu)\binom{K-1}{2}} x^{K-1} (1 - xq^{(K-1)(\lambda+\mu)}) \frac{(q^{\lambda(k-K+2)}; q^{\lambda})_{K-1} (q^{\mu(j-K+2)}; q^{\mu})_{K-2}}{(xq^{\lambda k + \mu}; q^{\mu})_{K-1} (xq^{\mu j}; q^{\lambda})_{K}}$$

$$= x^{K-1} q^{(\lambda+\mu)\binom{K-1}{2}} \frac{(q^{\lambda(k-K+2)}; q^{\lambda})_{K-1} (q^{\mu(j-K+2)}; q^{\mu})_{K-2}}{(1 - xq^{\lambda k + \mu j})(xq^{\mu j}; q^{\lambda})_{K} (xq^{\lambda k + \mu}; q^{\mu})_{K-1}}$$

$$\times (xq^{(\lambda+\mu)(K-1)} (1 - q^{\lambda j - \lambda(K-1)}) (1 - q^{\mu j - \mu k + \mu}) + (1 - xq^{\lambda k + \mu j}) (1 - xq^{(K-1)(\lambda+\mu)})).$$

After some simplifications, the expression in the last line can be rewritten as

$$(1 - xq^{\lambda k + \mu(K-1)})(1 - xq^{\mu j + \lambda(K-1)}).$$

Finally,

$$S_1(K-1) = x^{K-1} q^{(\lambda+\mu)\binom{K-1}{2}} \frac{(q^{\lambda(k-K+2)}; q^{\lambda})_{K-1} (q^{\mu(j-K+2)}; q^{\mu})_{K-2}}{(1 - xq^{\lambda k + \mu j})(xq^{\mu j}; q^{\lambda})_{K-1} (xq^{\lambda k + \mu}; q^{\mu})_{K-2}},$$

which completes the proof.

Lemma 4.5. *For* k > j,

$$S_2(K) = (-1)^K q^{\lambda \binom{K-j+1}{2}} \frac{(xq^{\lambda j+\mu}; q^{\mu})_K (q^{\lambda (k-K)}; q^{\lambda})_{K+1}}{(1-q^{\lambda (k-j)})(xq^{\lambda k+\mu}; q^{\mu})_K (q^{\lambda}; q^{\lambda})_{K-j}}.$$

Proof. This time we will use the usual induction method. Similarly, we denote the summand term by s_d . The initial case K = j is easily verified. So, the induction step amounts to show that

$$S_2(K+1) = S_2(K) + s_{K+1}.$$

Consider

$$S_2(K) + s_{K+1} = (-1)^K q^{\lambda \binom{K-j+1}{2}} \frac{(xq^{\lambda j+\mu}; q^{\mu})_K (q^{\lambda (k-K)}; q^{\lambda})_{K+1}}{(1 - q^{\lambda (k-j)})(xq^{\lambda k+\mu}; q^{\mu})_K (q^{\lambda}; q^{\lambda})_{K-j}}$$

$$+ (-1)^{K+1} q^{\lambda \binom{K+1-j}{2}} (1 - xq^{(K+1)(\lambda+\mu)}) \frac{(xq^{\lambda j+\mu}; q^{\mu})_K (q^{\lambda (k-K)}; q^{\lambda})_{K+1}}{(xq^{\lambda k+\mu}; q^{\mu})_{K+1} (q^{\lambda}; q^{\lambda})_{K+1-j}}$$

$$= (-1)^{K+1} q^{\lambda \binom{K-j+1}{2}} \frac{(xq^{\lambda j+\mu}; q^{\mu})_K (q^{\lambda (k-K)}; q^{\lambda})_{K+1}}{(1 - q^{\lambda (k-j)}) (xq^{\lambda k+\mu}; q^{\mu})_{K+1} (q^{\lambda}; q^{\lambda})_{K+1-j}}$$

$$\times ((1 - xq^{(K+1)(\lambda+\mu)}) (1 - q^{\lambda (k-j)}) - (1 - xq^{\lambda k+\mu(K+1)}) (1 - q^{\lambda (K+1-j)}))$$

$$= (-1)^{K+1} q^{\lambda \binom{K-j+1}{2}} \frac{(xq^{\lambda j+\mu}; q^{\mu})_K (q^{\lambda (k-K)}; q^{\lambda})_{K+1}}{(1 - q^{\lambda (k-j)}) (xq^{\lambda k+\mu}; q^{\mu})_{K+1} (q^{\lambda}; q^{\lambda})_{K+1-j}}$$

$$\times q^{\lambda (K-j+1)} (1 - xq^{\lambda j+\mu(K+1)}) (1 - q^{\lambda (k-K)-\lambda}),$$

which is equal to $S_2(K+1)$, as desired.

Lemma 4.6. For j > k,

$$S_3(K) = (-1)^K q^{\mu\binom{K}{2} + \mu(K(1-k)-k)} \frac{(xq^{\mu k}; q^{\lambda})_{K+1} (q^{\mu(j-k+1)}; q^{\mu})_k (q^{\mu(j-K)}; q^{\mu})_{K-k}}{(xq^{\mu j}; q^{\lambda})_{K+1} (q^{\mu}; q^{\mu})_{K-k}}.$$

The proof of Lemma 4.6 can be similarly done as in the proof of Lemma 4.5. We do not give it to avoid repetition.

Lemma 4.7.

$$S_4(K) = q^{-\mu Kk - \lambda Kj} x^{-K} \frac{(xq^{\lambda j + \mu}; q^{\mu})_K (xq^{\mu k}; q^{\lambda})_{K+1}}{(1 - xq^{\mu k + \lambda j})(q^{\mu}; q^{\mu})_{K-k} (q^{\lambda}; q^{\lambda})_{K-j}}.$$

Proof. Consider the case $k \geq j$,

$$S_4(k) = \sum_{k}^{k} q^{-\mu k^2 - \lambda j k} x^{-k} (1 - x q^{k(\lambda + \mu)}) \frac{(x q^{\mu k}; q^{\lambda})_k (x q^{\lambda j + \mu}; q^{\mu})_{k-1}}{(q^{\lambda}; q^{\lambda})_{k-j}}$$
$$= q^{-\mu k^2 - \lambda k j} x^{-k} \frac{(x q^{\lambda j + \mu}; q^{\mu})_k (x q^{\mu k}; q^{\lambda})_{k+1}}{(1 - x q^{\mu k + \lambda j}) (q^{\lambda}; q^{\lambda})_{k-j}},$$

as claimed. Similarly, the case k < j is obvious. Thus the initial claim is completed. Let's denote the summand by s_d . Then $S_4(K) + s_{K+1}$ equals

$$q^{-\mu kK - \lambda jK} x^{-K} \frac{(xq^{\lambda j + \mu}; q^{\mu})_K (xq^{\mu k}; q^{\lambda})_{K+1}}{(1 - xq^{\mu k + \lambda j})(q^{\mu}; q^{\mu})_{K-k} (q^{\lambda}; q^{\lambda})_{K-j}}$$

$$+ q^{-\mu k(K+1) - \lambda j(K+1)} x^{-K-1} (1 - xq^{(K+1)(\lambda + \mu)}) \frac{(xq^{\lambda j + \mu}; q^{\mu})_K (xq^{\mu k}; q^{\lambda})_{K+1}}{(q^{\mu}; q^{\mu})_{K+1-k} (q^{\lambda}; q^{\lambda})_{K+1-j}}$$

$$= q^{-\mu k(K+1) - \lambda j(K+1)} x^{-K-1} \frac{(xq^{\lambda j + \mu}; q^{\mu})_K (xq^{\mu k}; q^{\lambda})_{K+1}}{(1 - xq^{\mu k + \lambda j})(q^{\mu}; q^{\mu})_{K-k} (q^{\lambda}; q^{\lambda})_{K-j}}$$

$$+ (xq^{\mu k + \lambda j} (1 - q^{\mu (K+1-k)})(1 - q^{\lambda (K+1-j)}) + (1 - xq^{\mu k + \lambda j})(1 - xq^{(K+1)(\lambda + \mu)})).$$

After simplifications, the last line of the above equation is

$$(1 - xq^{\lambda j + \mu(K+1)})(1 - xq^{\mu k + \lambda(K+1)}).$$

Finally, $S_4(K) + s_{K+1} = S_4(K+1)$, which completes the proof.

Now we can give the proofs of our main results.

For the LU-decomposition of the matrix M, we have to prove that

$$\sum_{0 \le d \le \min(k,j)} L_{kd} U_{dj} = M_{kj}.$$

By Lemma 4.4, we obtain

$$\sum_{0 \le d \le \min(k,j)} L_{kd} U_{dj} = \frac{1 - xq^{-\mu j}}{1 - xq^{\mu j}} + q^{-\mu j} (1 - q^{2\mu j}) S_1(1)$$

$$= \frac{1 - xq^{-\mu j}}{1 - xq^{\mu j}} + xq^{-\mu j} \frac{(1 - q^{2\mu j})(1 - q^{\lambda k})}{(1 - xq^{\lambda k + \mu j})(1 - xq^{\mu j})}$$

$$= \frac{1 - xq^{\lambda k - \mu j}}{1 - xq^{\lambda k + \mu j}},$$

which completes the proof. As mentioned before, the q-Zeilberger algorithm can only compute the related sum for the special numerical values of λ and μ . For example, when $\lambda = \mu = 1$, the algorithm computes the sum $S_1(1)$ as

$$\frac{x(1-q^k)}{(1-xq^j)(1-xq^{k+j})}.$$

Then the claim is done for the case $\lambda = \mu = 1$. But in general we can not use this algorithm to prove the claim. That is why we need the previous lemmas.

For L and L^{-1} , it is obvious that $L_{kk}L_{kk}^{-1}=1$. For k>j,

$$\sum_{j < d < k} L_{kd} L_{dj}^{-1} = \frac{(-1)^j}{(q^{\lambda}; q^{\lambda})_j} S_2(k),$$

which equals 0 by Lemma 4.5. So we conclude

$$\sum_{j < d < k} L_{kd} L_{dj}^{-1} = [k = j],$$

as desired.

Before moving on, notice that the matrices U^{-1} and M^{-1} can be also written as follows:

$$U^{-1} = PR \text{ and } M_N^{-1} = P_N E_N,$$

where the matrix P is defined by

$$P_{00} = 1$$
 and $P_{0j} = -\frac{1 - xq^{-\mu j}}{1 - xq^{\mu j}}$ for $j > 0$,

$$P_{kj} = [k = j]$$
 for $j \ge 0$ and $k \ge 1$,

and

$$R_{00} = 1$$
 and $R_{0j} = 0$ for $j > 0$ and $R_{kj} = U_{kj}^{-1}$ for $j \ge k \ge 1$, $E_{00} = 1$ and $E_{0j} = 0$ for $0 < j < N$, and, $E_{kj} = M_{kj}^{-1}$ otherwise.

It is easily seen that the inverse matrix P^{-1} is computed as

$$P_{00}^{-1} = 1$$
 and $P_{0j}^{-1} = \frac{1 - xq^{-\mu j}}{1 - xq^{\mu j}}$ for $j > 0$, $P_{kj}^{-1} = [k = j]$ for $j \ge 0$ and $k \ge 1$.

In order to show that $U^{-1}U = I$, we will show that PRU = I. Consider the product matrix RU. The first row of this matrix is the same as the first row of the matrix U. Then for $k \geq 1$, obviously $R_{kk}U_{kk} = 1$, so when $k \neq j$ we have

$$\sum_{k \le d \le j} R_{kd} U_{dj} = (-1)^k q^{-\mu j + \frac{1}{2}\mu k(k+3)} \frac{1 + q^{\mu j}}{(1 - q^{2\mu k})(q^{\mu}; q^{\mu})_{k-1}} S_3(j) = 0,$$

which gives $RU = P^{-1}$, so the claim follows.

Finally, for the inverse matrix M_N^{-1} , we use the fact $M_N^{-1} = U_N^{-1} L_N^{-1} = P_N R_N L_N^{-1}$. The first row of the multiplication $R_N L_N^{-1}$ is [j=0] for $0 \le j \le N-1$. For $k \ge 1$, by Lemma 4.7, we obtain

$$\sum_{\max(k,j) \le d \le N-1} R_{kd} L_{dj}^{-1} = \frac{(-1)^{k+j} q^{\mu\binom{k+1}{2} + \lambda\binom{j}{2}}}{(1-q^{2\mu j})(q^{\lambda}; q^{\lambda})_j (q^{\mu}; q^{\mu})_{k-1}} S_4(N-1) = M_{kj}^{-1}.$$

So $R_N L_N^{-1} = E_N$, which completes the proof.

As a consequence of the LU-decomposition, the determinant of M_N is easily evaluated as the product of the diagonal entries of the matrix U. So the claim follows after some simplifications. The results for the matrix M^T follow by Proposition 4.3 after performing some simplifications.

4.5.3 Applications

In this subsection, we will give some applications of our main results. For example, consider the matrix \mathcal{F} , defined by

$$\mathcal{F}_{kj} = \frac{U_{\lambda k - \mu j + r}}{U_{\lambda k + \mu j + r}}$$

for positive integers λ, μ and r, where $\{U_n\}$ is the generalized Fibonacci sequence. By (2.6), the entries of the matrix \mathcal{F} can be rewritten as

$$\mathcal{F}_{kj} = q^{\mu j} (-1)^{\mu j} \frac{1 - q^{\lambda k - \mu j + r}}{1 - q^{\lambda k + \mu j + r}},$$

where $q = \beta/\alpha$. As seen, it is not directly obtained from the matrix M. However for $x = q^r$ and $q = \beta/\alpha$, we can write

$$\mathcal{F} = M \cdot D(q^{\mu n}(-1)^{\mu n}),$$

where $D(a_n)$ is the diagonal matrix defined as before. So by Proposition 4.2, we can easily derive all related results for the matrix \mathcal{F} from the results of the matrix M.

Note that an interesting feature of the matrix \mathcal{F} is that it includes some zero terms as entries. Especially, when $\lambda = \mu = 1$, then the entries on the rth superdiagonal are all zero. Similarly, we can find the results for the transposed matrix \mathcal{F}^T . This transposed matrix have zeros on the rth subdiagonal when $\lambda = \mu = 1$.

We find the LU-decomposition of the matrix \mathcal{F} and inverse matrices L^{-1} , U^{-1} and \mathcal{F}^{-1} as follows:

Corollary 4.11. For $k, j \geq 0$,

$$\mathcal{L}_{kj} = \frac{\left(\prod_{d=1}^{j} U_{\lambda j + \mu d + r}\right) \left(\prod_{d=1}^{j} U_{\lambda (k+1) - \lambda d}\right)}{\left(\prod_{d=1}^{j} U_{\lambda k + \mu d + r}\right) \left(\prod_{d=1}^{j} U_{\lambda d}\right)},$$

$$(4.5)$$

$$\mathcal{U}_{kj} = \begin{cases}
\frac{U_{-\mu j+r}}{U_{\mu j+r}} & if \ k = 0, \\
(-1)^{\mu j+(\lambda+\mu)\binom{k}{2}+rk} U_{2\mu j} \frac{\binom{k-1}{d-1} U_{\mu j-\mu d}}{\binom{k+1}{d-1} U_{\mu j+\lambda(d-1)+r}} \binom{k-1}{d-1} U_{\lambda k+\mu d+r}} & if \ k > 0,
\end{cases}$$

$$\mathcal{L}_{kj}^{-1} = (-1)^{k+j+\lambda\binom{k-j}{2}} \frac{\binom{k-1}{d-1} U_{\lambda j+\mu d+r}}{\binom{k-1}{d-1} U_{\lambda k+\mu d+r}} \binom{j}{m-1} U_{\lambda k+\mu d+r}},$$

$$\binom{k-1}{m-1} U_{\lambda k+\mu d+r} \binom{j}{m-1} U_{\lambda k},$$

and
$$\mathcal{U}_{kj}^{-1} =$$

$$\begin{cases} 1 & \text{if } k = j = 0, \\ (-1)^{k+j+\lambda\binom{j}{2}+rj+\mu kj+\mu\binom{k}{2}} \frac{1}{U_{2\mu k}} \frac{\left(\prod_{d=1}^{j} U_{\mu k+\lambda(d-1)+r}\right) \left(\prod_{d=1}^{j} U_{\lambda j+\mu d+r}\right)}{\left(\prod_{d=1}^{j} U_{\lambda d}\right) \left(\prod_{d=1}^{j-k} U_{\mu d}\right) \left(\prod_{d=1}^{k-1} U_{\mu d}\right)} & \text{if } j \geq k \geq 1, \end{cases}$$

and for $j \geq 1$,

$$\mathcal{U}_{0j}^{-1} = (-1)^{j+1+rj+\lambda\binom{j}{2}} \frac{\left(\prod_{d=1}^{j} U_{\lambda j + \mu d + r}\right)}{\left(\prod_{d=1}^{j} U_{\lambda d}\right)}$$

$$\times \sum_{t=1}^{j} (-1)^{t+\mu t j + \mu\binom{t}{2}} \frac{U_{-\mu t + r}}{U_{2\mu t} U_{\mu t + r}} \frac{\left(\prod_{d=1}^{j} U_{\mu t + \lambda (d-1) + r}\right)}{\left(\prod_{d=1}^{j-t} U_{\mu d}\right) \left(\prod_{d=1}^{t-1} U_{\mu d}\right)}$$

and 0 otherwise.

For the inverse matrix of order N, we have for $1 \le k < N$ and $0 \le j < N$,

$$\mathcal{F}_{kj}^{-1} = \frac{(-1)^{k+j+\lambda\binom{j}{2}+\mu\binom{k+1}{2}+N(\lambda j+\mu k)+r(N-1)}}{U_{2\mu k}U_{\mu k+\lambda j+r}} \times \frac{\left(\prod_{d=1}^{N} U_{\mu k+\lambda (d-1)+r}\right) \left(\prod_{d=1}^{N-1} U_{\lambda j+\mu d+r}\right)}{\left(\prod_{d=1}^{N-k-1} U_{\mu d}\right) \left(\prod_{d=1}^{N-j-1} U_{\lambda d}\right) \left(\prod_{d=1}^{j} U_{\lambda d}\right) \left(\prod_{d=1}^{k-1} U_{\mu d}\right)}$$

and for $0 \le j < N$,

$$\mathcal{F}_{0j}^{-1} = [j = 0] - \sum_{t=1}^{N-1} \frac{U_{-\mu t + r}}{U_{\mu t + r}} \mathcal{F}_{tj}^{-1}.$$

Proof. We only give the proof of the LU-decomposition. The others are very similar and the application of Proposition 4.2.

Since when $q = \beta/\alpha$ and $x = q^r$,

$$\mathcal{F}_{kj} = (-1)^{\mu j} q^{\mu j} M_{kj},$$

by Proposition 4.2 when $s_k = 1$ and $m_j = (-1)^{\mu j} q^{\mu j}$ and Theorem 4.20, we obtain the LU-decomposition of the matrix \mathcal{F} as follows

$$\mathcal{L}_{kj} = \frac{(q^{\lambda j + \mu + r}; q^{\mu})_j (q^{\lambda(k-j+1)}; q^{\lambda})_j}{(q^{\lambda k + \mu + r}; q^{\mu})_j (q^{\lambda}; q^{\lambda})_j}$$

and

$$\mathcal{U}_{kj} = \begin{cases}
\frac{1 - q^{-\mu j + r}}{1 - q^{\mu j + r}} & \text{if } k = 0, \\
(-1)^{\mu j} q^{(\lambda + \mu)\binom{k}{2} + rk} (1 + q^{\mu j}) \frac{(q^{\mu(j - k + 1)}; q^{\mu})_k (q^{\lambda}; q^{\lambda})_k}{(q^{\mu j + r}; q^{\lambda})_{k + 1} (q^{\lambda k + \mu + r}; q^{\mu})_{k - 1}} & \text{if } k > 0,
\end{cases}$$

where $q = \beta/\alpha$. These are the q-forms of the results given by (4.5) and (4.6). Thus the proof of the LU-decomposition is completed.

The matrix \mathcal{F} is a new non-symmetric variant of the Filbert matrix.

Similarly, we can obtain a non-symmetric variant of the Lilbert matrix. So we define the matrix \mathcal{T} with

$$\mathcal{T}_{kj} = \frac{V_{\lambda k - \mu j + r}}{V_{\lambda k + \mu j + r}} = q^{\mu j} (-1)^{\mu j} \frac{1 + q^{\lambda k - \mu j + r}}{1 + q^{\lambda k + \mu j + r}}$$

for positive integers λ, μ and integer r and $q = \beta/\alpha$, as the Lucas analogue of the matrix M, where $\{V_n\}$ is generalized Lucas sequence.

If we choose $x = -q^r$ in our main results and apply Proposition 4.2 to our main results, after converting the results to the generalized Fibonacci or Lucas numbers, we have the following corollary for the matrix \mathcal{T} .

Corollary 4.12. For $k, j \geq 0$,

$$\mathcal{L}_{kj} = \frac{\left(\prod_{d=1}^{j} V_{\lambda j + \mu d + r}\right) \left(\prod_{d=1}^{j} U_{\lambda (k+1) - \lambda d}\right)}{\left(\prod_{d=1}^{j} V_{\lambda k + \mu d + r}\right) \left(\prod_{d=1}^{j} U_{\lambda d}\right)},$$

$$\mathcal{U}_{kj} = \begin{cases}
\frac{V_{-\mu j+r}}{V_{\mu j+r}} & \text{if } k = 0, \\
\Delta^{k} (-1)^{\mu j+(\lambda+\mu)\binom{k}{2}+r(k+1)} U_{2\mu j} \frac{\binom{k-1}{d-1} U_{\mu j-\mu d}}{\binom{k+1}{d-1} V_{\mu j+\lambda(d-1)+r}} \binom{k}{d-1} V_{\lambda k+\mu d+r}} & \text{if } k > 0,
\end{cases}$$

where Δ defined as in equation (2.3).

$$\mathcal{L}_{kj}^{-1} = (-1)^{k+j+\lambda\binom{k-j}{2}} \frac{\left(\prod_{d=1}^{k-1} V_{\lambda j+\mu d+r}\right) \left(\prod_{d=1}^{j} U_{\lambda(k+1)-\lambda d}\right)}{\left(\prod_{d=1}^{k-1} V_{\lambda k+\mu d+r}\right) \left(\prod_{d=1}^{j} U_{\lambda d}\right)},$$

and $\mathcal{U}_{kj}^{-1} =$

$$\begin{cases}
1 & if k = j = 0, \\
\frac{(-1)^{k+\lambda\binom{j}{2}+rj+\mu kj+\mu\binom{k}{2}}}{\Delta^{j}U_{2\mu k}} \frac{\left(\prod_{d=1}^{j} V_{\mu k+\lambda(d-1)+r}\right) \left(\prod_{d=1}^{j} V_{\lambda j+\mu d+r}\right)}{\left(\prod_{d=1}^{j} U_{\lambda d}\right) \left(\prod_{d=1}^{j-k} U_{\mu d}\right) \left(\prod_{d=1}^{k-1} U_{\mu d}\right)} & if j \geq k \geq 1,
\end{cases}$$

and for $j \geq 1$

$$\mathcal{U}_{0j}^{-1} = \frac{(-1)^{rj+\lambda\binom{j}{2}+1}}{\Delta^{j}} \frac{\left(\prod_{d=1}^{j} V_{\lambda j+\mu d+r}\right)}{\left(\prod_{d=1}^{j} U_{\lambda d}\right)}$$

$$\times \sum_{t=1}^{j} (-1)^{t+\mu t j+\mu\binom{t}{2}} \frac{V_{-\mu t+r}}{U_{2\mu t} V_{\mu t+r}} \frac{\left(\prod_{d=1}^{j} V_{\mu t+\lambda (d-1)+r}\right)}{\left(\prod_{d=1}^{j-t} U_{\mu d}\right) \left(\prod_{d=1}^{t-1} U_{\mu d}\right)}$$

and 0 otherwise.

For the inverse matrix, we have for $1 \le k < N$ and $0 \le j < N$,

$$\mathcal{T}_{kj}^{-1} = \frac{(-1)^{k+j+\lambda\binom{j}{2}+\mu\binom{k+1}{2}-N(\mu k+\lambda j)+(r+1)(N-1)}}{\Delta^{N-1}U_{2\mu k}V_{\mu k+\lambda j+r}} \times \frac{\left(\prod_{d=1}^{N}V_{\mu k+\lambda(d-1)+r}\right)\left(\prod_{d=1}^{N}V_{\lambda j+\mu d+r}\right)}{\left(\prod_{d=1}^{N-k-1}U_{\mu d}\right)\left(\prod_{d=1}^{N-j-1}U_{\lambda d}\right)\left(\prod_{d=1}^{j}U_{\lambda d}\right)\left(\prod_{d=1}^{k-1}U_{\mu d}\right)}$$

and for $0 \le j < N$,

$$\mathcal{T}_{0j}^{-1} = [j=0] - \sum_{t=1}^{N-1} \frac{V_{-\mu t+r}}{V_{\mu t+r}} \mathcal{T}_{tj}^{-1}.$$

More specially, by choosing $x = q^r$ such that r is a positive integer and performing the limit $q \to 1$ in our main results, we obtain the related results for the matrix $\mathcal{H} = [\mathcal{H}_{kj}]_{k,j \geq 0}$ as a non-symmetric variant of the Hilbert matrix with entries

$$\mathcal{H}_{kj} = \frac{\lambda k - \mu j + r}{\lambda k + \mu j + r}.$$

So we list the results for the matrix \mathcal{H} below.

Corollary 4.13. For $k, j \geq 0$,

$$\mathcal{L}_{kj} = \frac{\left(\prod_{d=1}^{j} [\lambda j + \mu d + r]\right) \left(\prod_{d=1}^{j} \lambda [k - d + 1]\right)}{\left(\prod_{d=1}^{j} [\lambda k + \mu d + r]\right) \left(\prod_{d=1}^{j} \lambda d\right)},$$

$$if k = 0,$$

$$\mathcal{U}_{kj} = \begin{cases} \frac{r - \mu j}{r + \mu j} & \text{if } k = 0, \\ 2 \times \frac{\left(\prod_{d=1}^{k} \mu [j - d + 1]\right) \left(\prod_{d=1}^{k} \lambda d\right)}{\left(\prod_{d=1}^{k+1} [\mu j + \lambda (d - 1) + r]\right) \left(\prod_{d=1}^{k-1} [\lambda k + \mu d + r]\right)} & \text{if } k > 0, \end{cases}$$

$$\mathcal{L}_{kj}^{-1} = (-1)^{k+j} \frac{\left(\prod_{d=1}^{k-1} [\lambda j + \mu d + r]\right) \left(\prod_{d=1}^{j} \lambda [k - d + 1]\right)}{\left(\prod_{d=1}^{k-1} [\lambda k + \mu d + r]\right) \left(\prod_{d=1}^{j} \lambda d\right)}$$

and

$$\mathcal{U}_{kj}^{-1} = \begin{cases} 1 & \text{if } k = j = 0, \\ \left(\prod_{d=1}^{j} [\lambda j + \mu d + r]\right) \\ \left(\prod_{d=1}^{j} \lambda d\right) \\ \times \sum_{t=1}^{j} \frac{(-1)^{t} (r - \mu t)}{(r + \mu t)(2\mu t)} \frac{\left(\prod_{d=1}^{j} [\mu t + \lambda (d - 1) + r]\right)}{\left(\prod_{d=1}^{j-t} \mu d\right) \left(\prod_{d=1}^{t-1} \mu d\right)} & \text{if } j \geq 1 \text{ and } k = 0, \\ \frac{(-1)^{k+j}}{(2\mu k)} \frac{\left(\prod_{d=1}^{j} [\mu k + \lambda (d - 1) + r]\right) \left(\prod_{d=1}^{j} [\lambda j + \mu d + r]\right)}{\left(\prod_{d=1}^{j} \lambda d\right) \left(\prod_{d=1}^{j-k} \mu d\right) \left(\prod_{d=1}^{k-1} \mu d\right)} & \text{if } j \geq k \geq 1, \\ 0 & \text{otherwise.} \end{cases}$$

Finally, for $1 \le k < N$ and $0 \le j < N$,

$$\mathcal{H}_{kj}^{-1} = \frac{(-1)^{k+j}}{(\mu k + \lambda j + r)(2\mu k)} \frac{\left(\prod_{d=1}^{N-1} [\lambda j + \mu d + r]\right) \left(\prod_{d=1}^{N} [\mu k + \lambda (d-1) + r]\right)}{\left(\prod_{d=1}^{N-k-1} \mu d\right) \left(\prod_{d=1}^{N-j-1} \lambda d\right) \left(\prod_{d=1}^{j} \lambda d\right) \left(\prod_{d=1}^{k-1} \mu d\right)}$$

and for $0 \le j < N$,

$$\mathcal{H}_{0j}^{-1} = [j = 0] - \sum_{t=1}^{N-1} \frac{r - \mu t}{r + \mu t} \mathcal{H}_{tj}^{-1}.$$

4.6 A Nonlinear Generalization of the Filbert Matrix

As mentioned in Section 3.2.4, so far authors studied many generalizations and variants of the Filbert matrix. In previous section, we have also obtained a new non-symmetric variant of it. But no one has considered a generalization where the indexes of the recursive sequence is in nonlinear form yet. In this section, we will introduce a new nonlinear generalization of the Filbert matrix. We define the matrix M as a nonlinear generalization of the Filbert matrix with indexes in geometric progression for some parameters by

$$M_{kj} = \frac{1}{U_{\lambda(k+r)^n + \mu(j+s)^m + c}},$$

where U_n is nth generalized Fibonacci number. Moreover, as Lilbert analogue, for some parameters, we define the matrix T with entries

$$T_{kj} = \frac{1}{V_{\lambda(k+r)^n + \mu(j+s)^m + c}},$$

where V_n is the nth generalized Lucas number. Our study [6] is accepted for publication.

Note that when n = m = 1, our results will cover all Filbert-like matrices except the matrices whose entries include the products of the generalized Fibonacci or Lucas numbers.

For the matrices M and T, we derive explicit formulæ for the inverse matrix, LUdecomposition and inverse matrices L^{-1} , U^{-1} as well as we present the Cholesky decomposition when the matrices are symmetric. Later, we will give the q-forms of these
results. Actually, although the results related to the q-forms are more general, i.e.
when $q = \beta/\alpha$ gives the matrices M and T, we prefer to give Fibonacci and Lucas
forms first. Because they seem nicer and manipulating them is easier.

Note that any mechanic summation methods or q-Zeilberger algorithm will not work here due to the non-hypergeometric terms. This is another reason of presenting Fibonacci and Lucas form first. In order to prove our results we will use some traditional methods as in previous section.

Throughout this section, we assume that λ, μ, n and m are positive integers, r, s and c are any integers such that $\lambda(k+r)^n + \mu(j+s)^m + c > 0$ for all positive integers k and j.

4.6.1 A Nonlinear Filbert Matrix

For the matrix M, we will give explicit formulæ for its inverse, LU-decomposition, the inverse matrices L^{-1} and U^{-1} as well as we present its Cholesky decomposition when the matrix is symmetric, that is, the case r = s, n = m and $\lambda = \mu$.

We obtain the LU-decomposition:

Theorem 4.25. For k, j > 1,

$$L_{kj} = \frac{\left(\prod_{t=1}^{j} U_{\lambda(j+r)^n + \mu(t+s)^m + c}\right) \left(\prod_{t=1}^{j-1} U_{\lambda(k+r)^n - \lambda(t+r)^n}\right)}{\left(\prod_{t=1}^{j} U_{\lambda(k+r)^n + \mu(t+s)^m + c}\right) \left(\prod_{t=1}^{j-1} U_{\lambda(j+r)^n - \lambda(t+r)^n}\right)}$$

and

$$U_{kj} = (-1)^{(\lambda+\mu)\binom{k}{2} + (\lambda r + \mu s + c)(k+1)} \frac{\left(\prod_{t=1}^{k-1} U_{\lambda(k+r)^n - \lambda(t+r)^n}\right) \left(\prod_{t=1}^{k-1} U_{\mu(j+s)^m - \mu(t+s)^m}\right)}{\left(\prod_{t=1}^{k-1} U_{\lambda(k+r)^n + \mu(t+s)^m + c}\right) \left(\prod_{t=1}^{k} U_{\mu(j+s)^m + \lambda(t+r)^n + c}\right)}.$$

Similar to the previous sections, the determinant of the matrix M can be derived, as well. We also determine the inverse matrices L^{-1} and U^{-1} :

Theorem 4.26. For $k, j \geq 1$,

$$L_{kj}^{-1} = (-1)^{(\lambda+1)(k+j)+\lambda\binom{k-j+1}{2}} \frac{\left(\prod_{t=1}^{k-j-1} U_{\lambda(k+r)^n - \lambda(t+j+r)^n}\right)}{\left(\prod_{t=1}^{k-j-1} U_{\lambda(t+j+r)^n - \lambda(j+r)^n}\right)} \times \frac{\left(\prod_{t=1}^{k-1} U_{\lambda(j+r)^n + \mu(t+s)^m + c}\right) \left(\prod_{t=1}^{j-1} U_{\lambda(k+r)^n - \lambda(t+r)^n}\right)}{\left(\prod_{t=1}^{k-1} U_{\lambda(k+r)^n + \mu(t+s)^m + c}\right) \left(\prod_{t=1}^{j-1} U_{\lambda(j+r)^n - \lambda(t+r)^n}\right)}$$

and

$$\begin{split} U_{kj}^{-1} &= (-1)^{\lambda \binom{j+1}{2} + \mu \binom{k+1}{2} + k(\mu j + 1) + j(\lambda + 1) + (\lambda r + \mu s + c)(j + 1)} \\ &\times \frac{\left(\prod_{t=1}^{j-1} U_{\mu(k+s)^m + \lambda(t+r)^n + c}\right) \left(\prod_{t=1}^{j} U_{\lambda(j+r)^n + \mu(t+s)^m + c}\right)}{\left(\prod_{t=1}^{k-1} U_{\mu(k+s)^m - \mu(t+s)^m}\right) \left(\prod_{t=1}^{j-k} U_{\mu(j+s+1-t)^m - \mu(k+s)^m}\right) \left(\prod_{t=1}^{j-1} U_{\lambda(j+r)^n - \lambda(t+r)^n}\right)}. \end{split}$$

Now we give the explicit expression for the inverse matrix M_N^{-1} .

Theorem 4.27. For $1 \le k, j \le N$,

$$\begin{split} M_{kj}^{-1} &= (-1)^{k+j+\lambda \binom{j+1}{2}+\mu \binom{k+1}{2}+N(\lambda j+\mu k+c+\lambda r+\mu s)+c+\lambda r+\mu s} \\ &\times \frac{1}{U_{\lambda(j+r)^n+\mu(k+s)^m+c} \left(\prod_{t=1}^{k-1} U_{\mu(k+s)^m-\mu(t+s)^m}\right) \left(\prod_{t=1}^{j-1} U_{\lambda(j+r)^n-\lambda(t+r)^n}\right)} \\ &\times \frac{\left(\prod_{t=1}^{N} U_{\lambda(t+r)^n+\mu(k+s)^m+c}\right) \left(\prod_{t=1}^{N} U_{\mu(t+s)^m+\lambda(j+r)^n+c}\right)}{\left(\prod_{t=1}^{N-k} U_{\mu(N+s+1-t)^m-\mu(k+s)^m}\right) \left(\prod_{t=1}^{N-j} U_{\lambda(N+r+1-t)^n-\lambda(j+r)^n}\right)}. \end{split}$$

Finally, we provide the Cholesky decomposition of the matrix M when it is symmetric, that is r = s, n = m and $\lambda = \mu$.

Theorem 4.28. For $k, j \geq 1$,

$$C_{kj} = \frac{\prod_{t=1}^{j-1} U_{\lambda(k+r)^n - \lambda(t+r)^n}}{\prod_{t=1}^{j} U_{\lambda(k+r)^n + \lambda(t+r)^n + c}} \sqrt{(-1)^{c(j+1)} U_{2\lambda(j+r)^n + c}}.$$

Note that when n=m=1, $\lambda=\mu=1$, r=s=0 and c=-1 with p=1, the matrix M is reduced to Filbert matrix and so we obtain the results of [64]. Similarly, when n=m=1 and r=s=0 our results cover the results of [70]. For the cases n>1 or m>1, our results are all new.

Proofs

Define the following four sums:

$$S_{5}(K) = \sum_{d=K}^{\min(k,j)} (-1)^{(\lambda+\mu)\binom{d}{2} + (\lambda r + \mu s + c)(d+1)} U_{\lambda(d+r)^{n} + \mu(d+s)^{m} + c}$$

$$\times \frac{\left(\prod_{t=1}^{d-1} U_{\lambda(k+r)^{n} - \lambda(t+r)^{n}}\right) \left(\prod_{t=1}^{d-1} U_{\mu(j+s)^{m} - \mu(t+s)^{m}}\right)}{\left(\prod_{t=1}^{d} U_{\lambda(k+r)^{n} + \mu(t+s)^{m} + c}\right) \left(\prod_{t=1}^{d} U_{\mu(j+s)^{m} + \lambda(t+r)^{n} + c}\right)},$$

$$S_{6}(K) = \sum_{d=j}^{K} (-1)^{(\lambda+1)(d+j)+\lambda\binom{d-j+1}{2}} U_{\lambda(d+r)^{n}+\mu(d+s)^{m}+c} \frac{\left(\prod_{t=1}^{d-j-1} U_{\lambda(d+r)^{n}-\lambda(t+j+r)^{n}}\right)}{\left(\prod_{t=1}^{d-j-1} U_{\lambda(t+j+r)^{n}-\lambda(j+r)^{n}}\right)} \times \frac{\left(\prod_{t=1}^{d-1} U_{\lambda(k+r)^{n}-\lambda(t+r)^{n}}\right) \left(\prod_{t=1}^{d-1} U_{\lambda(j+r)^{n}+\mu(t+s)^{m}+c}\right) \left(\prod_{t=1}^{j-1} U_{\lambda(d+r)^{n}-\lambda(t+r)^{n}}\right)}{\left(\prod_{t=1}^{d-1} U_{\lambda(d+r)^{n}-\lambda(t+r)^{n}}\right) \left(\prod_{t=1}^{d} U_{\lambda(k+r)^{n}+\mu(t+s)^{m}+c}\right)},$$

$$S_7(K) = \sum_{d=\max(k,j)}^{K} (-1)^{k\mu d + \lambda dj + (\lambda r + \mu s + c)d} U_{\lambda(d+r)^n + \mu(d+s)^m + c} \frac{\left(\prod_{t=1}^{d-j-1} U_{\lambda(d+r)^n - \lambda(t+j+r)^n}\right)}{\left(\prod_{t=1}^{d-j-1} U_{\lambda(t+j+r)^n - \lambda(j+r)^n}\right)}$$

$$\times \frac{\left(\prod_{t=1}^{d-1} U_{\mu(k+s)^m + \lambda(t+r)^n + c}\right) \left(\prod_{t=1}^{d-1} U_{\lambda(j+r)^n + \mu(t+s)^m + c}\right) \left(\prod_{t=1}^{j-1} U_{\lambda(d+r)^n - \lambda(t+r)^n}\right)}{\left(\prod_{t=1}^{d-k} U_{\mu(d+s+1-t)^m - \mu(k+s)^m}\right) \left(\prod_{t=1}^{d-1} U_{\lambda(d+r)^n - \lambda(t+r)^n}\right)}$$

and

$$S_8(K) = \sum_{d=k}^{K} (-1)^{k\mu d + \mu \binom{d}{2} + d} U_{\lambda(d+r)^n + \mu(d+s)^m + c}$$

$$\times \frac{\left(\prod_{t=1}^{d-1} U_{\mu(k+s)^m + \lambda(t+r)^n + c}\right) \left(\prod_{t=1}^{d-1} U_{\mu(j+s)^m - \mu(t+s)^m}\right)}{\left(\prod_{t=1}^{d-k} U_{\mu(d+s+1-t)^m - \mu(k+s)^m}\right) \left(\prod_{t=1}^{d} U_{\mu(j+s)^m + \lambda(t+r)^n + c}\right)}.$$

We need the following lemmas for later use.

Lemma 4.8.

$$S_5(K) = \frac{(-1)^{(\lambda+\mu)\binom{K}{2} + (\lambda r + \mu s + c)(K+1)} \left(\prod_{t=1}^{K-1} U_{\lambda(k+r)^n - \lambda(t+r)^n}\right) \left(\prod_{t=1}^{K-1} U_{\mu(j+s)^m - \mu(t+s)^m}\right)}{U_{\lambda(k+r)^n + \mu(j+s)^m + c} \left(\prod_{t=1}^{K-1} U_{\lambda(k+r)^n + \mu(t+s)^m + c}\right) \left(\prod_{t=1}^{K-1} U_{\mu(j+s)^m + \lambda(t+r)^n + c}\right)}.$$

Proof. We will use the backward induction method. For brevity, denote the summand term by s_d . First, assume that $k \geq j$ so when K = j the claim is obvious. Similarly for the case j > k, the claim is clear. The backward induction step amounts to show that

$$S_5(K-1) = S_5(K) + s_{K-1}.$$

By the definition of $S_5(K)$ and S_{K-1} , consider the RHS of the above equality

$$\frac{(-1)^{(\lambda+\mu)\binom{K}{2}+(\lambda r+\mu s+c)(K+1)}}{U_{\lambda(k+r)^n+\mu(j+s)^m+c}} \frac{\left(\prod_{t=1}^{K-1} U_{\lambda(k+r)^n-\lambda(t+r)^n}\right) \left(\prod_{t=1}^{K-1} U_{\mu(j+s)^m-\mu(t+s)^m}\right)}{\left(\prod_{t=1}^{K-1} U_{\lambda(k+r)^n+\mu(t+s)^m+c}\right) \left(\prod_{t=1}^{K-1} U_{\mu(j+s)^m+\lambda(t+r)^n+c}\right)} \\ + (-1)^{(\lambda+\mu)\binom{K-1}{2}+(\lambda r+\mu s+c)K} U_{\lambda(K-1+r)^n+\mu(K-1+s)^m+c} \\ \times \frac{\left(\prod_{t=1}^{K-2} U_{\lambda(k+r)^n-\lambda(t+r)^n}\right) \left(\prod_{t=1}^{K-2} U_{\mu(j+s)^m-\mu(t+s)^m}\right)}{\left(\prod_{t=1}^{K-1} U_{\lambda(k+r)^n+\mu(t+s)^m+c}\right) \left(\prod_{t=1}^{K-1} U_{\mu(j+s)^m+\lambda(t+r)^n+c}\right)} \\ \times \frac{\left(\prod_{t=1}^{K-1} U_{\lambda(k+r)^n+\mu(t+s)^m+c}\right) \left(\prod_{t=1}^{K-1} U_{\mu(j+s)^m+\lambda(t+r)^n+c}\right)}{\left(\prod_{t=1}^{K-1} U_{\lambda(k+r)^n+\mu(t+s)^m+c}\right) \left(\prod_{t=1}^{K-1} U_{\mu(j+s)^m+\lambda(t+r)^n+c}\right)}$$

$$=\frac{(-1)^{(\lambda+\mu)\binom{K-1}{2}+(\lambda r+\mu s+c)K}}{U_{\lambda(k+r)^n+\mu(j+s)^m+c}}\frac{\left(\prod_{t=1}^{K-2}U_{\lambda(k+r)^n-\lambda(t+r)^n}\right)\left(\prod_{t=1}^{K-2}U_{\mu(j+s)^m-\mu(t+s)^m}\right)}{\left(\prod_{t=1}^{K-1}U_{\lambda(k+r)^n+\mu(t+s)^m+c}\right)\left(\prod_{t=1}^{K-1}U_{\mu(j+s)^m+\lambda(t+r)^n+c}\right)}$$

$$\times\left[(-1)(\lambda+\mu)(K-1)+(\lambda r+\mu s+c)U_{\lambda(k+r)^n-\lambda(K-1+r)^n}U_{\mu(j+s)^m-\mu(K-1+s)^m}\right.$$

$$\left.+U_{\lambda(K-1+r)^n+\mu(K-1+s)^m+c}U_{\lambda(k+r)^n+\mu(j+s)^m+c}\right].$$

By using the fact $U_n = (-1)^{n-1}U_{-n}$, the last expression in the bracket is rewritten as

$$(-1)^{\lambda(k+r)+\mu(j+s)+c}U_{\lambda(K-1+r)^n-\lambda(k+r)^n}U_{\mu(K-1+s)^m-\mu(j+s)^m} + U_{\lambda(K-1+r)^n+\mu(K-1+s)^m+c}U_{\lambda(k+r)^n+\mu(j+s)^m+c}$$

$$(4.7)$$

and then by using the identity (2.4) for $m \to \mu(j+s)^m + \lambda(K-1+r)^n + c$, $n \to \lambda(k+r)^n + \mu(K-1+s)^m + c$ and $k \to \mu(K-1+s)^m - \mu(j+s)^m$, the expression in (4.7) equals

$$U_{\lambda(k+r)^n+\mu(K-1+s)^m+c}U_{\mu(j+s)^m+\lambda(K-1+r)^n+c}.$$

Finally, we write

$$S_{5}(K) + s_{K-1} = \frac{(-1)^{(\lambda+\mu)\binom{K-1}{2} + (\lambda r + \mu s + c)K}}{U_{\lambda(k+r)^{n} + \mu(j+s)^{m} + c}} \times \frac{\binom{K-2}{\prod_{t=1}^{K-2} U_{\lambda(k+r)^{n} - \lambda(t+r)^{n}}}{\binom{K-2}{\prod_{t=1}^{K-2} U_{\lambda(k+r)^{n} + \mu(t+s)^{m} + c}}} \binom{K-2}{\prod_{t=1}^{K-2} U_{\mu(j+s)^{m} + \lambda(t+r)^{n} + c}},$$

which completes the proof.

Lemma 4.9. *For* k > j,

$$S_{6}(K) = \frac{(-1)^{\lambda \binom{K-j}{2} + (\lambda+1)(K+j)}}{U_{\lambda(k+r)^{n} - \lambda(j+r)^{n}}} \frac{\left(\prod_{t=1}^{K} U_{\lambda(k+r)^{n} - \lambda(t+r)^{n}}\right) \left(\prod_{t=1}^{K} U_{\lambda(j+r)^{n} + \mu(t+s)^{m} + c}\right)}{\left(\prod_{t=1}^{K} U_{\lambda(k+r)^{n} + \mu(t+s)^{m} + c}\right) \left(\prod_{t=1}^{K} U_{\lambda(t+j+r)^{n} - \lambda(j+r)^{n}}\right)}.$$

Proof. Denote the summand term by s_d . By using induction, the case K = j is obvious. So the induction step amounts to show that

$$S_6(K+1) = S_6(K) + s_{K+1}$$
.

So after some simplifications, $S_6(K) + s_{K+1}$ equals

$$\frac{(-1)^{\lambda\binom{K+1-j}{2}+(K+1+j)}}{U_{\lambda(k+r)^n-\lambda(j+r)^n}} \frac{\left(\prod_{t=1}^K U_{\lambda(k+r)^n-\lambda(t+r)^n}\right) \left(\prod_{t=1}^K U_{\lambda(j+r)^n+\mu(t+s)^m+c}\right)}{\left(\prod_{t=1}^{K+1} U_{\lambda(k+r)^n+\mu(t+s)^m+c}\right) \left(\prod_{t=1}^{K-j+1} U_{\lambda(t+j+r)^n-\lambda(j+r)^n}\right)} \\ \times \left[U_{\lambda(k+r)^n-\lambda(j+r)^n} U_{\lambda(K+1+r)^n+\mu(K+1+s)^m+c} - U_{\lambda(k+r)^n+\mu(K+1+s)^m+c} U_{\lambda(K+1+r)^n-\lambda(j+r)^n}\right].$$

Since again by (2.4), the last expression in the bracket equals

$$U_{\lambda(k+r)^n-\lambda(K+1+r)^n}U_{\lambda(j+r)^n+\mu(K+1+s)^m+c}$$

the claim follows. \Box

Lemma 4.10.

$$S_7(K) = \frac{(-1)^{K(\lambda j + \mu k + c + \lambda r + \mu s)}}{U_{\lambda(j+r)^n + \mu(k+s)^m + c}} \frac{\left(\prod_{t=1}^K U_{\mu(k+s)^m + \lambda(t+r)^n + c}\right) \left(\prod_{t=1}^K U_{\lambda(j+r)^n + \mu(t+s)^m + c}\right)}{\left(\prod_{t=1}^{K-k} U_{\mu(t+k+s)^m - \mu(k+s)^m}\right) \left(\prod_{t=1}^K U_{\lambda(t+j+r)^n - \lambda(j+r)^n}\right)}.$$

Proof. If $j \geq k$, the case K = j easily follows. If k > j, then

$$S_{7}(k) = \frac{(-1)^{k(\mu k + \lambda j + \lambda r + \mu s + c)}}{U_{\lambda(k+r)^{n} - \lambda(j+r)^{n}}} \frac{\left(\prod_{t=1}^{k} U_{\mu(k+s)^{m} + \lambda(t+r)^{n} + c}\right) \left(\prod_{t=1}^{k-1} U_{\lambda(j+r)^{n} + \mu(t+s)^{m} + c}\right)}{\left(\prod_{t=1}^{k-j-1} U_{\lambda(t+j+r)^{n} - \lambda(j+r)^{n}}\right)}$$

$$= \frac{(-1)^{k(\mu k + \lambda j + \lambda r + \mu s + c)}}{U_{\lambda(j+r)^{n} + \mu(k+s)^{m} + c}} \frac{\left(\prod_{t=1}^{k} U_{\mu(k+s)^{m} + \lambda(t+r)^{n} + c}\right) \left(\prod_{t=1}^{k} U_{\lambda(j+r)^{n} + \mu(t+s)^{m} + c}\right)}{\left(\prod_{t=1}^{k-j} U_{\lambda(t+j+r)^{n} - \lambda(j+r)^{n}}\right)}.$$

So the first step of induction is complete. For the next step, we have

$$S_{7}(K+1) = \frac{(-1)^{K(\lambda j + \mu k + c + \lambda r + \mu s)}}{U_{\lambda(j+r)^{n} + \mu(k+s)^{m} + c}} \frac{\left(\prod_{t=1}^{K} U_{\mu(k+s)^{m} + \lambda(t+r)^{n} + c}\right) \left(\prod_{t=1}^{K} U_{\lambda(j+r)^{n} + \mu(t+s)^{m} + c}\right)}{\left(\prod_{t=1}^{K-k} U_{\mu(t+k+s)^{m} - \mu(k+s)^{m}}\right) \left(\prod_{t=1}^{K-j} U_{\lambda(t+j+r)^{n} - \lambda(j+r)^{n}}\right)} + (-1)^{(\mu k + \lambda j + \lambda r + \mu s + c)(K+1)} U_{\lambda(K+1+r)^{n} + \mu(K+1+s)^{m} + c}$$

$$\times \frac{\left(\prod_{t=1}^{K} U_{\mu(k+s)^{m}+\lambda(t+r)^{n}+c}\right) \left(\prod_{t=1}^{K} U_{\lambda(j+r)^{n}+\mu(t+s)^{m}+c}\right)}{\left(\prod_{t=1}^{K+1-k} U_{\mu(t+k+s)^{m}-\mu(k+s)^{m}}\right) \left(\prod_{t=1}^{K+1-j} U_{\lambda(t+j+r)^{n}-\lambda(j+r)^{n}}\right)},$$

which, after some simplifications, equals

$$\frac{(-1)^{(\mu k + \lambda j + \lambda r + \mu s + c)(K+1)}}{U_{\lambda(j+r)^n + \mu(k+s)^m + c}} \frac{\left(\prod_{t=1}^K U_{\mu(k+s)^m + \lambda(t+r)^n + c}\right) \left(\prod_{t=1}^K U_{\lambda(j+r)^n + \mu(t+s)^m + c}\right)}{\left(\prod_{t=1}^K U_{\mu(t+k+s)^m - \mu(k+s)^m}\right) \left(\prod_{t=1}^K U_{\lambda(j+r)^n + \mu(t+s)^m - \lambda(j+r)^n}\right)} \times \left[(-1)^{\lambda(j+r) + \mu(k+s) + c} U_{\mu(K+1+s)^m - \mu(k+s)^m} U_{\lambda(K+1+r)^n - \lambda(j+r)^n} + U_{\lambda(K+1+r)^n + \mu(K+1+s)^m + c} U_{\lambda(j+r)^n + \mu(k+s)^m + c}\right].$$

By the identity (2.4) with appropriate parameters, the last expression in the bracket equals

$$U_{\mu(k+s)^m+\lambda(K+1+r)^n+c}U_{\lambda(j+r)^n+\mu(K+1+s)^m+c},$$

so the proof follows by induction.

Lemma 4.11. *For* j > k,

$$S_8(K) = \frac{(-1)^{K\mu k + \mu\binom{K}{2} + K}}{U_{\lambda(k+r)^n + \mu(j+s)^m + c}} \frac{\left(\prod_{t=1}^K U_{\mu(k+s)^m + \lambda(t+r)^n + c}\right) \left(\prod_{t=1}^K U_{\mu(j+s)^m - \mu(t+s)^m}\right)}{\left(\prod_{t=1}^{K-k} U_{\mu(K+s+1-t)^m - \mu(k+s)^m}\right) \left(\prod_{t=1}^K U_{\mu(j+s)^m + \lambda(t+r)^n + c}\right)}.$$

Proof of Lemma 4.11 can be done by induction similar to the previous two lemmas.

Now we shall give the proofs of the main results for the matrix M.

For the matrices L and L^{-1} , it is obviously seen that $L_{kk}L_{kk}^{-1}=1$. For k>j, by Lemma 4.9

$$\sum_{j \le d \le k} L_{kd} L_{dj}^{-1} = S_6(k) = 0,$$

so we conclude

$$\sum_{j \le d \le k} L_{kd} L_{dj}^{-1} = [k = j],$$

as desired.

For the matrices U and U^{-1} , $U_{kk}^{-1}U_{kk} = 1$ is clear. In order to show the case j > k, by Lemma 4.11, we have

$$\sum_{k \le d \le j} U_{kd}^{-1} U_{dj} = \frac{(-1)^{\mu \binom{k+1}{2} + k}}{\left(\prod_{t=1}^{k-1} U_{\mu(k+s)^m - \mu(t+s)^m}\right)} S_8(j) = 0.$$

Thus $U^{-1} \cdot U = I$, as claimed.

For the LU-decomposition, we have to prove that

$$\sum_{1 \le d \le \min(k,j)} L_{kd} U_{dj} = M_{kj}.$$

By Lemma 4.8, we obtain

$$\sum_{1 \le d \le \min(k,j)} L_{kd} U_{dj} = S_5(1) = \frac{1}{U_{\lambda(k+r)^n + \mu(j+s)^m + c}},$$

which completes the proof.

For the inverse matrix M_N^{-1} , we use the fact $M_N^{-1} = U_N^{-1} \cdot L_N^{-1}$. Consider

$$\sum_{\max(k,j) \le d \le N} U_{kd}^{-1} L_{dj}^{-1} = \frac{(-1)^{\mu \binom{k+1}{2} + k + \lambda r + \mu s + c + j + \lambda \binom{j+1}{2}}}{\left(\prod_{t=1}^{k-1} U_{\mu(k+s)^m - \mu(t+s)^m}\right) \left(\prod_{t=1}^{j-1} U_{\lambda(j+r)^n - \lambda(t+r)^n}\right)} S_7(N)$$

$$= (M_N^{-1})_{kj}.$$

The Cholesky decomposition follows by Corollary 4.1. Thus all proofs are complete.

4.6.2 A Nonlinear Lilbert Matrix

Now we give the LU-decomposition of the matrix T, inverse matrices L^{-1} and U^{-1} , inverse matrix T^{-1} and its Cholesky decomposition when r = s, n = m and $\lambda = \mu$, respectively. We don't give the proofs of these results because they may be done very similar to the proofs of the previous subsection. One needs only very small and proper changes in lemmas given above.

Theorem 4.29. For k, j > 1,

$$L_{kj} = \frac{\left(\prod_{t=1}^{j} V_{\lambda(j+r)^n + \mu(t+s)^m + c}\right) \left(\prod_{t=1}^{j-1} U_{\lambda(k+r)^n - \lambda(t+r)^n}\right)}{\left(\prod_{t=1}^{j} V_{\lambda(k+r)^n + \mu(t+s)^m + c}\right) \left(\prod_{t=1}^{j-1} U_{\lambda(j+r)^n - \lambda(t+r)^n}\right)}$$

and

$$U_{kj} = (-1)^{(\lambda+\mu)\binom{k}{2} + (\lambda r + \mu s + c + 1)(k+1)} \Delta^{k-1} \times \frac{\left(\prod_{t=1}^{k-1} U_{\lambda(k+r)^n - \lambda(t+r)^n}\right) \left(\prod_{t=1}^{k-1} U_{\mu(j+s)^m - \mu(t+s)^m}\right)}{\left(\prod_{t=1}^{k-1} V_{\lambda(k+r)^n + \mu(t+s)^m + c}\right) \left(\prod_{t=1}^{k} V_{\mu(j+s)^m + \lambda(t+r)^n + c}\right)},$$

where Δ is defined as before.

Theorem 4.30. For $k, j \ge 1$,

$$L_{kj}^{-1} = (-1)^{(\lambda+1)(k+j)+\lambda\binom{k-j+1}{2}} \frac{\left(\prod_{t=1}^{k-j-1} U_{\lambda(k+r)^n - \lambda(t+j+r)^n}\right)}{\left(\prod_{t=1}^{k-j-1} U_{\lambda(t+j+r)^n - \lambda(j+r)^n}\right)} \times \frac{\left(\prod_{t=1}^{k-1} V_{\lambda(j+r)^n + \mu(t+s)^m + c}\right) \left(\prod_{t=1}^{j-1} U_{\lambda(k+r)^n - \lambda(t+r)^n}\right)}{\left(\prod_{t=1}^{k-1} V_{\lambda(k+r)^n + \mu(t+s)^m + c}\right) \left(\prod_{t=1}^{j-1} U_{\lambda(j+r)^n - \lambda(t+r)^n}\right)}$$

and

$$\begin{split} U_{kj}^{-1} &= (-1)^{\lambda \binom{j+1}{2} + \mu \binom{k+1}{2} + k(\mu j + 1) + j(\lambda + 1) + (\lambda r + \mu s + c + 1)(j + 1)} \Delta^{1-j} \\ &\times \frac{\left(\prod_{t=1}^{j-1} V_{\mu(k+s)^m + \lambda(t+r)^n + c}\right) \left(\prod_{t=1}^{j} V_{\lambda(j+r)^n + \mu(t+s)^m + c}\right)}{\left(\prod_{t=1}^{k-1} U_{\mu(k+s)^m - \mu(t+s)^m}\right) \left(\prod_{t=1}^{j-k} U_{\mu(j+s+1-t)^m - \mu(k+s)^m}\right) \left(\prod_{t=1}^{j-1} U_{\lambda(j+r)^n - \lambda(t+r)^n}\right)}. \end{split}$$

Theorem 4.31. For $1 \le k, j \le N$,

$$\begin{split} M_{kj}^{-1} &= \frac{1}{\Delta^{N-1} V_{\lambda(j+r)^n + \mu(k+s)^m + c}} \frac{(-1)^{k+j + \lambda \binom{j+1}{2} + \mu \binom{k+1}{2} + N(\lambda j + \mu k) + (N+1)(c + \lambda r + \mu s + 1)}}{\left(\prod_{t=1}^{k-1} U_{\mu(k+s)^m - \mu(t+s)^m}\right) \left(\prod_{t=1}^{j-1} U_{\lambda(j+r)^n - \lambda(t+r)^n}\right)} \\ &\times \frac{\left(\prod_{t=1}^{N} V_{\lambda(t+r)^n + \mu(k+s)^m + c}\right) \left(\prod_{t=1}^{N} V_{\mu(t+s)^m + \lambda(j+r)^n + c}\right)}{\left(\prod_{t=1}^{N-k} U_{\mu(N+s+1-t)^m - \mu(k+s)^m}\right) \left(\prod_{t=1}^{N-j} U_{\lambda(N+r+1-t)^n - \lambda(j+r)^n}\right)}. \end{split}$$

Theorem 4.32. *For* $k, j \ge 1$,

$$C_{kj} = \frac{\prod_{t=1}^{j-1} U_{\lambda(k+r)^n - \lambda(t+r)^n}}{\prod_{t=1}^{j} V_{\lambda(k+r)^n + \lambda(t+r)^n + c}} \sqrt{(-1)^{(c+1)(j+1)} \Delta^{j-1} V_{2\lambda(j+r)^n + c}}.$$

Here note that when n=m=1 and r=s=0 the above results are reduced to the results of [70]. Similarly, when n=m=1, $\lambda=\mu=1$, r=s=0 and c=-1 with p=1 the matrix M is reduced to the usual Lilbert matrix. For the cases n>1 or m>1, our results are all new.

4.6.3 q-Analogues

We present the q-analogues of the results of Sections 4.6.1 and 4.6.2. The results for the matrices M and T given previously come out as corollaries of the below results for the special choice of $q = \beta/\alpha$, so that the results, will be provided after a while, are more general. Nevertheless, we prefer to give first the results related to the matrices M and T because they look nicer. We don't provide the proofs of the results of this section. They could be similarly done by finding the q-forms of the lemmas given before.

We denote the q-analogues of the matrices M and T by \mathcal{M} and \mathcal{T} :

$$\mathcal{M}_{kj} = \mathbf{i}^{-[\lambda(k+r)^n + \mu(j+s)^m + c - 1]} q^{\frac{1}{2}[\lambda(k+r)^n + \mu(j+s)^m + c - 1]} \frac{1 - q}{1 - q^{\lambda(k+r)^n + \mu(j+s)^m + c}}$$

and

$$\mathcal{T}_{kj} = \mathbf{i}^{-[\lambda(k+r)^n + \mu(j+s)^m + c]} q^{\frac{1}{2}[\lambda(k+r)^n + \mu(j+s)^m + c]} \frac{1}{1 + q^{\lambda(k+r)^n + \mu(j+s)^m + c}},$$

respectively.

For the convenience, we will define a generalization of the q-Pochhammer symbol with two additional parameters in which one of them is in geometric progression as follows

$$(a;q)_n^{(r,m)} := \left(1 - aq^{(1+r)^m}\right) \left(1 - aq^{(2+r)^m}\right) \dots \left(1 - aq^{(n+r)^m}\right) = \prod_{t=1}^n \left(1 - aq^{(t+r)^m}\right)$$

with $(a; q^{(r,m)})_0 = 1$, where a is a real number, r is an integer and n, m are positive integers. As examples, we note that

$$(1;q)_n^{(0,2)} = (1-q)(1-q^4)\dots(1-q^{n^2}),$$

$$(a;q^{2})_{n}^{(1,2)} = (1 - aq^{8})(1 - aq^{18})\dots(1 - aq^{2(n+1)^{2}}),$$

$$(-q;q)_{n}^{(-1,3)} = (1+q)(1+q^{2})(1+q^{9})\dots(1+q^{(n-1)^{3}+1}),$$

$$(a;q^{\lambda})_{n}^{(0,1)} = (1 - aq^{\lambda})(1 - aq^{2\lambda})\dots(1 - aq^{n\lambda}) = (aq^{\lambda};q^{\lambda})_{n}.$$

So the relationship between the usual and general q-Pochhammer symbol is

$$(x;q)_n = (x;q)_n^{(-1,1)}.$$

As the q-analogues of the results related to the matrix M, we present the following theorem for the matrix \mathcal{M} .

Theorem 4.33. For the matrix \mathcal{M} and $k, j \geq 1$,

$$\mathcal{L}_{kj} = q^{\frac{1}{2}\lambda[(k+r)^n - (j+r)^n]} \mathbf{i}^{\lambda[(j+r)^n - (k+r)^n]} \frac{\left(q^{\lambda(j+r)^n + c}; q^{\mu}\right)_j^{(s,m)} \left(q^{\lambda(k+r)^n}; q^{-\lambda}\right)_{j-1}^{(r,n)}}{\left(q^{\lambda(k+r)^n + c}; q^{\mu}\right)_j^{(s,m)} \left(q^{\lambda(j+r)^n}; q^{-\lambda}\right)_{j-1}^{(r,n)}},$$

$$\mathcal{U}_{kj} = q^{-\frac{1}{2}[\lambda(k+r)^n + \mu(j+s)^m + c+1] + k[\lambda(k+r)^n + \mu(j+s)^m + c]} \mathbf{i}^{[-\lambda(k+r)^n - \mu(j+s)^m - c+1]} \times \frac{(1-q) \left(q^{-\lambda(k+r)^n}; q^{\lambda}\right)_{k-1}^{(r,n)} \left(q^{-\mu(j+s)^m}; q^{\mu}\right)_{k-1}^{(s,m)}}{\left(q^{\lambda(k+r)^n + c}; q^{\mu}\right)_{k-1}^{(s,m)} \left(q^{\mu(j+s)^m + c}; q^{\lambda}\right)_k^{(r,n)}},$$

$$\mathcal{L}_{kj}^{-1} = (-1)^{\lambda(k+j)+1} \mathbf{i}^{\lambda[(k+r)^n - (j+r)^n]} q^{\lambda(j-k)[(j+r)^n - (k+r)^n] + \frac{1}{2}\lambda[(j+r)^n - (k+r)^n]} \times \frac{\left(q^{-\lambda(k+r)^n}; q^{\lambda}\right)_{k-j-1}^{(j+r,n)} \left(q^{\lambda(j+r)^n + c}; q^{\mu}\right)_{k-1}^{(s,m)} \left(q^{\lambda(k+r)^n}; q^{-\lambda}\right)_{j-1}^{(r,n)}}{\left(q^{-\lambda(j+r)^n}; q^{\lambda}\right)_{k-j-1}^{(j+r,n)} \left(q^{\lambda(k+r)^n + c}; q^{\mu}\right)_{k-1}^{(s,m)} \left(q^{\lambda(j+r)^n}; q^{-\lambda}\right)_{j-1}^{(r,n)}},$$

$$\mathcal{U}_{kj}^{-1} = (-1)^{\mu(k+s)+\lambda(j+r)+c} \mathbf{i}^{[-\mu(k+s)^m-\lambda(j+r)^n-c-1]} q^{\left(\frac{1}{2}-j\right)[\mu(k+s)^m+\lambda(j+r)^n+c]+\frac{1}{2}} \\ \times \frac{\left(q^{\mu(k+s)^m+c}; q^{\lambda}\right)_{j-1}^{(r,n)} \left(q^{\lambda(j+r)^n+c}; q^{\mu}\right)_{j}^{(s,m)}}{(1-q) \left(q^{-\mu(k+s)^m}; q^{\mu}\right)_{k-1}^{(s,m)} \left(q^{-\lambda(j+r)^n}; q^{\lambda}\right)_{j-1}^{(r,n)} \left(q^{-\mu(k+s)^m}; q^{\mu}\right)_{j-k}^{(k+s,m)}},$$

$$(\mathcal{M}_{N}^{-1})_{kj} = (-1)^{\mu k + \lambda j + \lambda r + \mu s + c} \mathbf{i}^{[-\mu(k+s)^{m} - \lambda(j+r)^{n} - c - 1]} q^{\left(\frac{1}{2} - N\right)[\mu(k+s)^{m} + \lambda(j+r)^{n} + c + 1] + \frac{1}{2}} \times \frac{\left(q^{\mu(k+s)^{m} + c}; q^{\lambda}\right)_{N}^{(r,n)} \left(q^{\lambda(j+r)^{n} + c}; q^{\mu}\right)_{N}^{(s,m)}}{(1 - q)(1 - q^{\lambda(j+r)^{n} + \mu(k+s)^{m} + c}) \left(q^{-\mu(k+s)^{m}}; q^{\mu}\right)_{k-1}^{(s,m)} \left(q^{-\lambda(j+r)^{n}}; q^{\lambda}\right)_{j-1}^{(r,n)}} \times \frac{1}{\left(q^{-\mu(k+s)^{m}}; q^{\mu}\right)_{N-k}^{(k+s,m)} \left(q^{-\lambda(j+r)^{n}}; q^{\lambda}\right)_{N-j}^{(j+r,n)}}$$

and when r = s, n = m and $\lambda = \mu$,

$$C_{kj} = (-1)^{\lambda \left[jr+k+\binom{j}{2}\right]+j+1} \mathbf{i}^{\left[\lambda(k+r)^n-\lambda(j+r)^n+1-cj\right]}$$

$$\times q^{\frac{1}{2}[\lambda(j+r)^{n}-\lambda(k+r)^{n}+cj-1]+\lambda(k+r)^{n}j} \frac{\left(q^{-\lambda(k+r)^{n}};q^{\lambda}\right)_{j-1}^{(r,n)}}{\left(q^{\lambda(k+r)^{n}+c};q^{\lambda}\right)_{j}^{(r,n)}} \times \sqrt{(1-q)q^{-\lambda(j+r)^{n}+\frac{1-c}{2}}(-1)^{\lambda(j+r)^{n}+c(j+1)+\frac{c-1}{2}}(1-q^{2\lambda(j+r)^{n}+c})}.$$

Similarly, as the q-analogue of the matrix T, we present the following theorem.

Theorem 4.34. For the matrix \mathcal{T} and $k, j \geq 1$,

$$\mathcal{L}_{kj} = q^{\frac{1}{2}\lambda[(k+r)^n - (j+r)^n]} \mathbf{i}^{\lambda[(j+r)^n - (k+r)^n]} \times \frac{\left(-q^{\lambda(j+r)^n + c}; q^{\mu}\right)_j^{(s,m)} \left(q^{\lambda(k+r)^n}; q^{-\lambda}\right)_{j-1}^{(r,n)}}{\left(-q^{\lambda(k+r)^n + c}; q^{\mu}\right)_j^{(s,m)} \left(q^{\lambda(j+r)^n}; q^{-\lambda}\right)_{j-1}^{(r,n)}},$$

$$\mathcal{U}_{kj} = q^{(k-1)[\lambda(k+r)^n + \mu(j+s)^m + c] + \frac{1}{2}[\lambda(k+r)^n + \mu(j+s)^m + c]} (-1)^{k-1} \mathbf{i}^{[-\lambda(k+r)^n - \mu(j+s)^m - c]} \times \frac{\left(q^{-\lambda(k+r)^n}; q^{\lambda}\right)_{k-1}^{(r,n)} \left(q^{-\mu(j+s)^m}; q^{\mu}\right)_{k-1}^{(s,m)}}{\left(-q^{\lambda(k+r)^n + c}; q^{\mu}\right)_{k-1}^{(s,m)} \left(-q^{\mu(j+s)^m + c}; q^{\lambda}\right)_k^{(r,n)}},$$

$$\mathcal{L}_{kj}^{-1} = (-1)^{\lambda(k-j)+1} \mathbf{i}^{\lambda[(k+r)^n - (j+r)^n]} q^{\lambda(j-k)[(j+r)^n - (k+r)^n] + \frac{1}{2}\lambda[(j+r)^n - (k+r)^n]} \times \frac{\left(q^{-\lambda(k+r)^n}; q^{\lambda}\right)_{k-j-1}^{(j+r,n)} \left(-q^{\lambda(j+r)^n + c}; q^{\mu}\right)_{k-1}^{(s,m)} \left(q^{\lambda(k+r)^n}; q^{-\lambda}\right)_{j-1}^{(r,n)}}{\left(q^{-\lambda(j+r)^n}; q^{\lambda}\right)_{k-j-1}^{(j+r,n)} \left(-q^{\lambda(k+r)^n + c}; q^{\mu}\right)_{k-1}^{(s,m)} \left(q^{\lambda(j+r)^n}; q^{-\lambda}\right)_{j-1}^{(r,n)}},$$

$$\mathcal{U}_{kj}^{-1} = q^{\left(\frac{1}{2} - j\right)[\mu(k+s)^m + \lambda(j+r)^n + c]} (-1)^{j-1} \mathbf{i}^{[-\mu(k+s)^m - \lambda(j+r)^n - c]}$$

$$\times \frac{\left(-q^{\mu(k+s)^m + c}; q^{\lambda}\right)_{j-1}^{(r,n)} \left(-q^{\lambda(j+r)^n + c}; q^{\mu}\right)_{j}^{(s,m)}}{\left(q^{-\mu(k+s)^m}; q^{\mu}\right)_{k-1}^{(s,m)} \left(q^{-\lambda(j+r)^n}; q^{\lambda}\right)_{j-1}^{(r,n)} \left(q^{-\mu(k+s)^m}; q^{\mu}\right)_{j-k}^{(k+s,m)}},$$

$$(\mathcal{T}_{N}^{-1})_{kj} = q^{\left(\frac{1}{2}-N\right)[\lambda(j+r)^{n}+\mu(k+s)^{m}+c]}(-1)^{\lambda j+\mu k+c+\lambda r+\mu s+N+1}\mathbf{i}^{\left[-\lambda(j+r)^{n}-\mu(k+s)^{m}-c\right]} \times \frac{\left(-q^{\mu(k+s)^{m}+c};q^{\lambda}\right)_{N}^{(r,n)}\left(-q^{\lambda(j+r)^{n}+c};q^{\mu}\right)_{N}^{(s,m)}}{\left(1+q^{\lambda(j+r)^{n}+\mu(k+s)^{m}+c}\right)\left(q^{-\mu(k+s)^{m}};q^{\mu}\right)_{N-k}^{(k+s,m)}\left(q^{-\lambda(j+r)^{n}};q^{\lambda}\right)_{N-j}^{(j+r,n)}} \times \frac{1}{\left(q^{-\mu(k+s)^{m}};q^{\mu}\right)_{k-1}^{(s,m)}\left(q^{-\lambda(j+r)^{n}};q^{\lambda}\right)_{j-1}^{(r,n)}}$$

and when r = s, n = m and $\lambda = \mu$,

$$C_{kj} = (-1)^{j+1+\lambda\binom{j}{2}+\lambda r(j-1)} \mathbf{i}^{[-\lambda(k+r)^n - \lambda(j+r)^n - cj]} q^{\frac{1}{2}[\lambda(k+r)^n + \lambda(j+r)^n + cj] + \lambda(j-1)(k+r)^n} \times \frac{\left(q^{-\lambda(k+r)^n}; q^{\lambda}\right)_{j-1}^{(r,n)}}{\left(-q^{\lambda(k+r)^n + c}; q^{\lambda}\right)_{j}^{(r,n)}} \sqrt{q^{-\lambda(j+r)^n - \frac{c}{2}}(-1)^{\lambda(j+r) + (c+1)(j+1) + \frac{c}{2}}(1 + q^{2\lambda(j+r)^n + c})}.$$

Note that the determinants of each matrices studied in this section can be evaluated by the multiplication of the diagonal elements of the related matrix U. We don't state them because they are overlong.

If you look at the q-forms of the matrices \mathcal{M} and \mathcal{T} , you may realize that there is some separable parts. So one may give some simple formulæ for these matrices by using Proposition 4.2. However we would prefer to present our results in this form because nonlinear generalizations of the Filbert and Lilbert matrices can directly obtain by choosing $q = \beta/\alpha$.

At the end of this section, we will give a nonlinear generalization of the Hilbert matrix as a corollary of Theorem 4.33 and Proposition 4.2. Namely, when $q \to 1$ the entries of the matrix \mathcal{M} takes the form

$$\lim_{q \to 1} \mathcal{M}_{kj} = \mathbf{i}^{[-\lambda(k+r)^n - \mu(j+s)^m - c + 1]} \frac{1}{\lambda(k+r)^n + \mu(j+s)^m + c}.$$

Since the factor in front of the ratio is separable with regards to the variables k and j, by Proposition 4.2 and Theorem 4.33, one could derive the results for the matrix \hat{M} with entries for $k, j \geq 1$,

$$\hat{M}_{kj} = \frac{1}{\lambda(k+r)^n + \mu(j+s)^m + c},$$

which is a nonlinear generalization of the Hilbert matrix.

We only state the LU-decomposition of the matrix \hat{M} by the following corollary. The others can be similarly derived.

Corollary 4.14. For $k, j \geq 1$,

$$\hat{L}_{kj} = \frac{\left(\prod_{t=1}^{j} \left[\lambda(j+r)^{n} + \mu(t+s)^{m} + c\right]\right) \left(\prod_{t=1}^{j-1} \left[\lambda(k+r)^{n} - \lambda(t+r)^{n}\right]\right)}{\left(\prod_{t=1}^{j} \left[\lambda(k+r)^{n} + \mu(t+s)^{m} + c\right]\right) \left(\prod_{t=1}^{j-1} \left[\lambda(j+r)^{n} - \lambda(t+r)^{n}\right]\right)}$$

and

$$\hat{U}_{kj} = \frac{\left(\prod_{t=1}^{k-1} \left[\lambda(t+r)^n - \lambda(k+r)^n\right]\right) \left(\prod_{t=1}^{k-1} \left[\mu(t+s)^m - \mu(j+s)^m\right]\right)}{\left(\prod_{t=1}^{k-1} \left[\lambda(k+r)^n + \mu(t+s)^m + c\right]\right) \left(\prod_{t=1}^{k} \left[\mu(j+s)^m + \lambda(t+r)^n + c\right]\right)}.$$

4.7 Hessenberg Determinants via Generating Functions

As mentioned in Section 3.2.5, the determinants of the Hessenberg matrices are investigated by several authors. In [7], author introduced a new method to compute the determinant of a special class of the Hessenberg matrices via generating functions. In this section, we extend his method to new three classes of the Hessenberg matrices. Another extension of it to the convolution-like matrices could be found in [93]. This method is based on to determine the relationships between determinants of the Hessenberg matrices whose entries are terms of some certain number sequences and generating functions of these sequences. As an application of our main results, we give an elegant method to compute the determinants of the Hessenberg matrices whose entries consist of the terms of the higher order linear recursive sequences, which based on to find an adjacency-factor matrix. Our results cover many previous results about determinants of the Hessenberg matrices. The obtained results are presented in [8].

In Section 4.7.1, we introduce these three classes of the Hessenberg matrices and show how to compute determinants of these matrices. Also we provide many useful examples to understand the method well.

We would like to remind that we use n instead of N for the order of the matrix throughout this section.

4.7.1 Main Results

Let $\{b_n\}_{n\geq 0}$ and $\{c_n\}_{n\geq 1}$ be any sequences. Denote their generating functions as $B(x) = \sum_{k\geq 0} b_k x^k$ and $C(x) = \sum_{k\geq 1} c_k x^k$, respectively. The capital calligraphic letter denotes the generating function of the determinant of the related matrix.

To generalize the result of [7], we define the Hessenberg matrix $H_n(r,s)$ of order n+1:

For arbitrary nonzero real numbers r and s,

$$H_{n}(r,s) := \begin{bmatrix} b_{0} & r & & & & & & & \\ b_{1} & c_{1} & s & & & & & \\ b_{2} & c_{2} & c_{1} & r & & & & \\ b_{3} & c_{3} & c_{2} & c_{1} & s & & & & \\ \vdots & \vdots & \vdots & \ddots & \ddots & \ddots & & \\ b_{n-1} & c_{n-1} & c_{n-2} & \cdots & \cdots & c_{1} & d_{n}(r,s) \\ b_{n} & c_{n} & c_{n-1} & \cdots & \cdots & c_{2} & c_{1} \end{bmatrix}, \tag{4.8}$$

where

$$d_n(r,s) = \begin{cases} r & \text{if } n \text{ is even,} \\ s & \text{if } n \text{ is odd.} \end{cases}$$

Briefly, we use H_n instead of $H_n(r,s)$ if there is no restrictions on r and s. The case r=s=1 is reduced to matrix considered in [7].

To compute the determinant of H_n via generating function method, we have the following result:

Theorem 4.35. If

$$H(x) = \frac{B(x)\left(C(-x) + \frac{r+s}{2}\right) - B(-x)\left(\frac{r-s}{2}\right)}{C(x)C(-x) + \left(\frac{r+s}{2}\right)\left(C(x) + C(-x)\right) + rs},$$

then

(i) for even n such that n = 2t,

$$\det H_n = r^{t+1} s^t h_n.$$

(ii) for odd n such that n = 2t + 1,

$$\det H_n = -r^{t+1}s^{t+1}h_n,$$

where H(x) is the generating function of $\{h_n\}_{n\geq 0}$.

Proof. We consider the infinite linear system of equations

$$\begin{bmatrix} r & & & & & & & \\ c_{1}x & sx & & & & & \\ c_{2}x^{2} & c_{1}x^{2} & rx^{2} & & & \\ c_{3}x^{3} & c_{2}x^{3} & c_{1}x^{3} & sx^{3} & & \\ c_{4}x^{4} & c_{3}x^{4} & c_{2}x^{4} & c_{1}x^{4} & rx^{4} & \\ \vdots & \vdots & \vdots & \vdots & \vdots & \ddots \end{bmatrix} \begin{bmatrix} h_{0} \\ h_{1} \\ h_{2} \\ h_{3} \\ h_{4} \\ \vdots \end{bmatrix} = \begin{bmatrix} b_{0} \\ b_{1}x \\ b_{2}x^{2} \\ b_{3}x^{3} \\ b_{4}x^{4} \\ \vdots \end{bmatrix}.$$
(4.9)

Here we write

$$rh_0 = b_0$$

$$c_1h_0x + sh_1x = b_1x$$

$$c_2h_0x^2 + c_1h_1x^2 + rh_2x^2 = b_2x^2$$

$$c_3h_0x^3 + c_2h_1x^3 + c_1h_2x^3 + sh_3x^3 = b_3x^3$$

$$\vdots = \vdots$$

By summing both side of the above equalities and (2.13), we obtain

$$H(x)C(x) + r\sum_{k>0} h_{2k}x^{2k} + s\sum_{k>0} h_{2k+1}x^{2k+1} = B(x).$$
(4.10)

By the relations (2.14), the above equation could be rewritten as

$$H(x)\left[C(x) + \frac{r+s}{2}\right] + H(-x)\left[\frac{r-s}{2}\right] = B(x).$$

Taking (-x) instead of x, we get

$$H(-x)\left[C(-x) + \frac{r+s}{2}\right] + H(x)\left[\frac{r-s}{2}\right] = B(-x).$$

Solving these two equations in terms of H(x), we get

$$H(x) = \frac{B(x)\left(C(-x) + \frac{r+s}{2}\right) - B(-x)\left(\frac{r-s}{2}\right)}{C(x)C(-x) + \left(\frac{r+s}{2}\right)\left(C(x) + C(-x)\right) + rs},$$

as desired.

Now we examine the relationship between the sequences $\{h_n\}_{n\geq 0}$ and $\{\det H_n\}_{n\geq 0}$. If we consider the system (4.9) for the only first n+1 equations and take x=1, the system (4.9) turns to

$$\begin{bmatrix} r & & & & & & & \\ c_1 & s & & & & & \\ c_2 & c_1 & r & & & & \\ c_3 & c_2 & c_1 & s & & & \\ \vdots & \vdots & \vdots & \vdots & \ddots & & \\ c_n & c_{n-1} & c_{n-2} & \cdots & \vdots & d_{n+1}(r,s) \end{bmatrix} \begin{bmatrix} h_0 \\ h_1 \\ h_2 \\ h_3 \\ \vdots \\ h_n \end{bmatrix} = \begin{bmatrix} b_0 \\ b_1 \\ b_2 \\ b_3 \\ \vdots \\ b_n \end{bmatrix},$$

where $d_n(r, s)$ is defined as before.

By Cramer's rule (see page 24 in [43]), we obtain $h_n = \frac{\det H_n}{r^{t+1}s^t}$ for even n such that n = 2t and $h_n = -\frac{\det H_n}{r^{t+1}s^{t+1}}$ for odd n such that n = 2t + 1, which completes the proof.

We want to note some important and useful special cases of Theorem 4.35 with the following corollaries:

Corollary 4.15. For the matrix H(1,1), we have that $h_n = (-1)^n \det H_n$ and the generating function of the sequence $\{\det H_n(1,1)\}_{n\geq 0}$ is

$$\mathcal{H}(x) = \frac{B(-x)}{1 + C(-x)}.$$

This result was firstly given in [7]. We refer to it for some examples.

Corollary 4.16. For the matrix H(-1,-1), we have that $h_n = -\det H_n$ and the generating function of the sequence $\{\det H_n(-1,-1)\}_{n\geq 0}$ is

$$\mathcal{H}(x) = \frac{B(x)}{1 - C(x)}. (4.11)$$

Let's give some examples.

Example 4.1. For $n \geq 0$, we have that

$$\begin{vmatrix} F_1 & -1 & & & & 0 \\ F_2 & 1 & -1 & & & & \\ F_3 & 1 & 1 & -1 & & & \\ F_4 & 0 & 1 & 1 & -1 & & \\ \vdots & \vdots & \vdots & \cdots & \ddots & \ddots & \\ F_n & 0 & 0 & \cdots & \cdots & 1 & -1 \\ F_{n+1} & 0 & 0 & \cdots & \cdots & 1 & 1 \end{vmatrix} = \sum_{k=0}^n F_{k+1} F_{n+1-k}.$$

Proof. If $b_n = F_{n+1}$ and $\{c_n\}_{n\geq 1} = \{1,1,0,\ldots\}$, then $B(x) = \frac{1}{1-x-x^2}$ and $C(x) = x+x^2$. So the generating function of $\{\det H_n(-1,-1)\}_{n\geq 0}$ by Corollary 4.16, is $\frac{1}{(1-x-x^2)^2}$, which is the generating function of $\{\sum_{k=0}^n F_{k+1}F_{n+1-k}\}_{n\geq 0}$, as well. Thus the proof is complete by Theorem 2.5.

Example 4.2. For n > 0, we have that

$$\begin{vmatrix} L_0 & -1 & & & & & 0 \\ L_1 & -F_1 & -1 & & & & \\ L_2 & -F_2 & -F_1 & -1 & & & \\ L_3 & -F_3 & -F_2 & -F_1 & -1 & & \\ \vdots & \vdots & \vdots & \ddots & \ddots & \ddots & \\ L_{n-1} & -F_{n-1} & -F_{n-2} & \cdots & \cdots & -F_1 & -1 \\ L_n & -F_n & -F_{n-1} & \cdots & \cdots & -F_2 & -F_1 \end{vmatrix} = \begin{cases} 2 & \text{if } n \text{ is even,} \\ -1 & \text{if } n \text{ is odd.} \end{cases}$$

Proof. Since $b_n = L_n$ and $\{c_n\}_{n \ge 1} = \{-F_n\}_{n \ge 1}$, $B(x) = \frac{2-x}{1-x-x^2}$ and $C(x) = \frac{-x}{1-x-x^2}$. By Corollary 4.16, the generating function of $\{\det H_n(-1,-1)\}_{n \ge 0}$ is

$$\mathcal{H}(x) = \frac{B(x)}{1 - C(x)} = \frac{2 - x}{1 - x^2},$$

which gives the periodic sequence $\{2, -1, 2, -1, \ldots\}_{n \geq 0}$.

Let $\{b_n\}$ be any sequence and $\{c_n\}_{n\geq 1} = \{1,0,0,\ldots\}$. Since $\frac{1}{1-x}B(x)$ is the generating function of the sum of the first *n*th term of the sequence $\{b_n\}$, by Corollary 4.16 and Theorem 2.5, we have

$$\det H_n(-1, -1) = \sum_{k=0}^n b_k.$$

For example,

$$\begin{vmatrix} 1 & -1 & & & & 0 \\ \frac{1}{2} & 1 & -1 & & & \\ \frac{1}{3} & 0 & 1 & -1 & & \\ \frac{1}{4} & 0 & 0 & 1 & -1 & & \\ \vdots & \vdots & \vdots & \cdots & \ddots & \ddots & \\ \frac{1}{n} & 0 & 0 & \cdots & \cdots & 1 & -1 \\ \frac{1}{n+1} & 0 & 0 & \cdots & \cdots & 0 & 1 \end{vmatrix} = \mathcal{H}_{n+1},$$

where \mathscr{H}_n stands for *n*th harmonic number, which is $\sum_{k=1}^n \frac{1}{k}$.

Since the permanental and determinantal relationships between the matrices $H_n(1,1)$ and $H_n(-1,-1)$ are

$$\det H_n(1,1) = \operatorname{per} H_n(-1,-1)$$
 and $\operatorname{per} H_n(1,1) = \det H_n(-1,-1)$,

one can easily derive some permanental relations for the Hessenberg matrices by the help of the above corollaries.

Corollary 4.17. If

$$H(x) = \frac{C(-x)B(x) - B(-x)}{C(x)C(-x) - 1},$$

then we have

$$\det H_n(1,-1) = (-1)^{\binom{n}{2}} h_n.$$

We shall give an example:

Example 4.3. If we take $\{c_n\} = \{(-1)^n F_{n-1}\}$ and define the sequence $\{b_n\}_{n\geq 0}$ as $b_{2n} = -b_{2n+1} = F_{2n+2}$, then for even n such that n = 2k, the matrix $H_n(1, -1)$ takes the form

$$H_{2k}(1,-1) = \begin{bmatrix} F_2 & 1 & & & & 0 \\ -F_2 & 0 & -1 & & & \\ F_4 & F_1 & 0 & 1 & & & \\ -F_4 & -F_2 & F_1 & 0 & -1 & & \\ \vdots & \vdots & \vdots & \cdots & \ddots & \ddots & \\ -F_{2k} & -F_{2k-2} & F_{2k-3} & \cdots & \cdots & 0 & -1 \\ F_{2k+2} & F_{2k-1} & -F_{2k-2} & \cdots & \cdots & F_1 & 0 \end{bmatrix}$$

and for odd n such that n = 2k - 1, the matrix $H_n(1, -1)$ takes the form

$$H_{2k-1}(1,-1) = \begin{bmatrix} F_2 & 1 & & & & 0 \\ -F_2 & 0 & -1 & & & \\ F_4 & F_1 & 0 & 1 & & & \\ -F_4 & -F_2 & F_1 & 0 & -1 & & \\ \vdots & \vdots & \vdots & \cdots & \ddots & \ddots & \\ F_{2k} & F_{2k-2} & -F_{2k-3} & \cdots & \cdots & 0 & -1 \\ -F_{2k} & -F_{2k-2} & F_{2k-2} & \cdots & \cdots & F_1 & 0 \end{bmatrix}.$$

So that

$$\det H_{2k}(1,-1) = (-1)^k F_{2k+1}$$
 and $\det H_{2k-1}(1,-1) = (-1)^k F_{2k}$.

Proof. The generating functions of $\{b_n\}_{n\geq 0}$ and $\{c_n\}$ are $B(x)=\frac{1-x}{(1+x-x^2)(1-x-x^2)}$ and $C(x)=\frac{x^2}{1+x-x^2}$, respectively. So we get $H(x)=\frac{1}{1-x-x^2}$. By Corollary 4.17, the claim follows.

Corollary 4.18. For $d \neq 0$, if

$$H(x) = \frac{B(x)}{C(x) + d},$$

then

$$\det H_n(d, d) = (-1)^n d^{n+1} h_n$$

and the generating function of $\{\det H_n(d,d)\}_{n\geq 0}$ is

$$\mathcal{H}(x) = d \cdot H(-dx).$$

Example 4.4. If $b_n = -(\mathcal{H}_n + 1)$ with $b_0 = -1$ and $c_n = \frac{2}{n}$, then

$$\begin{vmatrix}
-1 & 2 & & & 0 \\
-(\mathcal{H}_1+1) & 2 & 2 & & \\
-(\mathcal{H}_2+1) & 1 & 2 & 2 & & \\
-(\mathcal{H}_3+1) & \frac{2}{3} & 1 & 2 & 2 & \\
\vdots & \vdots & \vdots & \cdots & \ddots & \ddots & \\
-(\mathcal{H}_{n-1}+1) & \frac{2}{n-1} & \frac{2}{n-2} & \cdots & \cdots & 2 & 2 \\
-(\mathcal{H}_n+1) & \frac{2}{n} & \frac{2}{n-1} & \cdots & \cdots & 1 & 2
\end{vmatrix} = (-1)^{n-1} 2^n.$$

Proof. If we take d=2, $b_n=-(\mathcal{H}_n+1)$ with $b_0=-1$ and $c_n=\frac{2}{n}$ in Corollary 4.18, then we get

$$B(x) = \frac{\ln(1-x) - 1}{1-x}$$
 and $C(x) = \ln(1-x)^{-2}$.

Thus $H(x) = \frac{1}{2x-2}$ and $\det H_n = 2(-2)^n h_n$, which give us $\det H_n = (-1)^{n-1} 2^n$, as claimed.

When $c_0 = d$, by Corollary 4.18, we obtain $H(x) = \frac{B(x)}{C(x)}$, where $C(x) = \sum_{k \geq 0} c_k x^k$. For example, if we choose $B(x) = x + 4x^2 + x^3$ and $C(x) = (1 - x)^4$, then

$$\begin{vmatrix} 0 & 1 & & & & & 0 \\ 1 & -4 & 1 & & & & \\ 4 & 6 & -4 & 1 & & & & \\ 1 & -4 & 6 & -4 & 1 & & & \\ 0 & 1 & -4 & 6 & \ddots & \ddots & & \\ 0 & 0 & 1 & -4 & \ddots & -4 & 1 & & \\ \vdots & \vdots & \vdots & \vdots & \ddots & 6 & -4 & 1 \\ 0 & 0 & 0 & 0 & \cdots & -4 & 6 & -4 \end{vmatrix}_{(n+1)\times(n+1)} = (-1)^n n^3.$$

Now we recall an already known result given in [89]. But we will give an alternative and simpler proof for it.

Corollary 4.19. If $\{c_n\}_{n\geq 0}$ is any sequence such that $c_0\neq 0$, then we have

$$\begin{vmatrix} c_1 & c_0 & 0 & \cdots & 0 \\ c_2 & c_1 & c_0 & \cdots & 0 \\ c_3 & c_2 & c_1 & \cdots & 0 \\ \vdots & \vdots & \vdots & \ddots & \vdots \\ c_n & c_{n-1} & c_{n-2} & \cdots & c_1 \end{vmatrix}_{n \times n} = [x^n] \frac{c_0}{C(-c_0 x)},$$

where $C\left(x\right)=\sum_{k\geq0}c_{k}x^{k}$ and $\left[\circ\right]$ is the coefficient extraction operator, i.e. $\left[x^{n}\right]\sum_{k\geq0}a_{k}x^{k}=a_{n}$.

Proof. First we consider an equal determinant to the claimed determinant by the following equality

$$\begin{vmatrix} c_1 & c_0 & 0 & \cdots & 0 \\ c_2 & c_1 & c_0 & \cdots & 0 \\ c_3 & c_2 & c_1 & \cdots & 0 \\ \vdots & \vdots & \vdots & \ddots & \vdots \\ c_n & c_{n-1} & c_{n-2} & \cdots & c_1 \end{vmatrix}_{n \times n} = \begin{vmatrix} 1 & c_0 & 0 & 0 & \cdots & 0 \\ 0 & c_1 & c_0 & 0 & \cdots & 0 \\ 0 & c_2 & c_1 & c_0 & \cdots & 0 \\ 0 & c_3 & c_2 & c_1 & \cdots & 0 \\ \vdots & \vdots & \vdots & \vdots & \ddots & \vdots \\ 0 & c_n & c_{n-1} & c_{n-2} & \cdots & c_1 \end{vmatrix}_{(n+1) \times (n+1)}$$

The value of the determinant on the RHS of the above equation could be easily found by Corollary 4.18. So that the claimed result directly follows. \Box

Let's give an example related to Theorem 4.35.

Example 4.5. Let $\{b_n\}_{n\geq 0}$ be the alternating of the sequence A135491 in [94]. Then for n=2k,

$$\begin{vmatrix} b_0 & 1 & & & & 0 \\ b_1 & 1 & -3 & & & & \\ b_2 & 1 & 1 & 1 & & & \\ b_3 & 1 & 1 & 1 & -3 & & & \\ \vdots & \vdots & \vdots & \ddots & \ddots & \ddots & \\ b_{2k-1} & 0 & 0 & \cdots & \cdots & 1 & -3 \\ b_{2k} & 0 & 0 & \cdots & \cdots & 1 & 1 \end{vmatrix} = (-3)^k T_{2k+2}.$$

Similarly, for n = 2k + 1, determinant of the corresponding Hessenberg matrix is equal to $-T_{2k+3}(-3)^{k+1}$, where T_n stands for the nth Tribonacci number defined in Table 1.

Proof. The generating functions of $\{b_n\}_{n\geq 0}$ and $\{c_n\}$ are $B(x) = \frac{1-x+x^2-x^3}{1+x-x^2+x^3}$ and $C(x) = x + x^2 + x^3$, respectively. By Theorem 4.35, when r = 1 and s = -3, we obtain $\det H_n = T_{n+2}(-3)^k$ for n = 2k and $\det H_n = -T_{n+2}(-3)^{k+1}$ for n = 2k + 1, as desired.

Up to now, we consider certain Hessenberg matrices whose superdiagonal are constant or two periodic. Now we give a general idea for Hessenberg matrices with arbitrary superdiagonal entries. To show how this idea will be applied, we present two Hessenberg matrices whose superdiagonals will consist of the terms of two special sequences, $\{n\}$ and $\{2^{n-1}\}$, respectively.

Let $\{b_n\}_{n\geq 0}$, $\{c_n\}$ and $\{d_n\}_{n\geq 0}$ such that $d_n\neq 0$ for all $n\geq 0$ be any sequences. First define the Hessenberg matrix H_n of order n+1 as

$$H_n := \begin{bmatrix} b_0 & d_0 & & & & & 0 \\ b_1 & c_1 & d_1 & & & & \\ b_2 & c_2 & c_1 & d_2 & & & & \\ b_3 & c_3 & c_2 & c_1 & d_3 & & & \\ \vdots & \vdots & \vdots & \cdots & \ddots & \ddots & \\ b_{n-1} & c_{n-1} & c_{n-2} & \cdots & \cdots & c_1 & d_{n-1} \\ b_n & c_n & c_{n-1} & \cdots & \cdots & c_2 & c_1 \end{bmatrix}.$$

Now consider the following infinite linear system of equations

$$\begin{bmatrix} d_0 & & & & & & & & \\ c_1x & d_1x & & & & & & \\ c_2x^2 & c_1x^2 & d_2x^2 & & & & \\ c_3x^3 & c_2x^3 & c_1x^3 & d_3x^3 & & & & \\ c_4x^4 & c_3x^4 & c_2x^4 & c_1x^4 & d_4x^4 & & & \\ \vdots & \vdots & \vdots & \vdots & \vdots & \ddots & & \vdots & \vdots \end{bmatrix} \begin{bmatrix} h_0 \\ h_1 \\ h_2 \\ h_3 \\ h_4 \\ \vdots & \vdots & \vdots & \vdots \\ \vdots & \vdots & \vdots \end{bmatrix},$$

which gives us the relation

$$H(x)C(x) + \sum_{k=0}^{\infty} h_k d_k x^k = B(x),$$
 (4.12)

where $C(x) = \sum_{k \geq 1} c_k x^k$. If we restricted this infinite system to the first n+1 equations with x = 1, then by Cramer's rule we have

$$h_n = \frac{(-1)^n \det H_n}{\prod_{k=0}^n d_k}.$$

Unfortunately, the series $\sum_{k=0}^{\infty} h_k d_k x^k$ in (4.12), which is the generating function of the Hadamard product of the sequences $\{h_n\}_{n\geq 0}$ and $\{d_n\}_{n\geq 0}$, can not be always computed explicitly in terms of the generating functions H(x) and D(x). Nevertheless, it is possible to compute it for some special cases. So that one can compute the determinant of these type of matrices via generating function method. Now we present two special example to show how we can use the idea mentioned above.

Theorem 4.36. If $\{d_n\}_{n>0} = \{n+1\}$, then

$$xH(x)\left(e^{\int \frac{C(x)}{x}dx}\right) = \int e^{\int \frac{C(x)}{x}dx}B(x)dx + C,$$

with

$$\det H_n = (-1)^n (n+1)! h_n,$$

where C is a constant.

Proof. By (4.12), we have

$$H(x)C(x) + \sum_{k=0}^{\infty} h_k(k+1)x^k = B(x),$$

which, equivalently, gives us

$$H(x)C(x) + (xH(x))' = B(x).$$

By taking $y = x \cdot H(x)$, we get the first order linear differential equation

$$y\frac{C(x)}{x} + y' = B(x).$$

The solution of this differential equation is

$$y = \left(e^{\int \frac{C(x)}{x} dx}\right)^{-1} \left(\int e^{\int \frac{C(x)}{x} dx} B(x) dx + C\right),$$

which completes the proof. Note that the constant C will be determined according to the initial value y(0) = 0.

Example 4.6. For $n \geq 0$, we have

$$\begin{vmatrix} 1 & 1 & & & & & 0 \\ 3 & 1 & 2 & & & & \\ 5 & 1 & 1 & 3 & & & & \\ 7 & 1 & 1 & 1 & 4 & & & \\ \vdots & \vdots & \vdots & \cdots & \ddots & \ddots & \\ 2n-1 & 1 & 1 & \cdots & \cdots & 1 & n \\ 2n+1 & 1 & 1 & \cdots & \cdots & 1 & 1 \end{vmatrix} = (-1)^n (n+1)!.$$

Proof. Since $b_n = 2n + 1$ and $c_n = 1$, we obtain $B(x) = \frac{x+1}{(x-1)^2}$ and $C(x) = \frac{x}{1-x}$. So we get

$$\int \frac{1}{1-x} dx = -\ln(x-1) \text{ and } e^{\int \frac{C(x)}{x} dx} = \frac{1}{x-1}.$$

By Theorem 4.36, we have that

$$xH(x)\frac{1}{x-1} = \int \frac{x+1}{(x-1)^3} dx + C$$
$$xH(x)\frac{1}{x-1} = -\frac{x}{(x-1)^2} + C.$$

For x = 0, we find that C = 0 and so

$$H(x) = \frac{1}{1 - x},$$

which gives det $H_n = (-1)^n (n+1)!$.

For the case $b_n = c_{n+1}$, i.e. $B(x) = \frac{C(x)}{x}$, the relation given in Theorem 4.36 turns

$$xH(x) = 1 + C\left(e^{\int \frac{C(x)}{x} dx}\right)^{-1}.$$

Now we present the other interesting special case with an example which could be produced by the help of relation (4.12).

Example 4.7. For $n \geq 0$,

Proof. Since $b_n = \frac{2^{n-1}(n+2)}{n!}$ and $c_n = \frac{1}{(n-1)!}$, their generating functions are $B(x) = \frac{1}{n!}$ $e^{2x}(x+1)$ and $C(x)=xe^x$, respectively. By (4.12), we have

$$xe^{x}H(x) + H(2x) = e^{2x}(x+1).$$

Hence we find that $H(x) = e^x$, which gives $h_n = \frac{1}{n!}$. Finally, from the relation $h_n =$ $\frac{(-1)^n \det H_n}{2^{\binom{n+1}{2}}}$, we obtain claimed result. Now we consider different two classes of Hessenberg determinants, which have not been studied before. We start with the first one: For any nonzero real d, we define a Hessenberg matrix of order n + 1 as follows:

$$H_n := \begin{bmatrix} b_0 & d & & & & & 0 \\ b_1 & c_1 & d & & & & \\ b_2 & c_2 & d_1 & d & & & \\ b_3 & c_3 & d_2 & d_1 & d & & \\ \vdots & \vdots & \vdots & \cdots & \ddots & \ddots & \\ b_{n-1} & c_{n-1} & d_{n-2} & \cdots & \cdots & d_1 & d \\ b_n & c_n & d_{n-1} & \cdots & \cdots & d_2 & d_1 \end{bmatrix}.$$

Theorem 4.37. If

$$H(x) = \frac{B(x) + h_0 D(x) - h_0 C(x)}{D(x) + d} \quad with \ h_0 = b_0/d, \tag{4.13}$$

where B(x) and C(x) defined as before and $D(x) = \sum_{k \geq 1} d_k x^k$, then

$$\det H_n = (-1)^n d^{n+1} h_n$$

and the generating function of $\{\det H_n\}_{n\geq 0}$ is

$$\mathcal{H}(x) = d \cdot H(-dx).$$

Proof. Similar to the proof of Theorem 4.35, we have the following infinite linear system of equations

$$\begin{bmatrix} d & & & & & & & \\ c_1 x & dx & & & & & \\ c_2 x^2 & d_1 x^2 & dx^2 & & & \\ c_3 x^3 & d_2 x^3 & d_1 x^3 & dx^3 & & \\ c_4 x^4 & d_3 x^4 & d_2 x^4 & d_1 x^4 & dx^4 & \\ \vdots & \vdots & \vdots & \vdots & \vdots & \ddots & \end{bmatrix} \begin{bmatrix} h_0 \\ h_1 \\ h_2 \\ h_3 \\ h_4 \\ \vdots & \vdots & \vdots & \vdots \\ h_4 \end{bmatrix} = \begin{bmatrix} b_0 \\ b_1 x \\ b_2 x^2 \\ b_3 x^3 \\ b_4 x^4 \\ \vdots & \vdots & \vdots \\ \vdots & \vdots & \vdots \end{bmatrix}.$$

By summing the equations come from this infinite linear system of equations and adding $h_0D(x)$ to the both sides of it, we obtain

$$h_0C(x) + H(x)D(x) + h_0H(x) = B(x) + h_0D(x),$$

which gives

$$H(x) = \frac{B(x) + h_0 D(x) - h_0 C(x)}{D(x) + d},$$

as desired. Finally, if we restrict this linear system of equations to the first (n +1) equations and take x=1, then by Cramer's rule, we get $h_n=\frac{(-1)^n\det H_n}{d^{n+1}}$, as ${
m claimed}.$

As an example,

Example 4.8. For n > 0,

where F_n and P_n are the nth Fibonacci and Pell numbers, given in Table 1.

Proof. It is a consequence of Theorem 4.37. When
$$d=1$$
, $B(x)=\sum_{k\geq 0}P_{k+1}x^k=\frac{1}{1-2x-x^2}$, $C(x)=\sum_{k\geq 1}F_{k+1}x^k=\frac{x+x^2}{1-x-x^2}$ and $D(x)=\sum_{k\geq 1}P_{k+1}x^k=\frac{2x+x^2}{1-2x-x^2}$, then
$$H(x)=\frac{1-2x-x^2}{1-x-x^2},$$

which completes the proof.

Now we define the second class of Hessenberg matrices of order n+1, whose columns are periodic after first column, as follows:

where

$$s(n,k) = \begin{cases} c_k & \text{if } n \text{ is even,} \\ d_k & \text{if } n \text{ is odd.} \end{cases}$$

We have the following theorem for the generating function of the determinant of the just above matrix.

Theorem 4.38. If

$$H(x) = \frac{B(x)(C(-x) + D(-x) + 2) - B(-x)(C(x) - D(x))}{C(x)(1 + D(-x)) + D(x)(1 + C(-x)) + (C(-x) + D(-x)) + 2d},$$

then

$$\det H_n = (-1)^n d^{n+1} h_n$$

and the generating function of $\{\det H_n\}_{n\geq 0}$ is

$$\mathcal{H}(x) = d \cdot H(-dx).$$

Proof. Similar to the previous theorems, if we consider the infinite linear system of equations, then we obtain

$$C(x)\sum_{k>0}h_{2k}x^{2k} + D(x)\sum_{k>0}h_{2k+1}x^{2k+1} + dH(x) = B(x).$$
(4.14)

By the formulæ (2.14), the equation (4.14) is written as

$$H(x)\left(\frac{C(x) + D(x)}{2} + 1\right) + H(-x)\left(\frac{C(x) - D(x)}{2}\right) = B(x),$$

which, by solving in terms of H(x), gives us

$$H(x) = \frac{B(x)(C(-x) + D(-x) + 2) - B(-x)(C(x) - D(x))}{C(x)(1 + D(-x)) + D(x)(1 + C(-x)) + (C(-x) + D(-x)) + 2d},$$

as desired. When we restricted the infinite system of equations to the first n+1 equations with x=1, we complete the proof by Cramer's rule.

Example 4.9. For even n, we have

$$\begin{vmatrix} L_0 & 1 & & & & & 0 \\ L_1 & F_1 & 1 & & & & \\ L_2 & F_2 & L_0 & 1 & & & \\ L_3 & F_3 & L_1 & F_1 & 1 & & \\ L_4 & F_4 & L_2 & F_2 & L_0 & \ddots & \\ \vdots & \vdots & \vdots & \vdots & \cdots & \ddots & 1 \\ L_{n-1} & F_{n-1} & L_{n-3} & F_{n-3} & L_{n-5} & \cdots & F_1 & 1 \\ L_n & F_n & L_{n-2} & F_{n-2} & L_{n-4} & \cdots & F_2 & L_0 \end{vmatrix} = 2^{\frac{n}{2}} + 1.$$

If n = 2k + 1, the determinant of the corresponding matrix is equal to 2^k .

Proof. Since $b_n = L_n$, $c_n = F_n$ and $d_n = L_{n-1}$, we have $B(x) = \frac{2-x}{1-x-x^2}$, $C(x) = \frac{x}{1-x-x^2}$ and $D(x) = \frac{2x-x^2}{1-x-x^2}$. Hence, for d = 1 by Theorem 4.38, we obtain

$$A(x) = \frac{-x - 3x^2 + x^3 + 2}{(x - 1)(x + 1)(2x^2 - 1)} = \frac{1}{1 - x^2} + \frac{1 + x}{1 - 2x^2}$$
$$= \sum_{k=0}^{\infty} x^{2k} + \sum_{k=0}^{\infty} 2^k x^{2k} + \sum_{k=0}^{\infty} 2^k x^{2k+1}$$
$$= \sum_{k=0}^{\infty} (2^k + 1)x^{2k} + \sum_{k=0}^{\infty} 2^k x^{2k+1},$$

as claimed. \Box

Remark 4.3. This generating function method works for only Hessenberg matrices. If a matrix has nonzero two superdiagonal bands, then the corresponding infinite linear system of equations is inconsistent. So we can't apply the same steps for this matrix. In that case, we may reduce this matrix to a Hessenberg matrix by applying some row or column operations. Then we may use Theorems 4.37 or 4.35. If the number of nonzero superdiagonals is increased, then computing their determinants via generating functions would become harder and more complicated.

4.7.2 A Matrix Method to Compute a Class of Hessenberg Determinants

Now we give a new and simple method to compute a class of Hessenberg determinants whose entries consist of the terms of the higher order linear recursive sequence with constant coefficients.

Consider the following lower Hessenberg matrix of order n for nonzero real r:

$$E_n(r) = \begin{bmatrix} u_1 & r & & & & & 0 \\ u_2 & u_1 & r & & & & \\ u_3 & u_2 & u_1 & r & & & \\ u_4 & u_3 & u_2 & u_1 & \ddots & & \\ \vdots & \vdots & \vdots & \vdots & \ddots & \ddots & \\ u_{n-1} & u_{n-2} & u_{n-3} & u_{n-4} & \cdots & u_1 & r \\ u_n & u_{n-1} & u_{n-2} & u_{-3} & \cdots & u_2 & u_1 \end{bmatrix},$$

where the terms u_n 's are defined as in (2.1). Briefly, we use E_n instead of $E_n(r)$.

Indeed one can compute the determinant of the matrix E_n by using the results of Section 4.7.1. Here we would like to present a new and simple method to compute $\det E_n$. For this, we define an adjacency-factor matrix related to the matrix E_n : Define a lower triangular adjacency-factor matrix M of order n as

$$M_{ij} = \begin{cases} 1 & \text{if } i = j, \\ -p_{i-j} & \text{if } 1 \le i - j \le k, \\ 0 & \text{otherwise.} \end{cases}$$

Clearly the matrix M is of the form

Then we obtain that

$$ME_n = \widehat{E}_n$$

where

$$\left(\widehat{E}_n\right)_{ij} = \begin{cases} r & \text{if } j = i+1, \\ b_i & \text{if } j = 1 \text{ and } i \leq k, \\ d_{i-j+1} & \text{if } i \geq j > 1 \text{ and } i-j \leq k-1, \\ 0 & \text{otherwise,} \end{cases}$$

with

$$b_m = u_m - \sum_{l=1}^{m-1} u_{m-l} p_l$$
 and $d_m = u_m - \sum_{l=1}^{m-1} u_{m-l} p_l - r p_m$,

for $1 \le m \le k$.

Here since $\det M = 1$, we have $\det E_n = \det \widehat{E}_n$. Afterwards, we prefer to compute the value of the determinant of the matrix \widehat{E}_n instead of the matrix E_n because the matrix \widehat{E}_n is a banded matrix with bandwidth (k+1) and includes many zeros and so it gives us advantage to choose the matrix \widehat{E}_n rather than E_n regard to use of the results of Section 4.7.1. That means one can easily apply the results given in Section 4.7.1 to the matrix \widehat{E}_n with less computation.

For example, when r = -1, by Corollary 4.16, we have that

$$\sum_{i>0} \det E_{i+1}(-1)x^i = \sum_{i>0} \det \widehat{E}_{i+1}(-1)x^i = \frac{\sum_{i=1}^k b_i x^{i-1}}{1 - \sum_{i=1}^k d_i x^i}.$$
 (4.15)

In [88], author computed the determinant of the matrix $E_n(-1)$ when k=2 by using the cofactor expansion and he only gave complicated formula for the case k=3 without proof. Our method is simpler to determine those formulæ and also one can find related formula for larger k with less effort.

As a special case, if we consider the recurrence relation of the sequence $\{u_n\}$ defined in (2.1) with the initials $u_{-k+2} = u_{-k+3} = \cdots = u_{-1} = u_0 = 0$ and $u_1 = 1$, then we have

$$b_1 = 1$$
 and $b_i = 0$ for $1 < i \le k$,
 $d_1 = 1 + p_1$ and $d_i = p_i$ for $1 < i \le k$.

Hence the generating function of the determinant of the matrix $E_{n+1}(-1)$ is written as

$$\frac{1}{1 - (1 + p_1)x - p_2x^2 - \dots - p_kx^k}. (4.16)$$

Now we give an example to show how to use the method described above.

Example 4.10. For positive integer m, define the sequence $\{u_n\}$ with $u_n = \binom{m+n-1}{m}$ and construct the following matrix $G_n(m)$ of order n

$$G_n(m) := \begin{bmatrix} \binom{m}{m} & -1 & & & & & & & & & \\ \binom{m+1}{m} & \binom{m}{m} & -1 & & & & & \\ \binom{m+2}{m} & \binom{m+1}{m} & \binom{m}{m} & \ddots & & & & \\ \binom{m+3}{m} & \binom{m+2}{m} & \binom{m+1}{m} & \cdots & -1 & & & \\ & \ddots & & \ddots & & \ddots & & \binom{m}{m} & -1 & \\ \binom{m+n-2}{m} & \binom{m+n-3}{m} & \binom{m+n-4}{m} & \vdots & \binom{m+1}{m} & \binom{m}{m} & -1 \\ \binom{m+n-1}{m} & \binom{m+n-2}{m} & \binom{m+n-3}{m} & \vdots & \binom{m+2}{m} & \binom{m+1}{m} & \binom{m}{m} \end{bmatrix}.$$

Then

$$\det G_{n+1}(m) = \sum_{k=0}^{n} \binom{(m+1)n + m(1-k)}{k}.$$

Proof. We should find the recurrence relation for the sequence $\{u_n\}$. From [13], we recall the Equation 5.24: For $l \geq 0$ and integers m, n,

$$\sum_{k} {l \choose m+k} {s+k \choose n} (-1)^k = (-1)^{l+m} {s-m \choose n-l},$$

which is a variant of the Vandermonde's identity. If we choose $l \to m+1, m \to 1$, $s \to m-n$ and $n \to m$ in the equation above, then we obtain

$$\sum_{k=-1}^{m} (-1)^k \binom{m+1}{k+1} \binom{n-k-1}{m} = \sum_{k=-1}^{m} (-1)^{k+m} \binom{m+1}{k+1} \binom{m-n+k}{m}$$
$$= \binom{n-m-1}{-1} = 0.$$

So, we can deduce that

$$\sum_{k=0}^{m} (-1)^k \binom{m+1}{k+1} \binom{n-k-1}{m} = \binom{n}{m}.$$

If we take n = n + m - 1, then we get the recurrence relation of order m + 1 for the sequence $\{u_n\}$:

$$u_n = \sum_{k=0}^{m} (-1)^k {m+1 \choose k+1} u_{n-k-1},$$

with $u_{-m+1} = u_{-m+2} = \cdots = u_{-1} = u_0 = 0$ and $u_1 = 1$. By our method, we see that the adjacency-factor matrix for the matrix $G_n(m)$ is

$$M_{ij} = (-1)^{i-j} \binom{m+1}{i-j}.$$

Thus by (4.16), we find the generating function of the sequence $\{\det G_{n+1}(m)\}_{n\geq 0}$ as follows

$$\frac{1}{1 - \left(1 + \binom{m+1}{1}\right)x + \binom{m+1}{2}x^2 - \dots - (-1)^m \binom{m+1}{m+1}x^{m+1}} = \frac{1}{(1 - x)^{m+1} - x}.$$

In other words, we have that

$$[x^n] \frac{1}{(1-x)^{m+1} - x} = \det G_{n+1}(m). \tag{4.17}$$

To prove this claim, by Theorem 2.5 it is sufficient to show that

$$\sum_{n\geq 0} \sum_{k=0}^{n} \binom{(m+1)n + m(1-k)}{k} x^n = \frac{1}{(1-x)^{m+1} - x}.$$

Consider,

$$\sum_{n\geq 0} \sum_{k=0}^{n} \binom{(m+1)n + m(1-k)}{k} x^n = \sum_{k\geq 0} \sum_{n\geq k} \binom{(m+1)n + m(1-k)}{k} x^n$$
$$= \sum_{n\geq 0} x^n \sum_{k\geq 0} \binom{m+n+mn+k}{k} x^k$$

$$= \frac{1}{(1-x)^{m+1}} \sum_{n\geq 0} \left(\frac{x}{(1-x)^{m+1}}\right)^n$$
$$= \frac{1}{(1-x)^{m+1} - x},$$

which completes the proof.

As a special case for m = 1, we get

$$\begin{vmatrix} 1 & -1 & & & 0 \\ 2 & 1 & -1 & & & \\ 3 & 2 & \ddots & \ddots & & \\ \dots & \dots & \dots & 1 & -1 & 0 \\ n-1 & n-2 & \vdots & 2 & 1 & -1 \\ n & n-1 & \vdots & 3 & 2 & 1 \end{vmatrix} = \sum_{k=0}^{n-1} {2n-k-1 \choose k} = F_{2n},$$

which could be also found in [88].

REFERENCES

- [1] Hoskins, W.D., Ponzo, P.J., Some properties of a class of band matrices, Mathematics of Computation, 26, 393–400, **1972**.
- [2] Arıkan, T., Kılıç, E., A class of non-symmetric band determinants with the Gaussian q-binomial coefficients, Quaestiones Mathematicae, 40(5), 645–660, **2017**.
- [3] Prodinger, H., The reciprocal super Catalan matrix, *Special Matrices*, 3, 111–117, **2015**.
- [4] Kılıç, E., Arıkan, T., The generalized reciprocal super Catalan matrix, *Turkish Journal of Mathematics*, 40, 960–972, **2016**.
- [5] Mattila, M., Haukkanen, P., Studying the various properties of MIN and MAX matrices elementary vs. more advanced methods, *Special Matrices*, 4(1), 101–109, **2016**.
- [6] Kılıç, E., Arıkan, T., A nonlinear generalizations of the Filbert matrix and its Lucas analogue, accepted in Linear and Multilinear Algebra, DOI: 10.1080/03081087.2017.1412393.
- [7] Getu, S., Evaluating determinants via generating functions, *Mathematics Magazine*, 64(1), 45–53, **1991**.
- [8] Kılıç, E., Arıkan, T., Evaluation of Hessenberg determinants via generating function approach, *Filomat*, 31(15), 4945–4962, **2017**.
- [9] Carlitz, L., A note on Fibonacci numbers, *The Fibonacci Quarterly*, 2, 15–28, **1964**.
- [10] Horadam, A.F., Basic properties of a certain generalized sequence of numbers, *The Fibonacci Quarterly*, 3, 161–176, **1965**.

- [11] Koshy, T., Fibonacci and Lucas Numbers with Applications, John Wiley and Sons, New York, **2001**.
- [12] Vajda, S., Fibonacci and Lucas Numbers and the Golden Section, Dover Publication Inc., New York, 1989.
- [13] Graham, R.L., Knuth, D.E., Patashnik, P., Concrete Mathematics, Addison-Wesley, 1994.
- [14] Jarden, D., Motzkin, T., The product of sequences with a common linear recursion formula of order 2, *Riveon Lematematika*, 3, 25–27, **1949**.
- [15] Gould, H.W., The bracket function and Fountené–Ward generalized binomial coefficients with application to fibonomial coefficients, *The Fibonacci Quarterly*, 7, 23–40, **1969**.
- [16] Hoggatt Jr., V.E., Fibonacci numbers and generalized binomial coefficients, The Fibonacci Quarterly, 5, 383–400, **1967**.
- [17] Kılıç, E., The generalized Fibonomial matrix, European Journal of Combinatorics, 31, 193–209, **2010**.
- [18] Torretto, R., Fuchs, A., Generalized binomial coefficients, *The Fibonacci Quarterly*, 2, 296–302, **1964**.
- [19] Andrews, G.E., Askey, R., Roy, R., Special Functions, Cambridge University Press, 1999.
- [20] Ernst, T., A method for q-calculus, Journal of Nonlinear Mathematical Physics, 10(4), 487–525, **2003**.
- [21] Kac, V., Cheung, P., Quantum Calculus, Springer, 2002.
- [22] Stanley, R.P., Enumerative Combinatorics, Volume 1, Cambridge University Press, 2nd edition, **2011**.
- [23] Wilf, H.S., Generatingfunctionology, Academic Press, San Diego, 2nd edition, 1994.

- [24] Flajolet, P., Sedgewic, R., *Analytic Combinatorics*, Cambridge University Press, Cambridge, **2009**.
- [25] Kauers, M., Paule, P., The Concrete Tetrahedron, Springer, Wien, 2011.
- [26] Bailey, W.N., Generalized Hypergeometric Series, Cambridge University Press, Cambridge, 1935.
- [27] Gasper, G., Rahman, M., *Basic Hypergeometric Series*, Cambridge University Press, **1990**.
- [28] Petkovšek, M., Wilf, H.S., Zeilberger, D., A = B, A. K. Peters Ltd., Wellesley, MA, **1996**.
- [29] Exton, H., q-Hypergeometric Functions and Applications, Halsted/Wiley, New York, 1983.
- [30] Fasenmyer, S.M.C., Some generalized hypergeometric Polynomials, Technical report, Ph.D. Dissertation, University of Michigan, USA, 1945.
- [31] Gosper, R.W., Decision procedures for indefinite hypergeometric summation,

 Proceedings of the National Academy of Sciences of the United States of

 America, 75, 40–42, 1978.
- [32] Zeilberger, D., A holonomic systems approach to special function identities,

 Journal of Computational and Applied Mathematics, 32, 321–368, 1990.
- [33] Zeilberger, D., A fast algorithm for proving terminating hypergeometric identities, *Discrete Mathematics*, 80, 207–211, **1990**.
- [34] Zeilberger, D., The method of creative telescoping, *Journal of Symbolic Computation*, 11, 195–204, **1991**.
- [35] Zeilberger, D., Maple program for proving hypergeometric identities, SIGSAM Bulletin, 25, 4–13, 1991.
- [36] Koornwinder, T.H., On Zeilberger's algorithm and its q-analogue, Journal of Computational and Applied Mathematics, 48, 91–111, 1993.

- [37] Paule, P., Shorn, M., A Mathematica version of Zeilberger's algorithm for proving binomial coefficient identities, *Journal of Symbolic Computation*, 20, 673–698, **1995**.
- [38] Research Institute for Symbolic Computation, R.I.S.C., http://www.risc.jku.at/, (2018, August).
- [39] Wilf, H.S., Zeilberger, D., Rational function certification of multi-sum/integral/ 'q'identities, Bulletin of the American Mathematical Society, 27, 148–153, **1992**.
- [40] Wilf, H.S., Zeilberger, D., An algorithmic proof theory for hypergeometric (ordinary and "q") multisum/integral identities, *Inventiones Mathematicae*, 108, 575–633, **1992**.
- [41] Paule, P., Riese, A., A Mathematica q-analogue of Zeilberger's algorithm based on an algebraically motivated approach to q-hypergeometric telescoping, in special functions, q-series and related topics, Fields Institute Communications, 14, 179–210, 1997.
- [42] Brualdi, R.A., Cvetkovic, D., A Combinatorial Approach to Matrix Theory and Its Applications, Chapman and Hall/CRC, 2008.
- [43] Horn, R.A., Johnson, C.R., *Matrix Analysis*, Cambridge University Press, **1985**.
- [44] Krattenthaler, C., Advanced determinant calculus, Séminaire Lotharingien de Combinatoire, 42, 349–426, **1999**.
- [45] Krattenthaler, C., Advanced determinant calculus: A complement, *Linear Algebra and its Applications*, 411(1), 68–166, **2005**.
- [46] Okunev, P., Johnson, C.R., Necessary and sufficient conditions for existence of the *LU* factorization of an arbitrary matrix, *eprint arXiv:math/0506382*, 1–21, **2005**.
- [47] Golub, G.H., Van Loan, C.F., *Matrix Computations*, Baltimore: Johns Hopkins, 3rd edition, **1996**.

- [48] Pan, V.Y., Structured Matrices and Polynomials, Springer-Verlag, New York
 USA, 2001.
- [49] Newman, M., Todd, J., The evaluation of matrix inversion programs, *Journal* of the Society for Industrial and Applied Mathematics, 6, 466–476, **158**.
- [50] Bini, D., Capovani, M., Spectral and computational properties of band symmetric toeplitz matrices, *Linear Algebra and its Applications*, 52/53, 99–126, 1983.
- [51] Kılıç, E., Explicit formula for the inverse of a tridiagonal matrix by backward continued fractions, *Applied Mathematics and Computation*, 197, 345–357, **2008**.
- [52] Kılıç, E., Stanica, P., The inverse of banded matrices, *Journal of Computational and Applied Mathematics*, 237(1), 126–135, **2013**.
- [53] Li, H.B., Huang, T.Z., Liu, X.P., Li, H., On the inverses of general tridiagonal matrices, *Linear Algebra and its Applications*, 433(5), 965–983, **2010**.
- [54] Papatheodorou, T.S., Inverses for a class of banded matrices and applications to piecewise cubic approximation, *Journal of Computational and Applied Mathematics*, 8(4), 285–288, **1982**.
- [55] Trench, W.F., On the eigenvalue problem for toeplitz band matrices, *Linear Algebra and its Applications*, 64, 199–214, **1985**.
- [56] Carlitz, L., The characteristic polynomial of a certain matrix of binomial coefficients, *The Fibonacci Quarterly*, 3, 81–89, **1965**.
- [57] Edelman, A., Strang, G., Pascal matrices, *The American Mathematical Monthly*, 111, 189–197, **2004**.
- [58] Prodinger, H., Factorizations related to the reciprocal Pascal matrix, *Turkish Journal of Mathematics*, 40, 986–994, **2016**.
- [59] Franc, W.L., Computing eigenvalues of complex matrices by determinant evaluation and by methods of Danilewski and Wielandt, *Journal of the Society for Industrial and Applied Mathematics*, 6, 378–392, **1958**.

- [60] Choi, M., Tricks or treats with the hilbert matrix, *The American Mathematical Monthly*, 90(5), 301–312, **1983**.
- [61] Trench, W.F., Eigenvalues and eigenvectors of two symmetric matrices, *The Bulletin of the International Linear Algebra Society*, 22, 28–29, **1999**.
- [62] da Fonseca, C.M., On the eigenvalues of some tridiagonal matrices, *Journal* of Computational and Applied Mathematics, 200, 283–286, **2007**.
- [63] Haukkanen, P., On meet matrices on posets, *Linear Algebra and its Applications*, 249, 111–123, **1996**.
- [64] Richardson, T., The Filbert matrix, The Fibonacci Quarterly, 39(3), 268–275, **2001**.
- [65] Kılıç, E., Prodinger, H., A generalized Filbert matrix, *The Fibonacci Quarterly*, 48(1), 29–33, **2010**.
- [66] Prodinger, H., A generalization of a Filbert matrix with 3 additional parameters, Transactions of the Royal Society of South Africa, 65, 169–172, **2010**.
- [67] Kılıç, E., Prodinger, H., Variants of the Filbert matrix, *The Fibonacci Quarterly*, 51(2), 153–162, **2013**.
- [68] Kılıç, E., Prodinger, H., The q-Pilbert matrix, International Journal of Computer Mathematics, 89, 1370–1377, **2012**.
- [69] Kılıç, E., Prodinger, H., The generalized q-Pilbert matrix, Mathematica Slovaca, 64(5), 1083–1092, **2014**.
- [70] Kılıç, E., Prodinger, H., Asymmetric generalizations of the Filbert matrix and variants, *Publications de l'Institut Mathématique*, 95, 267–280, **2014**.
- [71] Kılıç, E., Prodinger, H., The generalized lilbert matrix, *Periodica Mathematica Hungarica*, 73(1), 62–72, **2016**.
- [72] Elouafi, M., Aiat Hadj, A.D., A new recursive algorithm for inverting Hessenberg matrices, *Applied Mathematics and Computation*, 214, 497–499, **2009**.

- [73] Ikebe, Y., On inverses of Hessenberg matrices, *Linear Algebra and its Applications*, 24, 93–97, **1979**.
- [74] Abderraman Marrero, J., Tomeo, V., Torrano, E., On inverses of infinite Hessenberg matrices, *Journal of Computational and Applied Mathematics*, 275, 356–365, **2014**.
- [75] Cahill, N.D., D'Errico, J.R., Narayan, D.A., Narayan, J.Y., Fibonacci determinants, *The College Mathematics Journal*, 33, 221–225, **2002**.
- [76] Cereceda, J.L., Determinantal representations for generalized Fibonacci and Tribonacci numbers, *International Journal of Contemporary Mathematical Sciences*, 9(6), 269–285, **2014**.
- [77] Kılıç, E., Taşcı, D., On sums of second order linear recurrences by Hessenberg matrices, *Rocky Mountain Journal of Mathematics*, 38(2), 531–544, **2008**.
- [78] Kılıç, E., Taşcı, D., On the generalized Fibonacci and Pell sequences by Hessenberg matrices, Ars Combinatoria, 94, 161–174, **2010**.
- [79] Ramírez, J.L., On convolved generalized Fibonacci and Lucas polynomials, Applied Mathematics and Computation, 229, 208–213, **2014**.
- [80] Ramírez, J.L., Hessenberg matrices and the generalized Fibonacci-Narayana sequence, *Filomat*, 29(7), 1557–1563, **2015**.
- [81] Cahill, N.D., Narayan, D.A., Narayan, Fibonacci and Lucas numbers as tridiagonal matrix determinants, *The Fibonacci Quarterly*, 42, 216–221, **2004**.
- [82] Kılıç, E., Taşcı, D., On the permanents of some tridiagonal matrices with applications to the Fibonacci and Lucas numbers, *Rocky Mountain Journal of Mathematics*, 37(6), 203–219, **2007**.
- [83] Kılıç, E., On the second order linear recurrences by tridiagonal matrices, *Ars Combinatoria*, 91, 11–18, **2009**.
- [84] Li, H.C., On Fibonacci-Hessenberg matrices and the Pell and Perrin numbers, *Applied Mathematics and Computation*, 218(17), 8353–8358, **2012**.

- [85] Chen, Y.H., Yu, C.Y., A new algorithm for computing the inverse and the determinant of a Hessenberg matrix, *Applied Mathematics and Computation*, 218, 4433–4436, **2011**.
- [86] Benjamin, A.T., Shattuck, M.A., Recounting determinants for a class of Hessenberg matrices, *Integers*, 7, A55, 1–7, **2007**.
- [87] M. Janjic', M., Hessenberg matrices and integer sequencess, *Journal of Integer Sequences*, 13, Article ID 10.7.8, 1–10, **2010**.
- [88] Macfarlane, A.J., Use of determinants to present identities involving Fibonacci and related numbers, *The Fibonacci Quarterly*, 48(1), 68–76, **2010**.
- [89] Merca, M., A note on the determinant of a Toeplitz-Hessenberg matrix, Special Matrices, 1, 10–16, **2013**.
- [90] Belbachir, H., Bencherif, H., Szalay, L., Unimodality of certain sequences connected to binomial coefficients, *Journal of Integer Sequences*, 10, Article 07.2.3, **2007**.
- [91] Akkuş, I., The Lehmer matrix with recursive factorial entries, *Kuwait Journal of Science*, 42(2), 37–41, **2015**.
- [92] Kılıç, E., Stanica, P., The Lehmer matrix and its recursive analogue, *The Journal of Combinatorial Mathematics and Combinatorial Computing*, 74, 193–207, **2010**.
- [93] Yang, Y., Leonard, M., Evaluating determinants of convolution-like matrices via generating functions, *International Journal of Information and Systems Sciences*, 3(4), 569–580, **2007**.
- [94] The On-Line Encyclopedia of Integer Sequences, O.E.I.S., https://oeis.org, (2018, August).

CURRICULUM VITAE

Credentials

Name, Surname : Talha Arıkan

Place of Birth : Cankaya, Ankara

Marital Status : Married

E-mail : tarikan@hacettepe.edu.tr, talhapasa@gmail.com

Address : Hacettepe University Department of Mathematics

06800 Beytepe Çankaya Ankara TURKEY

Education

High School : 2001 - 2005, Ankara Anatolian High School

BSc. : 2005 - 2011, TOBB University of Economics and Technology,

Department of Mathematics

MSc. : 2011 - 2013, TOBB University of Economics and Technology,

Department of Mathematics

PhD. : 2013 - 2018, Hacettepe University,

Department of Mathematics

Foreign Languages

English (Advanced), German (Beginner)

Work Experience

2011-2013, TOBB University of Economy and Technology, Department of Mathematics, Teaching Assistant

2013 - ..., Hacettepe University, Department of Mathematics, Research Assistant

Areas of Experiences

Linear Algebra, Combinatorics, Recursive Sequences, Special Matrices

Projects and Budgets

...

Publications

- 1. Kılıç E., Arıkan T., The generalized reciprocal super Catalan matrix, *Turkish Journal of Mathematics*, 40, 960–972, **2016.**
- 2. Arıkan T., Kılıç E., A class of non-symmetric band determinants with the Gaussian q-binomial coefficients, Quaestiones Mathematicae, 40(5), 645–660, 2017.
- 3. Kılıç E., Arıkan T., Evaluation of Hessenberg determinants via generating function approach, *Filomat*, 31(15), 4945–4962, **2017**.
- Kılıç E., Arıkan T., A nonlinear generalizations of the Filbert matrix and its Lucas analogue, accepted in Linear and Multilinear Algebra, DOI: 10.1080/03081087.2017.1412393.

Oral and Poster Presentations

- "Some Classes of Band Determinants of q-binomial Coefficients", LICMA'15 Lebanese International Conference on Mathematics and Applications, Beirut, Lebanon, 2015 May.
- "Evaluating Some Hessenberg Determinants via Generating Functions", International Conference on Approximation Theory and its Applications, Sibiu/Romania, 2016 May.
- "Hessenberg Determinants via Generating Function Method", International Conference on Mathematics and Mathematics Education, Şanlıurfa/Turkey, 2017
 May.



HACETTEPE UNIVERSITY GRADUATE SCHOOL OF SCIENCE AND ENGINEERING THESIS/DISSERTATION ORIGINALITY REPORT

HACETTEPE UNIVERSITY GRADUATE SCHOOL OF SCIENCE AND ENGINEERING TO THE DEPARTMENT OF MATHEMATICS

Date: 25/09/2018

Thesis Title / Topic: Some families of combinatorial matrices and their algebraic properties

According to the originality report obtained by my thesis advisor by using the *Turnitin* plagiarism detection software and by applying the filtering options stated below on 24/09/2018 for the total of 129 pages including the a) Title Page, b) Introduction, c) Main Chapters, d) Conclusion sections of my thesis entitled as above, the similarity index of my thesis is 9 %.

Filtering options applied:

- 1. Bibliography/Works Cited excluded
- 2. Quotes excluded
- 3. Match size up to 5 words excluded

I declare that I have carefully read Hacettepe University Graduate School of Sciene and Engineering Guidelines for Obtaining and Using Thesis Originality Reports; that according to the maximum similarity index values specified in the Guidelines, my thesis does not include any form of plagiarism; that in any future detection of possible infringement of the regulations I accept all legal responsibility; and that all the information I have provided is correct to the best of my knowledge.

I respectfully submit this for approval.

				Dat	Date and Signature	
Name Surname:	Talha Arıkan	0 0 0			25.09.2018	
Student No:	N13142394				Est)	
Department:	Mathematics	*				
Program:						
Status:	Masters	⊠ Ph.D.	☐ Integrated Ph.D.			

ADVISOR APPROVAL

APPROVED.

Prof. Dr. Adnan Tercan